

# THE UTILISATION OF ENTERPRISE RISK MANAGEMENT IN FAST-FOOD SMALL, MEDIUM AND MICRO ENTERPRISES OPERATING IN THE CAPE PENINSULA

by

# **BRUCE TAONA MASAMA**

Thesis submitted in fulfilment of the requirements for the degree

Master of Technology: Internal Auditing

in the Faculty of Business and Management Sciences

at the Cape Peninsula University of Technology

Supervisor: Dr Juan-Pierré Bruwer

Cape Town Date Submitted: November 2017

# **CPUT** copyright information

This thesis may not be published either in part (e.g. in scholarly, scientific or technical journals), or as a whole (e.g. as a monograph), unless permission has been obtained from the Cape Peninsula University of Technology.

#### DECLARATION

I, Bruce Taona Masama, declare that the content of this thesis represents my own unaided work, and that this thesis has not previously been submitted for academic examination towards any qualification, at any tertiary institution, either nationally or internationally. Furthermore, this thesis represents my own opinions and perceptions and not necessarily those of the Cape Peninsula University of Technology.

Signed: \_\_\_\_\_

Date: \_\_\_\_\_

#### ABSTRACT

It is widely accepted that small, medium and micro enterprises (SMMEs) are pivotal to any country's economic growth. In a South African dispensation, these business entities are believed to be the panacea to the economic challenges facing the country. Consequently, a number of initiatives have been deployed by national government to assist these business entities, inter alia, the formation of the Ministry of Small Business Development in 2014. Notwithstanding government support these business entities have received over the years, the sustainability of South African SMMEs still remains among the worst in the world. One of the main cited reasons for this weak sustainability is a lack of management skills, particularly risk management skills. Taking the latter into account, it comes as no surprise that previous studies show that South African SMMEs make use of customised risk management initiatives which are regarded as inadequate and/or ineffective. For this research study, the main objective was to determine the extent to which South African SMMEs utilise enterprise risk management (ERM) - a formal approach to manage risks in a holistic manner. To achieve the latter, a literature review was conducted from which relevant terms were conceptualised (see Chapter 2). Subsequently, primary data were collected using questionnaires, while taking into account various ethical considerations, and gleaned responses were analysed using both descriptive and inferential statistics (see Chapter 4). Although the inferential statistics suggested that the sampled South African SMMEs made use of ERM initiatives, further investigation revealed that these initiatives were used by chance as opposed to choice. Hence, it was concluded that these entities did not make use of ERM. In essence the results echoed the notion from scholarly literature that South African SMMEs unknowingly use ERM initiatives which are customised, inadequate and/or ineffective. Stemming from the conclusions reached, relevant recommendations were articulated to assist these business entities towards the implementation of ERM in a structured manner.

# ACKNOWLEDGEMENTS

I should like to acknowledge the following individuals who made this thesis a reality:

- Firstly, I should like to thank God for giving me strength during the *valleys* of this journey.
- My supervisor, Dr Juan-Pierré Bruwer, who sacrificed a great deal to ensure the success of this journey. He was patient with me, even during times when I gave him reasons to drop me. In addition, he gave me a reason to continue when I felt like giving up. In essence, I am a better researcher today because of him.
- My wife, Prisca Kupeyaveya, who stood by me throughout this journey.
- My mother, Shamiso Masama, and siblings. These people sacrificed their comfort for my sake, in order to make this journey a success.
- My CPUT family (Mr Trevor Aspeling, Mr Konstantinos Papathanasiou, Mrs Lara Aspeling, and colleagues). Words cannot adequately encompass my gratitude to these people for their support during this journey. May God bless them beyond their imagination!
- The government of Zimbabwe that sponsored my undergraduate studies, enabling me to embark on this research journey.

# DEDICATION

I wish to dedicate this piece of work to my late father, Kurai Jestias Masama, whose dream was to see me excelling in my studies.

# TABLE OF CONTENTS

DECLARATION	ii
ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
DEDICATION	
LIST OF FIGURES	ix
LIST OF TABLES	x
LIST OF ABBREVIATIONS	xi
LIST OF TERMINOLOGY	xii

CHAPTER 1: INTRODUCTION TO THE RESEARCH STUDY	1
SYNOPSIS	1
1.1 BACKGROUND TO RESEARCH STUDY	1
1.2 STATEMENT OF RESEARCH PROBLEM	4
1.3 RESEARCH QUESTIONS AND RESEARCH OBJECTIVES	4
1.3.1 Primary research question and primary research objective	5
1.3.2 Investigative research questions and secondary research objectives	5
1.4 RESEARCH DESIGN, RESEARCH METHODOLOGY AND RESEARCH METHO	DS 6
1.5 CONTRIBUTION OF THE RESEARCH STUDY	7
1.6 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY	7
CHAPTER 2: LITERATURE REVIEW	9
CHAPTER 2: LITERATURE REVIEW	
	9
SYNOPSIS	9 10
SYNOPSIS 2.1 INTRODUCTION	9 10 11
SYNOPSIS 2.1 INTRODUCTION 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs	9 10 11 ICAN
SYNOPSIS 2.1 INTRODUCTION 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs 2.3 ECONOMIC FACTORS INFLUENCING THE SUSTAINABILITY OF SOUTH AFR	9 10 11 ICAN 15
SYNOPSIS 2.1 INTRODUCTION 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMES 2.3 ECONOMIC FACTORS INFLUENCING THE SUSTAINABILITY OF SOUTH AFR SMMES AND THE ECONOMIC LANDSCAPE OF SOUTH AFRICA	9 10 11 ICAN 15 21
SYNOPSIS 2.1 INTRODUCTION 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs 2.3 ECONOMIC FACTORS INFLUENCING THE SUSTAINABILITY OF SOUTH AFR SMMEs AND THE ECONOMIC LANDSCAPE OF SOUTH AFRICA 2.4 RISKS AND RISK MANAGEMENT	9 10 11 ICAN 15 21 29
SYNOPSIS 2.1 INTRODUCTION 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs 2.3 ECONOMIC FACTORS INFLUENCING THE SUSTAINABILITY OF SOUTH AFR SMMEs AND THE ECONOMIC LANDSCAPE OF SOUTH AFRICA 2.4 RISKS AND RISK MANAGEMENT 2.5 RISK-MANAGEMENT APPROACHES	9 10 11 ICAN 15 21 29 31

2.7 RISK MANAGEMENT IN SOUTH AFRICAN SMMEs	38
2.8 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY	39

CHAPTER 3: RESEARCH DESIGN, RESEARCH METHODOLOGY AND RESEARCH	
METHODS	41
SYNOPSIS	41
3.1 INTRODUCTION	42
3.2 RESEARCH DESIGN AND RESEARCH METHODOLOGY	42
3.3 RESEARCH METHODS	43
3.4 ETHICAL CONSIDERATIONS	45
3.5 RESEARCH LIMITATIONS	47
3.6 DATA VALIDITY	47
3.7 DATA-COLLECTION TOOL	48
3.7.1 Questionnaire tool: Section A	48
3.7.2 Questionnaire tool: Section B	48
3.7.3 Questionnaire tool: Section C	49
3.7.4 Questionnaire tool: Section D	50
3.7.5 Questionnaire tool: Section E	51
3.8 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY	52
CHAPTER 4: DATA ANALYSIS, RESULTS AND DISCUSSION	53

SYNOPSIS	53
4.1 INTRODUCTION	54
4.2 DATA VALIDITY	54
4.3 DATA RELIABILITY	55
4.4 DEMOGRAPHIC INFORMATION	56
4.5 ECONOMIC FACTORS AFFECTING SOUTH AFRICAN SMMEs	60
4.6 RISKS FACED BY SOUTH AFRICAN SMMEs	62
4.7 RISK-MANAGEMENT INITIATIVES USED BY FAST-FOOD SMMEs	65
4.8 ALIGNMENT OF SMME RISK-MANAGEMENT INITIATIVES WITH THE COSO ERM	
FRAMEWORK	67
4.9 PERCEPTIONS OF RESPONDENTS REGARDING ERM	68
4.10 CONCLUSION AND RELEVANCE TO THE STUDY	70

CHAPTER 5: CONCLUSION	71
SYNOPSIS	71
5.1 INTRODUCTION	72
5.2 RESEARCH PROBLEM REVISITED	73
5.3 PRIMARY RESEARCH QUESTION AND PRIMARY RESEARCH OBJECTIVE	
REVISITED	73
5.4 INVESTIGATIVE RESEARCH QUESTIONS AND SECONDARY RESEARCH	
OBJECTIVES REVISITED	
5.5 CONCLUSION OF THE RESEARCH STUDY	
5.6 RECOMMENDATIONS	79
5.7 AVENUES FOR FURTHER RESEARCH	80
REFERENCES	81
ANNEXURES	105
ANNEXURE A: RESEARCH QUESTIONNAIRE	105
ANNEXURE B: DEMOGRAPHICS OF SAMPLE	108
ANNEXURE C: FREQUENCIES OF ORDINAL DATA	110
ANNEXURE D: FACTOR ANALYSIS	122

# LIST OF FIGURES

FIGURE 2.1: RISK MANAGEMENT PROCESS	26
FIGURE 2.2: RISK MATRIX	27
FIGURE 2.3: FOUR RISK TREATMENT OPTIONS	28
FIGURE 2.4: SME RISK ARCHITECTURE MODEL	32
FIGURE 2.5: SME RISK MANAGEMENT FRAMEWORK	33
FIGURE 2.6: THE COSO ERM FRAMEWORK	34
FIGURE 4.1: POSITION OF SAMPLED RESPONDENTS IN THEIR RESPECTIVE SMMES	56
FIGURE 4.2: NUMBER OF YEARS WHICH SAMPLED SMMES HAD BEEN IN EXISTENCE	57
FIGURE 4.3: NUMBER OF FULL-TIME EMPLOYEES BY SAMPLED SMMES	58

# LIST OF TABLES

TABLE 1.1: CLASSIFICATION OF SOUTH AFRICAN SMME SIZES	. 1
TABLE 1.2: SUMMARY OF THE INVESTIGATIVE RESEARCH QUESTIONS AND SECONDARY	
RESEARCH OBJECTIVES	.5
TABLE 2.1: NON-EXHAUSTIVE LIST OF DEFINITIONS OF THE TERM 'SUSTAINABILITY'	13
TABLE 2.2: ECONOMIC INDICATORS USED TO MEASURE THE SOUTH AFRICAN ECONOMIC	
LANDSCAPE1	18
TABLE 2.3: A NON-EXHAUSTIVE LIST OF DEFINITIONS OF THE TERM 'RISK'	22
TABLE 2.4: NON-EXHAUSTIVE LIST OF DEFINITIONS OF THE TERM 'RISK MANAGEMENT'	25
TABLE 3.1: QUESTIONS IN THE QUESTIONNAIRE TOOL - SECTION A	48
TABLE 3.2: QUESTIONS IN THE QUESTIONNAIRE TOOL – SECTION B	49
TABLE 3.3: QUESTIONS IN THE QUESTIONNAIRE TOOL - SECTION C	50
TABLE 3.4: QUESTIONS IN THE QUESTIONNAIRE TOOL - SECTION D	51
TABLE 3.5: QUESTIONS IN THE QUESTIONNAIRE TOOL - SECTION E	51
TABLE 4.1: CRONBACH'S ALPHA VALUES FOR THE FOUR LIKERT-SCALE QUESTIONS IN THE	
QUESTIONNAIRE	55
TABLE 4.2: PERCEIVED FINANCIAL SUSTAINABILITY OF SAMPLED SMMES	59
TABLE 4.3: ECONOMIC FACTORS AFFECTING THE SUSTAINABILITY OF SAMPLED SMMES6	31
TABLE 4.4: RISKS WHICH AFFECTED SAMPLED SMMES' SUSTAINABILITY	64
TABLE 4.5: RISK-MANAGEMENT INITIATIVES USED BY SAMPLED SMMES TO MANAGE RISKS 6	66
TABLE 4.6: SUMMARY OF EXPLORATORY FACTOR ANALYSIS RESULTS	67
TABLE 4.7: PERCEPTIONS OF RESPONDENTS REGARDING ERM	66

# LIST OF ABBREVIATIONS

Abbreviation	Meaning	
ACCA	Association of Chartered Certified Accountants	
ACI	Applied Communications Inc	
BBC	British Broadcasting Corporation	
CCTV	Closed-Circuit Television	
COSO	Committee of Sponsoring Organisations of the Treadway Commission	
ECO	Economic Cooperation Organization	
ERM	Enterprise Risk Management	
GDP	Gross Domestic Product	
GEM	Global Entrepreneurship Monitor	
IDC	Industrial Development Corporation	
IMF	International Monetary Fund	
IRMSA	Institute of Risk Management South Africa	
ISO	International Organization for Standardization	
КМО	Kaiser-Meyer-Olkin	
OECD	Organisation for Economic Co-operation and Development	
PWC	PricewaterhouseCoopers	
SBP	Small Business Project	
SME	Small and Medium Enterprise	
SMME	Small, Medium and Micro Enterprise	
UCLA	University of Carlifonia, Los Angeles	
US\$	United States Dollar	

# LIST OF TERMINOLOGY

Term	Conceptualised definition/definition/explanation
Adequacy	Adequacy refers to whether something is sufficient for a specific need (Merriam-Webster, 2017a).
Business entity	An organisation that is created by one or more natural persons, with the intention to conduct business (Perez, 2015).
Customise	To change something so that it can suit one's needs (Merriam- Webster, 2017b) (see Footnote 10 for further clarification).
Effective	Effectiveness refers to the ability to produce a desired effect (Merriam-Webster, 2017c).
Enterprise risk management	It is a process that is implemented by senior management in order to identify and manage risks that may affect the entity, thereby providing reasonable assurance regarding the achievement of entity objectives (Steinberg et al., 2004:2).
Failure	When a business entity ceases its operations, owing to its inability to generate profits (Financial Dictionary, 2017).
Gross domestic product	The monetary value of all final goods and services that are produced within the borders of a country during a given period of time (normally 12 months) (Callen, 2012).
Risk	The chances of threats and/or opportunities occurring, materialising and/or impacting, either positively or negatively, on the attainment of a business's overall sustainability (conceptualised in Chapter 2).
Risk management	It is the step-by-step process whereby potential threats (and opportunities) are identified and analysed in order to mitigate their realisation and potential impact (spur on their realisation and potential impact), with the main intent to provide reasonable assurance surrounding the attainment of relevant business objectives in the foreseeable future (conceptualised in Chapter 2).
Sound	Having characteristics of both adequacy and effectiveness.
Sustainability (general definition)	The long-term existence of a business entity through achieving set economic objectives, social objectives and/or environmental objectives, as implemented and agreed upon by relevant stakeholders of the relevant business entity (conceptualised in Chapter 2).
Sustainability (South African SMME context)	The long-term existence of an SMME through its attainment of sound financial performance and position in order to achieve relevant economic and social objectives as implemented and agreed upon by relevant stakeholders of the relevant business entity (conceptualised in Chapter 2).
Unemployment (broad definition)	The percentage of the labour workforce in a country not working, but actively seeking employment. In essence, this definition excludes those individuals that are unemployed and are not actively seeking employment (discouraged people) (South African Reserve Bank, 2017a).
Unemployment (narrow definition)	It refers to the percentage of total workforce, between 18 years and 64 years of age, who are unemployed, including those individuals who are looking for a paid job and those who have become discouraged to do so (Høj & Lewis, 2015:14).

# **CHAPTER 1**

## INTRODUCTION TO THE RESEARCH STUDY

#### SYNOPSIS

For this research study, a methodical process was followed whereby each chapter built on the previous chapter. This process is depicted below.

**Chapter 1**: Introduction to the research study

- •Background to the research study
- Statement of research problem
- Research questions and research objectives
- •Research design, research methodology and research methods
- ·Contribution of the research study
- •Conclusion and relevance of the research study

### Chapter 2: Literature review

**Chapter 3**: Research design, research methodolody and research methods

Chapter 4: Data analysis, results and discussion

Chapter 5: Conclusion

#### **1.1 BACKGROUND TO RESEARCH STUDY**

A large proportion of business entities in developing countries<sup>1</sup> take the form of small, medium and micro enterprises (hereafter referred to as SMMEs) (Gollin, 2008:219; Agwu & Emeti, 2014:102; Mbuyisa & Leonard, 2015:859). The definition of SMMEs differs from one country to another, and in South Africa these entities are defined by the *National Small Business Act No. 102 of 1996* (South Africa, 1996:2) as separate and distinct business entities including cooperative enterprises and non-governmental organisations, managed by one owner or more, which can be classified as a micro, a very small, a small or a medium enterprise. The abovementioned classifications are achieved by means of the number of full-time employees, their annual turnover, and/or by their total gross asset value. This is shown in Table 1.1.

Category	Employees	Turnover	Gross Asset Value
Micro	0 – 5	R0 – R150 000	R0 – R100 000
Very small	6 – 10	R150 001 – R1 000 000	R100 001 – R200 000
Small	11 – 50	R1 000 001 – R5 000 000	R200 001 – R1 000 000
Medium	51 – 100	R5 000 001 - R10 000 000	R1 000 001 – R2 000 000

Table 1.1: Classification of South African SMME sizes (Source: South Africa, 1996:16)

In looking at the classification criteria for employees, it is evident that South African SMMEs should play a pivotal role in the reduction of inequality through job creation (Ramukumba, 2014:20). In essence, these business entities add value to the national economy of South Africa, particularly through means of employing at least 55% of the national workforce (Ramukumba, 2014:19) and contributing between 30% and 57% to the national Gross Domestic Product (GDP) (Department of Trade and Industry, 2013; Bruwer & Coetzee, 2016:201).

Notwithstanding the aforesaid, it has been close to two decades since the South African government formally recognised SMMEs as the panacea to certain economic challenges facing the country (Maye, 2014). Given the support<sup>2</sup> these entities have received from government over the years, one would expect these economic challenges to have been resolved to a great extent (Ramukumba, 2014:22). Unfortunately, previous research studies (McGroarty, 2013; Maye, 2014; Zulu, 2014a; South African Reserve Bank, 2016:5) show that South Africa is still plagued with

<sup>&</sup>lt;sup>1</sup>A developing country is a country in which the majority of the population lives on less money, with fewer basic public services, compared with other countries around the world. In essence, a developing country has a weak industrial base (Tella & Tella, 2013:191).

<sup>&</sup>lt;sup>2</sup> Government support to SMMEs is evident from the formulation of various institutions like Ntsika Enterprise Promotion Agency and Khula Enterprise Finance, and the subsequent formulation of the Ministry of Small Business Development (Maye, 2014; South Africa, 2014).

poverty, unemployment and inequality (in terms of wealth dissemination). The latter view is supported by the South African unemployment rate (narrow definition<sup>3</sup>) which has remained along the lines of 25%, while the Gini index<sup>4</sup> has remained in the region of 0.7 (Lehohla, 2014:14; Statistics South Africa, 2015; Nedbank, 2017). This is quite concerning as more than 50% of the South African population is regarded as poor (Lehohla, 2014:12, South African Reserve Bank, 2015). Stemming from the aforementioned, the assumption can be made that South African SMMEs, to a large extent, do not achieve their legally imposed objectives. This sentiment is supported by a previous research study (Tustin, 2015:84) where it was found that the sustainability<sup>5</sup> of South African SMMEs is among the worst in the world.

To place the sustainability of South African SMMEs in better perspective, Brink et al. (2003:1) express the view that since the early 2000s, between 70% and 80% of South African SMMEs were believed to fail after being in operation for three years; resulting in millions of rands being lost to the national economy. In more recent times, research studies (Friedrich, 2016; Nicolson, 2017) found that approximately 70% of these business entities fail within their first four years of existence. This sentiment is greatly supported by Neneh and Van Zyl (2012:3364), who note that approximately 75% of newly established SMMEs cease to exist after being in operation for only five years. Although debatable, these statistics are quite disconcerting when taking into account that nothing much has changed with regard to the overall sustainability of these business entities over the years, when an estimated 800 000 South African SMMEs were believed to be in operation during the mid-1990s; during the early 2000s this number was estimated at 5 979 510 (South Africa, 1995; FinScope, 2010). Previous research studies have pinned the dismal sustainability of South African SMMEs on economic factors.

No business entity is immune from the influence of economic factors. In laypersons' terms, economic factors are variables in the economy that affect business operations (Fernando, 2011:41), and direct the movement of a nation's economy (Makos, 2015). More often than not,

<sup>&</sup>lt;sup>3</sup> The narrow definition of unemployment denotes the percentage of the labour workforce in a country that is not working, but actively seeking employment. In essence, this definition excludes those that are unemployed and are not actively seeking employment (discouraged people) (South African Reserve Bank, 2017a).

<sup>&</sup>lt;sup>4</sup> The Gini index measures inequality with regard to income distribution in a given country, ranging from 0 (absolute equality) to 1 (absolute inequality) (Lehohla, 2014:35).

<sup>&</sup>lt;sup>5</sup> The term 'sustainability' is conceptualised in Chapter 2.

economic factors are demarcated into macro-economic factors (those factors that have a direct influence on the economy of a country which are beyond the control of a business's management), and micro-economic factors (factors that have a direct effect on productivity of individual firms which can be controlled by a business's management) (De Beer, 2008:20; Delgado et al., 2012:10). According to previous research studies, the following economic factors, inter alia, have been found to have a direct influence on the sustainability of South African SMMEs: 1) changes in inflation rates, 2) changes in interest rates, 3) fluctuating economic growth, 4) changes in foreign exchange rates, 5) fluctuations in supply and demand, 6) employee satisfaction, and 7) stiff competition (Burda & Wyplosz, 2013:4; Kokemulle, 2016). For this very reason, it is imperative that these economic factors should be managed effectively. Based on previous research studies (Dzansi, 2004:139; Friedrich, 2016, Ayandibu & Houghton, 2017:137), however, one of the most common causes of business failure, especially in relation to South African SMMEs, is that of lack of management skills.

Management can be defined as a function which comprises the following four activities: 1) planning activities (formulation of business objectives, and ways to achieve them), 2) organising activities (mobilisation of resources and assigning of activities to be completed), 3) leading activities (motivating employees to attain set business objectives) and 4) controlling activities (measuring performance against set business objectives) (Norman, 2016). Although all activities are of equal importance, controlling activities are of great value when having to manage economic factors in and around any business. Specifically, controlling activities assist any business entity to achieve its relevant objectives in the foreseeable future through means of identifying, preventing and mitigating economic variables that may impede the attainment of its set objectives (Norman, 2016). In order to do so effectively, risk management<sup>6</sup> initiatives are often used.

Risk management can be regarded as a scientific process of well-defined, sequential steps that support better decision making through means of providing greater insights into risks and their actual and/or potential impacts (Endicott & Gardiner, 2011:20). Otherwise stated, risk management has to do with the identification, analysis, evaluation, and treatment of risks, with the main intent to enhance the attainment of relevant business objectives in the foreseeable future (Longenecker et al., 2013:632). In essence, risk management should enhance the sustainability of business entities (Nieman, 2001:446; McKay, 2016:62). Notwithstanding the latter, South

<sup>&</sup>lt;sup>6</sup> The terms 'risk' and 'risk management' are conceptualised in Chapter 2.

African SMMEs seem not to appreciate its importance. This is especially the case as previous research (Winks, 2008:20) shows that many of these business entities regard risk management to be relevant to large business entities only. Longenecker et al. (2013:632) concur in noting that risk management is mostly neglected by SMMEs to the detriment of their sustainability. This is of concern, since the sustainability of South African SMMEs is deemed poor, and ultimately adversely influence the national economy (Industrial Development Corporation, 2015). In core, regardless of the nature of a business entity, risk management is crucial since poorly managed risks can threaten a business's overall existence (Howard & Jawahar, 2002:95; Nichifor, 2016:250).

### **1.2 STATEMENT OF RESEARCH PROBLEM**

Stemming from the above, it appears that a large proportion of South African SMMEs do not make use of effective<sup>7</sup> and/or adequate<sup>8</sup> risk management initiatives (in general). As such, it culminates in a struggle for these business entities to become sustainable, ultimately having an adverse influence on their overall existence. Hence, for this study, the perception was formulated that South African SMMEs have a weak sustainability 'track record', owing to their insufficient utilisation of sound<sup>9</sup> risk management initiatives – particularly those related to enterprise risk management<sup>10</sup> (hereafter referred to as ERM).

# 1.3 RESEARCH QUESTIONS AND RESEARCH OBJECTIVES

Below, the research questions that guided this study, together with their objectives, are elaborated on. For the remainder of this section, discussions take place under the following headings: 1) primary research question and primary research objective, and 2) investigative research questions and secondary research objectives.

<sup>&</sup>lt;sup>7</sup> Effectiveness refers to the ability of producing a desired effect (Merriam-Webster, 2017c).

<sup>&</sup>lt;sup>8</sup> Adequacy refers to whether something is enough for a specific need (Merriam-Webster, 2017a).

<sup>&</sup>lt;sup>9</sup> Sound risk-management initiatives are both effective and adequate.

<sup>&</sup>lt;sup>10</sup> Enterprise risk management entails the continuous identification, analysis, evaluation, and treatment of risks across an organisation, holistically, in order to minimise unexpected losses (Lam, 2017:11).

### **1.3.1 Primary research question and primary research objective**

Based on the identified research problem, the primary research question within the ambit of this study reads as follows:

Do South African fast-food SMMEs operating in the Cape Peninsula utilise ERM to help enhance their overall sustainability?

The aim of the primary research question above was to achieve the primary research objective, which reads as follows:

To ascertain whether South African fast-food SMMEs operating in the Cape Peninsula utilise ERM to help enhance their overall sustainability.

### **1.3.2** Investigative research questions and secondary research objectives

Building on the primary research question and primary research objective above (see Section 1.3.1), a total of three investigative research questions and three secondary research objectives were identified. The latter are summarised in Table 1.2.

Sub-Question	Objectives
What are the risks which South African fast	To determine the risks that South African fast-food
food SMMEs in the Cape Peninsula face?	SMMEs in the Cape Peninsula face.
How are these risks treated by South African	To understand the risk-management initiatives used
fast-food SMMEs in the Cape Peninsula?	by South African fast food SMMEs in the Cape
	Peninsula, in dealing with the identified risks.
Are the risk treatments used by South African	To compare the risk-management initiatives used by
fast-food SMMEs in the Cape Peninsula	South African fast food SMMEs in the Cape
aligned to ERM?	Peninsula, against the COSO ERM framework, and
	determine if the initiatives are in accordance with this
	framework.

 Table 1.2: Summary of the investigative research questions and secondary research objectives

#### 1.4 RESEARCH DESIGN, RESEARCH METHODOLOGY AND RESEARCH METHODS

According to Mouton (2001:55), the research design of a research study pertains to the plan of how the actual study will be conducted. In essence, a research design can be categorised in terms of the:

- type of research study (empirical versus non-empirical);
- source of data (primary data versus secondary);
- nature of data (numerical versus textual); and
- level of control (structuring of the data-collection tool).

This research study was empirical in nature as it pertained to the observation of the status quo in South African fast-food SMMEs in the Cape Peninsula, regarding the utilisation of ERM. In order to establish a frame of reference, secondary data were used to develop relevant conceptual frameworks through means of a literature review (see Chapter 2), while primary data were collected through means of survey research in order to address the research questions within the ambit of this research study.

Data were collected through means of a standard questionnaire tool (see Annexure A) which comprised mostly closed-ended questions. Primary data were obtained from a representative sample size of a targeted population, namely that of members of management in South African fast-food SMMEs,<sup>11</sup> situated in the Cape Peninsula. To achieve this, non-probability sampling methods were deployed, particularly a mixture of purposive sampling and convenience sampling (see Section 3.3). Furthermore, members of management and their respective South African fast-food SMMEs had to adhere to relevant delineation criteria (see Section 3.3), and a total of 116 responses received were regarded as valid. After data were collected, they were appropriately analysed and discussed (see Chapter 4), while also taking into account appropriate ethical considerations throughout (see Section 3.4).

<sup>&</sup>lt;sup>11</sup> Owing to the ever-increasing levels of fast-food intake in South Africa (Maharaj, 2015; Holmes, 2016), it is not surprising that most South African entrepreneurs are starting their businesses in the fast-food industry. One of the reasons why most entrepreneurs prefer the fast-food industry is that this industry is not severely affected by an economic downturn (*Entrepreneur Magazine*, 2009:38). It is because of this growth in the fast-food industry that the researcher decided to focus on SMMEs in this industry, considering that the sustainability of these entities will generally entail the sustainability of the South African SMME sector as a whole.

### **1.5 CONTRIBUTION OF THE RESEARCH STUDY**

The topic of risk management has recently received much attention from regulators and senior management of organisations around the world (Guimond et al., 2010; Aven, 2016:1). In this study, risk-management initiatives utilised by South African fast-food SMMEs were investigated in order to add value to relevant stakeholders. In essence, this study does the following:

- Provides assistance to management of South African fast-food SMMEs to understand the status of their risk-management practices in relation to ERM, and also suggests areas where improvement can be effected. This is achieved through the conclusions and recommendations made in relation to the research problem, primary research question and primary research objective (see Chapter 5).
- Contributes to the body of knowledge through means of: 1) developed conceptual frameworks (see Chapter 2), 2) results and discussions that expound the risk-management initiatives used by South African fast-food SMMEs and their relevance to ERM (see Chapter 4), and 3) relevant conclusions and recommendations pertaining to the identified research problem, primary research question and primary research objective (see Chapter 5).
- Generates awareness among South African policy makers about the support needed by South African fast-food SMMEs in order to cultivate sound risk-management practices. This is done through the recommendations (see Chapter 5).

To add value to the abovementioned stakeholders, fundamental aspects in this study were shared (and will be shared in future) in the form of research articles and conference papers.

# **1.6 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY**

It is widely accepted that SMMEs are key economic drivers around the world. In a South African dispensation, these entities have received considerable attention owing to their immense contribution to employment creation, reduction of inequality (wealth dissemination) and the eradication of poverty. Notwithstanding the support which these business entities have received from national government over the years, their sustainability leaves much to be desired. In essence, the failure rate of South African SMMEs is among the highest in the world, and has been attributed to the various economic factors these entities face. Although these economic factors cannot be avoided, they can be managed using sound risk-management practices.

Using the above as foundation, a research problem was identified together with relevant research questions and objectives. Moreover, the research design, research methodology and research methods used in this research study were also briefly discussed in this chapter.

The remaining chapters which follow Chapter 1 are briefly explained below:

- **Chapter 2:** The main concepts introduced in Chapter 1 are expanded on in this chapter. Relevant conceptual frameworks guiding this study are developed, and general discussions (supported by theory) are provided in relation to the sustainability of South African SMMEs, economic factors affecting South African SMMEs and the economic landscape of South Africa, risk and risk management, risk-management approaches, enterprise riskmanagement frameworks, and risk management in South African SMMEs.
- **Chapter 3:** In this chapter, focus is placed on the research design, research methodology and research methods deployed in this research study. Furthermore, the ethical considerations taken into account for this research study are discussed, while emphasis is also placed on the questions in the data-collection tool.
- **Chapter 4:** The primary data gleaned are analysed by means of descriptive and inferential statistics, and the results are both presented and discussed in this chapter. The results were used to answer relevant research questions and, in turn, attain relevant research objectives.
- **Chapter 5:** Conclusions are drawn and relevant research questions and research objectives are revisited in this chapter. Recommendations are also made in order to mitigate the identified problem.

#### CHAPTER 2

### LITERATURE REVIEW

#### **SYNOPSIS**

For this research study, a methodical process was followed whereby each chapter built on the previous chapter. This process is depicted below.

Chapter 1: Introduction to the research study

#### Chapter 2: Literature review

#### Introduction

- •An overview of the sustainability of South African SMMEs
- •Economic factors influencing the sustainability of South African SMMEs and the economic landscape of South Africa
- Risk and risk management
- Risk-management approaches
- •Enterprise risk-management frameworks
- •Risk management in South African SMMEs
- •Conclusion and relevance to the research study

Chapter 3: Research design, research methodolody and research methods

Chapter 4: Data analysis, results and discussion

Chapter 5: Conclusion

#### 2.1 INTRODUCTION

SMMEs are often referred to as the lifeblood of many economies around the world (Hubbard & Ashton, 2013:9; Sippitt, 2014; Murisa & Chikweche, 2015:304; Musa & Chinniah, 2016:254; Ayandibu & Houghton, 2017:133). This view is supported by previous research studies (Association of Chartered Certified Accountants, 2010; Naidoo & Urban, 2010:234; Servon et al., 2010:301; Anane et al., 2013:1003; Small Business Project, 2013) where it was found that the socio-economic contribution of these business entities is of such an extent that policy makers around the world regard SMMEs as the driving forces of their respective economies. In a South African dispensation, these business entities account for at least 55% of total employment opportunities, while simultaneously contributing between 30% and 57% to the national GDP (Cant & Wiid, 2013:707; Ramukumba, 2014:19; Bruwer & Coetzee, 2016:201). Notwithstanding the notable contributions of South African SMMEs to the national economy, the sustainability of these entities still leaves much to be desired.

The sustainability of South African SMMEs is believed to be among the worst in the world (Smit & Fatoki, 2012:1136; Mafini & Muposhi, 2017:1). This sentiment is supported by Lekhanya (2015:412) and Small Business Project (2013) who share the view that about 70% of South African SMMEs fail within their first four years of operation, while Wiese (2014:38), Herrington et al. (2015:17), and Bruwer and Van den Berg (2017:8), state that close to 75% of South African SMMEs fail in their first 42 months of existence. In addition to the high failure rate of these business entities, South Africa's unemployment statistics further suggest that these SMMEs may not be sustainable since one of their core socio-economic objectives, to reduce unemployment, is not being achieved effectively (South Africa, 1995:11; Trading Economics, 2016a).

Notwithstanding the above, the weak sustainability of South African SMMEs is often pinned on a plethora of economic factors (Bruwer et al., 2013; Bruwer & Van den Berg, 2017:4) while operating in a harsh economic landscape (South Africa, 2016:2), giving rise to an abundance of risks. In quintessence, when these risks materialise, they may have an adverse influence on the sustainability of South African SMMEs (Howard & Jawahar, 2002:95; Pollard & Stephen, 2008; Duren, 2016; RobecoSAM, 2016). Though risks are inevitable, they can and should be managed through the deployment of sound risk-management initiatives (Louisot & Ketcham, 2014). Risk management pertains to the identification of potential risks in advance, their analysis, and taking steps to treat these risks to help achieve the objectives of a business entity (Ramakrishna,

2015:215). According to previous research studies (Duong, 2009:22; Chakabva, 2015:59), South African SMMEs do not have systematic processes to manage risks, as they rely mostly on their own customised<sup>12</sup> internal controls to manage relevant risks (Smit & Watkins, 2012:6328; Bruwer & Siwangaza, 2016:112).

Taking the above into account, it becomes apparent that the adequate management of risks is of paramount importance in South African SMMEs, as supported by their weak sustainability rates. For the remainder of this chapter, the following phenomena are discussed in greater detail: 1) the sustainability of South African SMMEs, 2) economic factors influencing the sustainability of South African SMMEs, 2) economic factors influencing the sustainability of South African SMMEs, 3) risks and risk management, 4) risk-management approaches, 5) enterprise risk-management frameworks, and 6) risk management in South African SMMEs.

# 2.2 AN OVERVIEW OF THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs

In South Africa, SMMEs are defined by the National Small Business Act No. 102 of 1996 (South Africa, 1996:2), as separate and distinct business entities including co-operative enterprises and non-governmental organisations, managed by one owner or more, which can be classified (see Section 1.1) as micro enterprises, very small enterprises, small enterprises or medium enterprises. Based on this piece of legislation, these business entities have three main socio-economic objectives, namely, to: 1) create jobs, 2) foster economic growth, and 3) alleviate poverty through the equal dissemination of wealth (Abor & Quartey, 2010:218; Masutha & Rogerson, 2014:141; Bruwer & Coetzee, 2016:201).

When taking into account that approximately 91% of all operational South African business entities are regarded as SMMEs (Abor & Quartey, 2010:219; Ayandibu & Houghton, 2017:135), their socio-economic importance to the national economy is placed in better perspective. It is therefore not surprising that the South African government regards SMMEs as a pivotal part of the national economy (South Africa, 1995:11), which is further substantiated by the establishment of the Ministry of Small Business Development in 2014 (South Africa, 2014). The importance of

<sup>&</sup>lt;sup>12</sup> The term 'customise' means to change something so that it can suit one's needs (Merriam-Webster, 2017b). South African SMMEs use internal controls that satisfy the requirements of mangers and/or owners of these entities (Smit & Watkins, 2012:6328). In essence, most of these individuals have a limited understanding of risk management; hence it can be expected that these customised internal controls will be far from ideal (University of Stellenbosch Business School, 2014; Sunjka & Emwanu, 2015:1469).

these entities are further clarified by previous research studies (National Planning Commission, 2012:140; Department of Trade and Industry, 2013; Makakane, 2014; Zulu, 2014a; Bureau for Economic Research, 2016:31; Kelley et al., 2016:105; Banking Association South Africa, 2017; Herrington & Kew, 2017:93) where it was found that South African SMMEs were responsible for:

- creating more than 50% of local employment opportunities in 2014;
- contributing more than 40% of the country's total remuneration in 2014;
- contributing an average of 40% to the national GDP in 2014;
- contributing an average of 36% to the national GDP in 2015; and
- providing employment to about 60% of the national labour force in 2016.

Hence, clear tangent planes emerge as to why South African SMMEs are actively supported by national government, especially through its National Development Plan 2030, where it is expected that these business entities will be at the forefront of job creation within the next few years (National Planning Commission, 2012:140). Taking the above into account, it becomes apparent that South African SMMEs do add immense socio-economic value to the national economy. Notwithstanding the aforementioned, South African SMMEs are believed to have one of the worst sustainability rates in the world (Bruwer & Coetzee, 2016:201).

Before elaborating on the sustainability of South African SMMEs, the term 'sustainability' needs to be conceptualised within the ambit of this study. In layperson's terms, sustainability generally refers to the capacity of a business entity to continue operating while meeting its various set objectives, as implemented by various stakeholders (Pollard & Stephen, 2008). However, the term 'sustainability' is still very broad in nature as the objectives which can be attained by any business entity can be demarcated into three types of objectives, namely: 1) economic objectives,<sup>13</sup> 2) environmental objectives,<sup>14</sup> and 3) social objectives<sup>15</sup> (Zink et al., 2008:10; Bruwer & Coetzee, 2016:202). To assist with the conceptualisation of the term 'sustainability', a non-exhaustive list of definitions is provided in Table 2.1.

<sup>&</sup>lt;sup>13</sup> Economic objectives are those goals that relate to the optimisation of financial performance and financial position of a business entity (Willard, 2012:11), and examples are the attainment of profitability, solvency, liquidity and efficiency (Dwivedi, 2002:280; Plumridge, 2010:281; Bruwer & Coetzee, 2016:201).

<sup>&</sup>lt;sup>14</sup> Environmental objectives are those goals that relate to the protection of natural resources, and examples include the prevention of water pollution and air pollution (Okuboyejo, 2013:9).

<sup>&</sup>lt;sup>15</sup> Social objectives refers to those goals that relate to the betterment of society at large (Conservation Gateway, 2017), and examples include the reduction of income inequality and poverty levels (Sustainable Environment, 2016).

**Table 2.1:** Non-exhaustive list of definitions of the term 'sustainability'

Definition	Source
Long-term continuation of a business entity through means of achieving its relevant objectives; relative to the fulfillment of economic responsibilities, environmental responsibilities and social responsibilities	Bruwer & Coetzee (2016:202)
Ability [of a business entity] to last or continue for a long time	Merriam-Webster (2016b)
Use of well-balanced, functional solutions, to create value in the economic, social and ecological realms of corporate performance	Flouris & Yilmaz (2016)
Creation of long-term shareholder value by embracing opportunities and managing risks deriving from economic, environmental and social developments	RobecoSAM (2016)
[It] includes longevity and retaining of core principles or objectives, regardless of internal and external changes over time	Bateh et al. (2013:1)
Meeting society's expectation that [business entities] add social, environmental and economic value from their operations, products and services	Hopkins (2012:20)
Being resilient and able to create economic value, healthy ecosystems and strong communities	Laughland & Bansal (2011)
Capacity of a business to continue to operate successfully	Pollard and Stephen (2008)

Stemming from the definitions in Table 2.1 above, although sustainability is defined in many ways, it encompasses two main aspects: 1) the long-term existence of a business entity, and 2) the attainment of core objectives (relative to economic objectives, social objectives and/or environmental objectives). Therefore, using the aforementioned as basis, the term 'sustainability' is conceptualised as follows throughout this research study:

The long-term existence of a business entity through achieving set economic objectives, social objectives and/or environmental objectives, as implemented and agreed upon by relevant stakeholders of the relevant business entity.

In a South African SMME dispensation, though the above conceptualisation holds truth, these business entities' objectives are more aligned towards social- and economic objectives. This view is supported by previous research studies (Vijfvinkel et al., 2011:7; Sokoto & Addullahi, 2013:189) where it was found that the sustainability of South African SMMEs are mostly influenced by their attainment of relevant socio-economic objectives, which include the decreasing of unemployment rates and the alleviation of poverty. In order to attain the aforementioned, these business entities should have sound financial performance and financial positions, especially for them to remain in existence for the foreseeable future. This view is supported by past research studies (Wiese, 2014:60; Tustin, 2015:84), which share the notion that most South African SMMEs fail because

of poor financial sustainability.<sup>16</sup> Therefore, building on the latter, the term 'sustainability', in a South African SMME dispensation, can be conceptualised as follows:

The long-term existence of an SMME through its attainment of a sound financial performance and financial position, in order to achieve relevant economic objectives and social objectives, as implemented and agreed upon by relevant stakeholders of the relevant business entity.

To obtain a better impression of South African SMMEs' weak sustainability, recent research studies (Maye, 2014; Herrington et al., 2015:17) show that up to 75% of South African SMMEs fail within their first 42 months of existence. This is quite concerning considering that since the formal recognition of South African SMMEs, through the implementation of the National Small Business Act No. 102 of 1996 (South Africa, 1996), their long-term existence has not improved to any great extent (Ligthelm, 2010:140; Bruwer et al., 2013; Tustin, 2015:84; Hendricks et al., 2015:87). A plausible reason for this is that these business entities' financial performance is generally regarded as poor (Kupka & Thomas, 2014:3200; Peters et al., 2014:1131). In addition, the financial position of South African SMMEs also leaves much to be desired (Chakraborty, 2015:227; Kengne, 2015:120). A possible reason for SMMEs' poor financial performance and/or financial position, inte alia, is inefficiencies in their business operations (Enow & Brijlal, 2014:11)

When taking into account that South African SMMEs' contribution to the national GDP was estimated at 52% in 2013, 45% in 2014 and 39% in 2015 (Zulu, 2014b; Bureau for Economic Research, 2016:31; Kelley et al., 2016:105; Herrington & Kew, 2017:93), it is apparent that the contribution of these business entities to the national economy is diminishing. This view is supported by previous research studies (Groepe, 2015:2; Høj & Lewis, 2015:14; Kengne, 2015:113), where it was found that the growth rate of the South African economy is insufficient to absorb the current supply of labour in the country. In essence, these business entities' contributions to the national GDP are drifting away from the idealistic expected contribution (between 60% and 80%), as set by the government (Zulu, 2014b). This notion is further supported

<sup>&</sup>lt;sup>16</sup> Financial sustainability refers to the ability of a business to obtain revenues in order to sustain productive processes, thereby resulting in the attainment of business objectives (León, 2001:11). It includes profitability, liquidity and solvency (Wiese, 2014:35)

by the South African unemployment rate (broad definition<sup>17</sup>) which has been hovering around 33% in recent years (Statistics South Africa, 2014; Statistics South Africa, 2015; Statistics South Africa, 2016; Nedbank, 2017; Statistics South Africa, 2017b).

In summation, when focus is shifted to the business discontinuance rate<sup>18</sup> of South African SMMEs, it was estimated at 4.9% in 2013, 3.9% in 2014, 4.8% in 2015, and 10% in 2016 (Kelley et al., 2016:21; Herrington & Kew, 2017:109). Although these percentages appear small, it should be noted that at least nine out of every ten South African businesses are regarded as SMMEs (Okuboyejo, 2013:2), meaning that in recent times a reasonable number of South African SMMEs had to close their doors (see above), which led to the probable layoff of employees, adversely influencing national unemployment and national poverty (Fatoki, 2014:926; Zulu, 2015).

Stemming from the above, it becomes apparent that South African SMMEs have weak sustainability rates. The weak sustainability alluded to above is mainly attributed to expansive economic factors which affect these business entities (Pilinkiene et al., 2016:23).

# 2.3 ECONOMIC FACTORS INFLUENCING THE SUSTAINABILITY OF SOUTH AFRICAN SMMEs AND THE ECONOMIC LANDSCAPE OF SOUTH AFRICA

Similarly to a 'factor', which is defined as a phenomenon that influences a particular result (Merriam-Webster, 2016a), an economic factor refers to any influencer that may have an impact on the national economy of a country, including its inhabitants (natural- and legal persons) (Fernando, 2011:41; Plan Projections, 2015). In general, economic factors are categorised into two groups: micro-economic factors and macro-economic factors (Kraja & Osmani, 2015:122). The two subsets of economic factors are briefly explicated below:

• <u>Macro-economic factors</u>: Macro-economic factors are those economic factors that affect an entire economy (i.e. natural and legal persons) and to a large extent cannot be controlled by

<sup>&</sup>lt;sup>17</sup> The broad definition of unemployment denotes the percentage of total workforce (between 18 years and 64 years of age) who are unemployed, including those people who are actively looking for a paid job and those who have become discouraged from doing so (Posel et al., 2013).

<sup>&</sup>lt;sup>18</sup>The business discontinuance rate is the percentage of individuals aged between 18 years and 64 years of age who, in the past 12 months, have discontinued a business they owned, either by selling it, shutting it down, or otherwise discontinuing an owner/management relationship with the business (Singer et al., 2015:24). These discontinuation rates mostly constitute businesses that cease their operations.

management (Ngary et al., 2014:912; Bruwer & Coetzee, 2016:205). Macro-economic factors are categorised under social-demographic factors, economic factors, political-legal factors and technological factors (Glowik & Smyczek, 2015). Examples of macro-economic factors which impact on the South African economy include, inter alia, high crime and corruption rates, high water costs, unreliable supply of raw materials, low demand, high unemployment, high interest rates, high inflation rates, expensive and unreliable electricity supply, strict government regulations, stiff external competition, poor infrastructure, and rapid changes in technology (Prescott et al., 2002:489; Von Ketelhodt & Wöcke, 2008:4; Karpak & Topcu, 2010:67; Watson, 2010:1; Kadocsa & Francsovics, 2011:30; Africagrowth Institute, 2012:7; Verma, 2012; Small Business Project, 2013; Gossow et al., 2015; Bureau for Economic Research, 2016; Pilinkienė et al., 2016:23, Bruwer & Van den Berg, 2017:7).

<u>Micro-economic factors</u>: Micro-economic factors are those economic factors which affect individual business entities but can be controlled by management to a large extent (Research & Education Association Editors & Fogiel, 2012:2). In addition, micro-economic factors also affect natural persons (i.e. customers and employees) residing in the applicable economy in which they exist (Lin et al., 2011:313). Unlike macro-economic factors that are usually homogeneous across a country's economy, significant micro-economic factors vary from one industry to another and from one business entity to another (Koralun-Bereźnicka, 2013:9). Examples of micro-economic factors that impact on the South African business entities include, but are not limited to, lack of access to credit facilities, lack of proper marketing skills, non-payment by debtors/customers, availability of substitute products, lack of business infrastructure, weak business locations, and poor management skills (Abor & Quartey, 2010:218; Hussain et al., 2012:1582; Sunjka & Sklar-Chik, 2012:741; Kotelnikov, 2013; Agwu & Emeti, 2014:104; Mthabela, 2015:ii; Nwankwo & Aiyeku, 2015:17; Bruwer & Van den Berg, 2017:7).

Notwithstanding the above, economic factors have a mammoth influence on the economic landscape<sup>19</sup> of any country in which business entities have to operate (Fagerberg, 2013:27).

<sup>&</sup>lt;sup>19</sup> Economic landscape refers to the *painted picture* of an economy that arises from a combination of various economic indicators (Opdam, 2006:54). Alternatively, an economic landscape can be defined as the condition/state of a country's economy that influences business entities and natural persons alike (Pushparaj, 2015; *Cambridge Dictionary*, 2016)

Based on the examples of economic factors which influence the South African economy, the inference can be made that the South African economic landscape is unfavourable for SMMEs to operate in. This view is supported by previous studies (South African Risk and Vulnerability Atlas, 2009; Bruwer, 2016:263), where the South African economic landscape is described as 'harsh'. In order to have a better understanding of how the South African economic landscape appears, a total of seven<sup>20</sup> key economic indicators were analysed (Smith, 2011; Wealth Management; 2014; Trading Economics, 2016b). These economic indicators are expanded on below:

- <u>GDP (real)</u>: The monetary value of all final goods and services that are produced within the borders of a country during a given period of time (normally 12 months) (Callen, 2012). This indicator is used to measure national income, national output and national expenditure, and is usually stated in US\$ the effects of inflation are factored out in calculating real GDP (Chowdhury, 2008:53). It is the real GDP that gives a better picture of a country's economic development (McTaggart et al., 2013:444). Furthermore, this indicator is mostly used to determine the overall health of an economy (Coyle, 2015:4).
- <u>GDP per capita (real)</u>: It is calculated by dividing the GDP (real) by the estimated population size, and it depicts the average expected income per person in a country (Pettinger, 2011). Otherwise stated, this indicator shows the average value which each citizen should have added to the relevant country's economy (Bruwer, 2016:111) the effects of inflation are factored out. This indicator can be used to measure the average income of a country's population, thereby giving a broad picture of the wellbeing of a country's population (Organisation for Economic Co-operation and Development, 2009; Kula et al., 2010:431).
- <u>Unemployment rate (broad definition)</u>: It refers to the percentage of total workforce, between 18 years and 64 years of age, who are unemployed, including those individuals who are looking for a paid job and those who have become discouraged from doing so (Høj & Lewis, 2015:14). The broad definition of unemployment accurately depicts the unemployment problem in a country (Posel et al., 2013:6).
- <u>Population</u>: The estimated number of people that are deemed official inhabitants (citizens) of a country.

<sup>&</sup>lt;sup>20</sup> Although there are many key economic indicators, the researcher made use of only seven. The reasoning behind the use of the selected seven indicators is as follows: the GDP, GDP per capita, Gini index and unemployment are directly connected to the socio-economic objectives of South African SMMEs, whereas inflation rate and interest rate constitute major factors that are perceived to impede South African SMMEs from achieving their objectives. The economic indicator of population was mainly included to draw inferences on statistics pertaining to the other economic indicators.

- <u>Gini index</u>: Measures inequality with regard to income distribution in a given country, ranging between 0 (absolute equality) and 1 (absolute inequality) (Lehohla, 2014:35).
- <u>Inflation rate</u>: The percentage increase in prices of basic goods and services (World Bank, 2016). Although the prices change constantly, it is the average annual change that will be focused on.
- <u>Prime interest rate</u>: The interest rate at which the reserve bank lends money to commercial banks (*Financial Dictionary*, 2016), which becomes the foundation which commercial banks use to lend money to individuals and business entities (Neiman, 2010).

Relevant statistics pertaining to the seven economic indicators above are summarised in Table 2.2 for 2013 to 2016, after which they are analysed and interpreted accordingly.

**Table 2.2:** Economic indicators used to measure the South African economic landscape (**Sources:** World Bank, 2013; Statistics South Africa, 2014; International Labour Organization, 2015:19; Forslund, 2016; Kganyago, 2016; Statistics South Africa, 2015; International Monetary Fund, 2016; Knoema, 2016; Kganyago, 2016; Liberta, 2016; South African Government News Agency, 2016; South African Reserve Bank, 2016; Statistics South Africa, 2016; South Africa, 2016; Trading Economics, 2016; USForex, 2016; World Bank, 2016; Worldometers, 2016; Clarno, 2017:213; Inflation.eu, 2017; Knoema, 2017; Nedbank, 2017; South African Reserve Bank, 2017; South Afri

	ion, oodan and an tooon o Bank, 2011 b, otadodoo oodan anda, 2011 a, tronaomotoro, 2011 j				
Indicator	2013	2014	2015	2016	
GDP (real)	US\$324 billion	US\$329 billion	US\$333 billion	US\$336 billion	
GDP per capita (real)	US\$6 090	US\$6 086	US\$6 057	US\$5 966	
Unemployment rate (broad definition)	35.3%	35.3%	34.8%	37.8%	
Population (estimated)	53 416 609	53 969 054	54 490 406	54 978 907	
Gini index	0.63	0.77	N/A	N/A	
Average annual inflation rate	5.77%	6.15%	4.51%	6.59%	
Average annual prime interest rate	8.5%	9.1%	9.4%	10.5%	

From the statistics in Table 2.2, the following observations and inferences are made:

GDP (real): The changes in real GDP for the four years under review entailed an increase of 1.5% (US\$5 billion) between 2013 and 2014, a 1.2% (US\$4 billion) increase between 2014 and 2015, a 0.9% (US\$3 billion) increase between 2015 and 2016, and a net increase of 3.70% (US\$12 billion) from 2013 to 2016. Stemming from the net increase in GDP over the applicable period, it can be inferred that: 1) there was an increase in the productivity of citizens, 2) an increase in capital investments, and/or 3) there was a decrease in national unemployment. Although the GDP has been on the rise, signalling an increase in volumes of goods and services produced, the increase noted above is regressive, since the increase in one year is less than the increase in the previous year.

- GDP per capita (real): Notwithstanding the GDP figures, the real GDP per capita figures decreased from year to year between 2013 and 2016. The changes is real GDP per capita are as follows: a decrease of 0.07% (US\$4) between 2013 and 2014, a decrease of 0.48% (US\$29) between 2014 and 2015, a decrease of 1.5% (US\$91) between 2015 and 2016, and a net decrease of 2.04% (US\$124) between 2013 and 2016. This implies that the growth in the GDP did not match the growth in the country's population. When taking into consideration that the GDP per capita can be used to show the average monetary value added by the average South African citizen, the above trend suggests that the productivity of South African citizens diminished over the four years in consideration. This nullifies the assumption made earlier that GDP might have increased owing to the increased productivity of South African citizens. Finally, if GDP per capita is used as an informal basis to measure poverty, this trend suggests that the majority of South Africans are getting poorer instead of getting richer, thereby resulting in inequality in the distribution of income. This view is supported by research studies (Grant, 2015; Musgrave, 2015; Nicolson, 2015; South African Reserve Bank, 2015) where it was found that approximately 54% (27 million) of South Africans lived below the national poverty line<sup>21</sup> in 2015.
- Unemployment rate (broad definition): For the period under review, the unemployment rate has been in the region of 33%. The changes in unemployment from 2013 to 2016 are as follows: an increase of 1.26% between 2013 and 2014, a decrease of 0.93% between 2014 and 2015, an increase of 3.78% between 2015 and 2016 and a net increase of 4.11% between 2013 and 2016. The marginal increase between 2013 and 2014, together with a decrease between 2014 and 2015, supports the assumption made earlier that the GDP could have increased because of a reduction in unemployment. In the same vein, the increase in unemployment in 2016 tallies with the minor change in GDP for the same year. Although the abovementioned changes might seem insignificant, thereby suggesting a general improvement in the unemployment pandemic in South Africa, current unemployment levels are still extremely high. To have a better understanding of the quagmire regarding unemployment, the number of unemployed persons should be considered. The number of unemployed persons in South Africa was approximated at 7 216 750 in 2013, 7 493 000 in 2014, 7 678 000 in 2015, and 8 138 500 in 2016. These estimated figures translate to an increase of 276 250 unemployed citizens (3.83% increase) between 2013 and 2014, an

<sup>&</sup>lt;sup>21</sup> Poverty line refers to the minimum monthly amount that a person needs for daily survival. In essence, individuals living below the poverty line earn less than R779 per month (Grant, 2015).

increase of 185 000 unemployed citizens (2.47% increase) between 2014 and 2015, an increase of 460 500 unemployed citizens (6% increase) between 2015 and 2016, and a net increase of 921 750 unemployed citizens (12.77% increase) between 2013 and 2016. Since the level of unemployment in a country affects the buying power of citizens in that country, the South African unemployment rates are of great concern as they will likely translate to insufficient demand of good and services, adversely impacting on the economy as a whole in the foreseeable future.

- <u>Gini index:</u> The statistics shown give an impression that the South African Gini index is moving towards 1 rather than the desired 0. Although the last measurement of the South African Gini index was done in November 2014 (Clarno, 2017:213), it can be expected that the current Gini index is above 0.77, considering the trend depicted by the real GDP per capita. This affirms the assumption made above that economic inequality is still a problem (Jones, 2017), particularly since the productivity of South African citizens is decreasing. This view is shared by recent studies (Keeton, 2014:29; Sanieni-Pour, 2015; Bond, 2016), which show that wealth is not equally shared in South Africa. In essence, South Africa is regarded as one of the most unequal countries with regard to wealth distribution (Landman et al., 2003:3; Keeton, 2014:29; BBC News, 2015).
- Average annual inflation rate: The inflation rate increased year on year, meaning that the cost of living also increased year on year between 2013 and 2016. In laypersons' terms, if an item had a cost of R20.00 at the start of 2013, this same item would cost R21.15 at the start of 2014, R22.45 at the start of 2015, R23.46 at the start of 2016, and R25 at the start of 2017. The inflation rate was highest in 2016, which could be a possible reason why the GDP growth in 2016 was somewhat dismal. In essence, inflation rates may have resulted in increased costs of production while simultaneously eroding the disposable income of consumers, which may be a possible reason why the GDP growth rates are well below the expectations set by national government (Vermeulen, 2015:2).
- <u>Average annual prime interest rate</u>: For the four years under review, the prime interest rate increased by 7.06% between 2013 and 2014, increased by 3.30% between 2014 and 2015, and increased by 11.70% between 2015 and 2016, resulting in a net increase of 23.53% between 2013 and 2016. The significant effect of high and increasing prime interest rates is that it limits new investments in business entities, particularly SMMEs, as they cannot access

relevant financial resources through external funding, owing to the extensive cost of obtaining financing (Schmidt et al., 2016). This also adds to the above possible reasons as to why the growth rate of South African GDP has been sluggish, and serving as an avenue for further research.

Based on the above, it appears that the South African economic landscape is quite 'harsh' and not ideal for optimising business entities' (especially SMMEs) operations and/or sustainability. The latter is particularly true as South Africa's entrepreneurial framework conditions<sup>22</sup> were regarded as insufficient in a recent Global Entrepreneurial Monitor (GEM) report (Kelley et al., 2016:138) – there is much improvement required in relation to the South African economic landscape (Gossow et al., 2015; South African Government News Agency, 2016). Taking into account the aforementioned, it is not surprising that the South African economic landscape is referred to as a sort of 'breeding ground' for many risks (Miles, 2011:19; Bruwer et al., 2013), particularly for South African SMMEs. Since approximately 75% of South African SMMEs fail after being in existence for 42 months (Herrington et al., 2015:17), it is highly probable that these business entities are not managing risks effectively.

#### 2.4 RISKS AND RISK MANAGEMENT

Risks are inevitable and evident in all business entities across the globe (Shaw, 2010:25; Bruwer, 2016:131). In a business environment, the term 'risk' is strongly associated with the materialisation of uncertain events which will impact (positively or negatively) on the attainment of business objectives (Akintoye et al., 2012:194; Aven, 2012; Firoozye & Ariff, 2015:xvi). Owing to the subjectivity existing around the term 'risk' (State of Queensland, 2011:7; Smit, 2012:26; Grose, 2014; Haimes, 2015:51), a non-exhaustive list of definitions is provided in Table 2.3 overleaf.

<sup>&</sup>lt;sup>22</sup> Entrepreneurial framework conditions are those socio-economic conditions that enhance and/or hinder new business creation and their growth (Global Entrepreneurship Monitor, 2017).

Table 2.3: A non-exhaustive list of definitions of the term 'risk'

Definition	Reference	
Possibility that an undesirable event will occur, thereby negatively	Ramakrishna (2015:214)	
affecting the attainment of business objectives		
An event that may adversely affect a business's ability to achieve	McNeil et al. (2015:3)	
its objectives and execute its strategies		
The potential for [the] realisation of unwanted, adverse	Society for Risk Analysis	
consequences [in a business environment]	(2013)	
The possibility of economic or financial losses or gains, as a		
consequence of the uncertainty associated with a specific plan of	(2013:187)	
action		
The function of an event's likelihood and its consequences on the	Dumbravă & lacob (2013:78)	
achievement of business objectives		
The negative or positive effect of uncertainty on an organisation's	International Organization for	
objectives	Standardization (2009)	
Positive or negative effect of the consequence of an uncertain	International Risk Governance	
event or activity on something that humans value	Council (2005)	
Combination of the probability of an event occurring and the	Institute of Risk Management	
consequences of that event on the achievement of an	(2002)	
organisation's objectives		

Stemming from the above, the inference can be made that risks have to do with the occurrences of uncertain events and their effects on the attainment of business objectives, particularly their overall sustainability. In some cases, risks can also impact on the manner in which strategies (operational and non-operational) are deployed in business entities. The latter view is greatly supported by previous research studies (Brockett & Rezaee, 2012; Duren, 2016; Flouris & Yilmaz, 2016:5) where it was found that risks affect the performance, continuation, and resilience of businesses. Using the aforementioned as basis, the term 'risk' is conceptualised as follows throughout this study:

The chances of threats and/or opportunities occurring, materialising and/or impacting, either positively or negatively, on the attainment of a business's overall sustainability.

There exist many different types of risks which may influence business entities' sustainability (Kaplan & Mikes, 2012:50). More often than not, risks are demarcated into four broad categories: 1) strategic risks, 2) operational risks, 3) reporting risks, and 4) compliance risks (Epstein & Buhovac, 2006:10; Curtis & Carey, 2012; Godbole, 2012; Deloitte, 2013:8). Each of these four categories is briefly expanded on below:

• <u>Strategic risks</u>: Strategic risks refer to those threats and/or opportunities that influence the attainment of strategic objectives (Mohammed & Sykes, 2012; Taylor, 2012:8). In essence,

these risks significantly influence the attainment of a business's vision and mission (Bruwer, 2016:48); hence, they impact on a business in its entirety (Mohammed & Sykes, 2012). If not managed properly, strategic risks can threaten a business's existence (Allan & Beer, 2006; Frigo & Anderson, 2011; Tonello, 2012). Examples of strategic risks include competition, changes in economic conditions, changes of customer preferences, changes in legislation, weak business growth, delays in supply chain process, access to long-term financing, and changes in customer demands (Institute of Risk Management, 2002:3; Allan & Beer, 2006; Association of Chartered Certified Accountants, 2015).

- <u>Operational risks</u>: Operational risks refer to threats and/or opportunities resulting from the economy, effectiveness and efficiency of procedures or systems of a business entity (Bank for International Settlements, 2011:3; Rawson, 2013; Kelliher et al., 2016). Operational risks are inherent in daily business operations (Hussain & Shafi, 2014), and if not managed appropriately, they may result in loss of income. Examples of operational risks include occupational theft and fraud, poor product quality, poor service quality, lack of qualified employees, human error in processing transactions, and system failures, to mention but a few (Muehlenbrock et al., 2012; Ayandibu & Houghton, 2017:136; Risk Management Association, 2017).
- Reporting risks: These are threats and/or opportunities that affect the reliability and effectiveness of both internal and external reporting of business operations (Rittenberg, 2006; Wurzler, 2013:2), which can be either financial or non-financial in nature (Von Rössing, 2007:26). In businesses, reporting risks emanate from people who are responsible for gathering and analysing data, processes and systems (Center for Audit Quality, 2012). More often than not, reporting risks may result in financial loss, and/or loss of reputation (Institute of Directors in Southern Africa, 2009). Examples of reporting risks include the generation of incomplete information, inaccessibility of information, information accessed by unauthorised persons, and the loss of information (Harrer, 2008:79; Jules Halpern Associates, 2010; Cumming, 2012; Cain et al., 2014). Although reporting risks influence access to funding for SMMEs, they are deemed not to significantly influence the sustainability of South African SMMEs (Stainbank, 2008:8).
<u>Compliance risks</u>: Compliance risks are threats and/or opportunities associated with the nonadherence and/or adherence to relevant laws, regulations, policies and procedures (Moeller, 2011; Brockett & Rezaee, 2012; Sales, 2014). These risks are synonymous with those events which lead to legal sanctions, financial loss, or reputational loss as a result of failure to comply with laws and regulations, codes of conduct, and standards of best/good practice (Economic Cooperation Organization Trade and Development Bank, 2007). Examples of compliance risks include violation of local tax laws, violation of employment-related laws, and nonadherence to health and safety rules (Johnson & Johnson, 2013).

The conceptualised definition of South African SMMEs' sustainability (see Section 2.2) is primarily related to their attainment of sound financial performance and financial position. Thus it can the concluded that reporting risks do not significantly influence the sustainability of South African SMMEs, as these risks do not significantly impact the actual financial performance and position of these entities but rather the reporting of the latter phenomena. Stemming from the above, it is highly likely that the sustainability of South African SMMEs is mostly affected by a combination of three types of risks – operational risks, strategic risks and compliance risks – as explained above. This is especially the case since these business entities operate in a 'harsh' economic landscape (see Section 2.3) which significantly impacts the operations of these business entities. It is therefore imperative for every business entity to adequately and effectively manage risks by means of sound risk-management initiatives (Institute of Risk Management, 2002; Haran, 2014:50; Aven, 2016:4).

In laypersons' terms, risk management pertains to a process that economically, effectively and efficiently reduces the potential negative consequences of risks (potential and realised), while simultaneously maximising the potential positive consequences and/or realisation of opportunities, resulting in optimal business sustainability (Bruwer, 2016:63). Taking into account that the term 'risk management' relates to the concept of 'risk', it is not surprising that the term is broadly defined. Therefore, a non-exhaustive list of definitions is provided in Table 2.4 to conceptualise 'risk management' within the ambit of this study.

**Table 2.4:** Non-exhaustive list of definitions of the term 'risk management'

Definition	Reference		
Risk management refers to the understanding, analysing and addressing	Institute of Risk		
of risks, in order to ensure that organisations achieve their objectives	Management (2016)		
The identification, analysis, assessment, control, and avoidance,	Business Dictionary		
minimization, or elimination of unacceptable risks	(2016a)		
Risk management is a customised and systematic process that entails	Bruwer (2016:xix)		
the appropriate identification, analysing and adequate treatment of			
uncertain events in and around an organisation			
Coordinated activities and methods that are used to direct an organisation	Risk Management		
and to control the risks that can affect its ability to achieve its objectives	Dictionary (2016)		
Process of identifying, evaluating and ranking the priority of risks, followed	Cagnin et al. (2016:489)		
by a coordinated and cost-effective application of resources which			
focuses on monitoring and controlling the probability or impact of			
uncertain events			
Risk management is a systematic approach that aligns strategy, people,	Dabari & Saidin		
technology, processes and knowledge with the purpose of assessing,	(2014:629)		
evaluating and managing the risk that an organisation faces			
Process which pertains to the identification of risks, the assessment of	Yusuf & Dansu		
risks and the treatment of risks	(2013:83)		
A scientific process of well-defined, sequential steps that supports better	Endicott & Gardiner		
decision making by providing greater insights into risks and their impacts	(2011:20)		

From the definitions in Table 2.4 above, the inference can be made that risk management pertains to the identification of possible threats and opportunities, in a logical manner, including their analysis, with the main intent to implement ways to treat such threats and/or take advantage of such opportunities. This view is supported by previous research studies (Godbole, 2012; Mwangi, 2014:13; Leverty, 2015) where it was found that business entities which appropriately managed risks tended to achieve their set objectives while allowing them to create value through seizing opportunities and/or mitigating risks. Using the aforementioned as a basis, 'risk management' is conceptualised throughout this research study as follows:

It is the step-by-step process whereby potential threats (and opportunities) are identified and analysed in order to mitigate their realisation and potential impact (spur on their realisation and potential impact), with the main intent to provide reasonable assurance surrounding the attainment of relevant business objectives in the foreseeable future.

Notwithstanding the aforementioned definition, risk management is regulated by professional institutions (e.g. Committee of Sponsoring Organisations of the Treadway Commission (hereafter referred to as COSO), the Institute of Risk Management, and International Organization for Standardization) (PricewaterhouseCoopers, 2009). When focus is shifted on the ISO 31000:2009 standard covering the risk management process, it is mentioned that this process is not a one-

size-fits-all process. This is particularly the case since the size and nature of business entities will always differ from one another, meaning that their risks will be unique (Erard, 2010; Bird, 2016). This risk management process, as per the International Organization for Standardization (2009), is depicted in Figure 2.1.





Communication and consultation: This is deemed the first step in the risk-management process, and it entails the obtaining and exchange of information about possible areas of concern, among others, with the main intent to devise a plan of action to address these areas of concern (Advisory, Conciliation and Arbitration Service, 2016). Within the context of risk management, the first objective of communication and consultation is to identify those who will be involved in the risk-management process (Kanona & Tawalbeh, 2007). Other issues like the risks to be managed, how often they will be managed, benefits of risk management, the media and methods to be used for communication and consultation, should also be clarified (Kanona & Tawalbeh, 2007; PricewaterhouseCoopers, 2015:2; Institute of Risk Management, 2016). In essence, communication and consultation occur at various levels in (and around) a business entity, throughout the entire risk-management process (Broadleaf Capital International, 2014).

- <u>Establishing the context</u>: The next step is to establish relevant contexts pertaining to areas of concern, which entails outlining the parameters and defining the relevant scope of the risk-management process (Kouns & Minoli, 2011). In general, there are three contexts that should be defined: 1) external context (factors outside a business which affect a business entity's ability to achieve its objectives), 2) internal context (factors within a business entity that influence the way its risks are managed), and 3) risk-management context (objectives, resources, scope and limits of risk management activities) (Pojasek, 2013:84, Hopkin, 2014:83; Tsiouras, 2015). In core, establishing the contexts ensures the smooth sailing of other steps in the risk-management process (Broadleaf Capital International, 2014).
- <u>Risk assessment</u>: Although some may consider risk assessment as the most cumbersome activity in the risk management process, it is perhaps the most crucial one (Taroun et al., 2011:87). Risk assessment encompasses three sub-activities: 1) risk identification (identification of threats and/or opportunities and their sources), 2) risk analysis (assessment of probability and consequences of identified risks), and 3) risk evaluation (prioritisation of risks with regard to the set risk criteria) (Institute of Risk Management, 2002:2; International Organization for Standardization, 2009:17; Garrido et al., 2011:243; Renn, 2012; Dumbravă & lacob, 2013:87; Green, 2015:5; Sadgrove, 2016:71; Tulashie et al., 2016:211).

In order to ensure effective and adequate treatment of risks, their probability of materialising and potential impact on objectives should be thoroughly assessed, leading to correct evaluation of identified risks. This can be achieved using a risk matrix as depicted in Figure 2.2.

		Impact>						
		Negligible	Minor	Moderate	Significant	Severe		
Likelihood	Very Likely	Low Med	Medium Med H		High	High		
	Likely	Low	Low Med	Medium	Med Hi	High		
	Possible	Low	Low Med	Medium	Med Hi	Med Hi		
	Unlikely	Low	Low Med	Low Med	Medium	Med Hi		
	Very Unlikely	Low	Low	Low Med	Medium	Medium		

Figure 2.2: Risk matrix (Source: Davies, 2015)

• <u>Risk Treatment</u>: As mentioned above, the treatment of risks stems from the results of risk analysis and risk evaluation. The International Organization for Standardization (2009:18) and Institute of Risk Management (2002:10) define risk treatment as the use of at least one risk-reduction action. In essence, it encompasses three activities: 1) identification of risk treatments (all possible available treatments are identified), 2) evaluation and selection of risk-treatment options (cost-benefit analysis of all identified treatments – option and selection thereof), and 3) preparing and implementing and/or applying the treatment option(s) (Rodríguez-Perez, 2012:76; Smit, 2012:282; Reniers & Van Erp, 2016:70). With regard to risk treatments, risks can be avoided, transferred, mitigated or tolerated, based on their probability of materialising and potential impact (Mishra, 2015; Sadgrove, 2016:71). To better understand the linkage between the abovementioned risk treatments and the risk matrix (see Figure 2.2), Figure 2.3 shows the four treatment options in relation to probability and impact of evaluated risks.

d
re
High impac w probability

Figure 2.3: Four risk treatment options (Source: Sadgrove, 2016:211)

Monitoring and review: Monitoring refers to the supervision of activities to ensure they are completed timely and meet the set objectives (*Business Dictionary*, 2016b), while review refers to the analysis of activities to determine if they are still functioning/operating as intended (Refsdal et al., 2015:23). Constant monitoring and review will enable the timeous detection of changes in risk management activities (Refsdal, et al., 2015:24). Furthermore, the monitoring and review of the risk treatment option(s) implemented will avoid control failure, thereby ensuring that risk levels are always within acceptable margins. In essence, review of all the risk-management activities is a prerequisite for effective risk management, as it ensures that risks are effectively identified and assessed and that appropriate controls and responses are

in place (Institute of Risk Management, 2002:11). During monitoring and review, new risks can be identified, and these new risks will go through the whole process from the beginning (i.e. communicated to all stakeholders, assessment and treatment) (Refsdal, et al., 2015:24).

Though International Organization for Standardization's risk management process is both universal and used as foundation in many business entities across the globe (Tsiouras, 2015), there are two main ways in which risk management can be approached.

## 2.5 RISK-MANAGEMENT APPROACHES

In literature, two common approaches to risk management exist, namely, risk management in silos (traditional approach) and ERM (Hohenwarter, 2014). In order to understand their derivation, a brief historic overview is provided on each approach below:

<u>Traditional approach</u>: Prior to the development of the holistic<sup>23</sup> management of risks in 1992, risk management was largely focused on the management of 'pure risks'<sup>24</sup> – risks that presented possibilities of financial losses (Simona-Iulia, 2014:277; Society of Actuaries, 2016). In this timeframe, pure risks were also viewed as unrelated to one another (Dabari & Saidin, 2014:628), as it was common practice for them to be managed in silos (i.e. per team, per department, per function). This approach was regarded as adequate between the early 1960s and mid-1970s, especially since most of the international companies had satisfactory financial results (Bruwer, 2016:38).

With the passing of time, during the late 1970s, more risks (mostly financial risks) started to materialise, and this attracted the attention of many stakeholders in business entities across the globe (Razali & Tahir, 2011:9; Braine, 2015). As a result, the management of risks in silos proved inadequate (Walker & Shenkir, 2008:31; Godbole, 2012; Simona-Iulia, 2014:276), mainly owing to the realisation of financial risks (Lam, 2014) thathad an adverse influence on the sustainability of many well-established business entities (Krosinsky, 2013). This resulted

<sup>&</sup>lt;sup>23</sup> Holistic risk management refers to the management of risks, across the entire organisation, while simultaneously taking into consideration their relationships among other identified and/or unidentified risks (Caroll, 2010:9).

<sup>&</sup>lt;sup>24</sup> A pure risk refers to an uncertain event that presents an opportunity for loss but not for gain (International Risk Management Institute, 2016).

in the questionability surrounding the effectiveness of risk management in silos (Wolf, 2008:20).

In addition to its inability to deal with the expansion of risks, the silo approach focused on individual departments, resulting in other departments being more efficient than others, but at the expense of the overall performance of the business (Lewis, 2012:141; Chiu, 2015:91). Furthermore, independence between the departments resulted in the inefficient use of resources (Lam, 2014:10; Zhao et al., 2015:38). In essence, it is after the collapse of some big businesses that the business world, together with regulators, realised the need for a more rigorous way to manage risks; an approach that was more proactive, unlike the silo approach that is reactive (ACI Worldwide Inc., 2011; Meyer et al., 2011:366; Hohenwarter, 2014; Bromiley et al., 2015:265; Pezzuto, 2016:332).

ERM approach: With the need to better manage risks holistically, and to manage them from a different perspective, ERM emerged (Hohenwarter, 2014). Although the term 'ERM' became popular in 2004, the concept of managing risks holistically had started in 1992 (Graham, 2015; Magan, 2016). The plethora of risks that bombarded business entities between 1970 and 1985 (Beck & Kewell, 2014:55; Shenkir & Walker, 2011:6; Bruwer, 2016:39) necessitated a change in the way risks were managed. In 1985, an organisation by the name of COSO was formed by the Treadway Commission, and one of its core objectives was to provide guidance on the holistic management of risks in businesses (Hayne & Free, 2014:311; Bruwer, 2016:39). COSO developed a framework known as the COSO Internal Control – Integrated Framework of 1992 (Gelinas et al., 2014:232), which became a recognised framework used to holistically address risks through the implementation of a sound system of internal control.

Although there was now a recognised framework that could assist business entities to manage risks holistically, its adequacy was still questioned (Moeller, 2013; Stamler et al., 2014:178). In 2001, after the scandals of business entities such as Enron, COSO, in partnership with PricewaterhouseCoopers, embarked on a process to develop a new framework to adequately manage risks (Steinberg et al., 2004:v; Bruwer, 2016:58). The COSO Internal Control – Integrated Framework of 1992 was then used as the foundation for the new framework, the COSO Enterprise Risk Management: Integrated Framework (hereafter referred to as COSO ERM Framework), which was introduced in 2004 (Stamler et al., 2014:178). Apart from being robust, ERM included 'opportunities' in the definition of risks, unlike the Silo approach that

only considered threats (Hampton, 2014:77). Over the years, the popularity of ERM has continued to grow (Hoyt & Liebenberg, 2011:795; Shenkir & Walker, 2011:7), especially because of its thorough approach. Hence it is of no surprise that ERM has become the new minimum standard to manage risk in and around business entities (Rao, 2007:170; Krause, 2015:35).

From the historic overview above, it is apparent that the ERM approach manages risks from a broad and integrated perspective, while simultaneously considering both positive and negative risks which may have an influence on the achievement of business objectives (Okhahlamba Local Municipality, 2013:5; Hohenwarter, 2014; Simona-Iulia, 2014:277; Beasley et al., 2017:1). According to previous research studies (Smit, 2012:118; Dornberger et al., 2014:1; Ashkenas, 2015), although some business entities still make use of the traditional approach to manage risks, most business entities favour managing risks in a holistic manner. This is particularly beneficial, since an ERM approach to risk management assists with, inter alia, 1) compliance with applicable laws, 2) management of multiple and cross-enterprise risks, and 3) minimisation of possible financial risks (Havenga, 2006:21)

The ERM approach to risk management has been found to have a positive influence on the sustainability of business entities (Dornberger et al., 2014:2), particularly since it provides a greater sense of reasonable assurance surrounding the attainment of relevant business objectives (Rao, 2007:168; Mandel, 2008:10; Shortreed, 2010:10). Notwithstanding the above, relevant ERM frameworks were developed for an array of business entities subsequent to the introduction of the COSO ERM Framework (Gilbert, 2007:3). Two ERM frameworks are discussed within the ambit of SMMEs below.

## 2.6 ENTERPRISE RISK-MANAGEMENT FRAMEWORKS FOR SMMES

A risk-management framework refers to a set of initiatives that support and sustain the riskmanagement process (see Section 2.4) (International Organization for Standardization, 2014). As previously mentioned, there exist numerous ERM frameworks which business entities across the globe can use (Olson & Wu, 2008:3). From a South African SMME perspective, however, there exist two major risk management frameworks (models), the SME Risk Architecture Model and the COSO ERM Framework. These frameworks are discussed at length overleaf.

## 2.6.1 SME Risk Architecture Model

The SME Risk Architecture Model was developed in 2012, and states that risks are best managed (through the adoption of an ERM approach to risk management) through three phases, namely 1) SME risk consciousness, 2) SME risk-management process, and 3) SME risk-management framework (Smit, 2012:253). The SME Risk Architecture Model is depicted in Figure 2.4.



Figure 2.4: SME Risk Architecture Model (Source: Smit, 2012:255)

Each of the three phases is briefly explained below:

- <u>Phase 1: SME risk consciousness</u>: Firstly, SMMEs should identify relevant sources of risks which may impact on their overall sustainability. When identifying the risk sources, both the internal and external risk areas should be considered. Since the identified risk sources will be transferred to the risk-management process (Phase 2), this phase should be conducted thoroughly as an unidentified risk source result in risks that will not be managed.
- <u>Phase 2: SME risk-management process</u>: The SME risk-management process has four initiatives: 1) risk context and strategy (understanding of business's internal and external environment; defining of objectives, risk criteria and resources required for risk-management process), 2) risk decisions (identification of risks, followed by their analysis, evaluation, and treatment), 3) communication (gathering and sharing of relevant information), and 4) monitoring, review and continuous improvement (checking if all risk-management activities are being carried out effectively).

<u>Phase 3: SME risk-management framework</u>: The SME risk-management framework is basically the synergy of the first two phases, and constitutes: 1) planning (establishment of business and risk-management objectives, and the determination of the sources of risks), 2) implementation (implementation of the risk-management process, which includes risk identification, analysis and evaluation, among others), 3) results (recording of key performance indicators in respect of the achievement of business objectives), and 4) measurement (assessment of the effectiveness of the risk-management process in achieving business objectives). The relationship of the above-mentioned initiatives is depicted in Figure 2.5 below.</u>



Figure 2.5: SME Risk Management Framework (Source: Smit, 2012:289)

Although the SME Risk Architecture Model is comprehensive in its coverage, while also being tailor-made for South African SMEs, its effectiveness is yet to be determined by empirical research studies. For this reason, this model was not used as basis for this research study.

## 2.6.2 COSO Enterprise Risk Management: Integrated Framework (COSO ERM Framework)

According to COSO, ERM can be defined as follows:

[It] is a process, effected by an entity's board of directors, management and other personnel, applied in strategy setting and across the enterprise, designed to identify potential events that may affect the entity, and manage risk to be within its risk appetite,<sup>25</sup> to provide reasonable assurance regarding the achievement of entity objectives (Steinberg et al., 2004:2).

The COSO ERM framework comprises eight components (Hillson, 2003:141-143; Liebenberg & Hoyt, 2003:40; Steinberg et al., 2004:4; Broadleaf Capital International, 2007:4; Cox, 2007:4; Gilbert, 2007:3; Harvey, 2008:8; Olson & Wu, 2008:37; Walker & Shenkir, 2008:32; Payne, 2009:28; Brandt, 2010:26-27; Shaw, 2010:20; Olson & Wu, 2010:696; ACI Worldwide Inc., 2011; Beaumier & DeLoach, 2011:46; Diaz-Garrastacho & Dickins, 2011:61; Hyden, 2011:100-101; Shahzad et al., 2011:2072; Chew, 2012; Gates et al., 2012:31; Smit & Watkins, 2012:6328; Curkovic et al., 2013:16; Hayne & Free, 2014:311; Lundquist, 2014:169; Rubino & Vitolla, 2014:321-322; Agrawal, 2016:121-122; Bruwer, 2016:59; Moeller, 2016:157), as depicted in Figure 2.6, with each component expanded on overleaf for the sake of clarity.



Figure 2.6: The COSO ERM Framework (Source: Steinberg et al., 2004:5)

<sup>&</sup>lt;sup>25</sup> Risk appetite refers to the level of risk an organisation's management is willing to accept (McKay, 2016:61).

- <u>Internal environment</u>: Internal environment refers to a business's attitude towards risk and risk management, and it comprises the tone at the top, risk appetite, integrity and ethical values. In essence, the internal environment relates to control environment.<sup>26</sup> As ERM requires a top-down approach, a strong internal environment can be achieved if top management reflects transparency, honesty, integrity, ethical behaviour, and commitment to continuous improvement, to note but a few. In essence, internal environment is regarded as the cornerstone of ERM.
- <u>Objective setting</u>: Based on the conceptualised definition of risk (see Section 2.4), it is evident that if business objectives are not clearly defined, the risk-management process may very likely be ineffective and futile. All business objectives should be well defined at all levels, from entity to subsidiary level, to ensure that all possible risks are dealt with. Furthermore, objectives should be classified under the following areas: strategic, operations, reporting, and compliance. This will facilitate the grouping of risks into the four major risk categories. Also during this stage, business entities should define their risk-appetite levels.
- Event identification: Event identification can be used in place of risk identification, and relates to risk identification as per ISO 31000 (see Section 2.4.1). Both internal and external risks should be considered, and there should be a clear distinction between threats and opportunities. All the levels (i.e. business unit, division) should be considered when identifying the events that might impact the attainment of objectives. With the current volatile business environment (see Section 2.3), risks keep on increasing and changing, requiring business entities to regard event identification as a continuous process. Furthermore, interdependencies between risks should also be noted at this stage. In essence, failure to identify some risks may result in the business suffering loss or missing an opportunity; hence this stage should be done thoroughly.
- <u>Risk assessment</u>: This stage relates to risk assessment and evaluation under ISO 31000 (see Section 2.4.1). At this stage, risks are analysed and evaluated, taking cognisance of the likelihood and impact of the identified risks. Stemming from the scientific definition of risk, 'Risk = Impact × Probability', the impact and likelihood values should be used in risk ranking

<sup>&</sup>lt;sup>26</sup> The control environment is a collection of expected norms, expected qualities and expected actions that should be in place to provide an important foundation for attaining a sound level of internal control in a business entity (Bruwer, 2016:80).

(see Figure 2.2). Technological advances have made risk quantification and analysis less cumbersome, as there are various techniques that can be used to determine the significance of risks. Risks that are unlikely to materialise, and with insignificant consequences, pose no immediate danger to a business, while those risks with a high likelihood and moderate to catastrophic consequences should be given first priority when determining risk responses

- <u>Risk response</u>: This relates to the manner in which risks are treated (see Section 2.4.1). Once identified and analysed, risks need to be managed effectively if their effects are to be minimised. This is where management develops actions to reduce risks until they are within an acceptable level. Not all risks must be responded to, as some might have to be tolerated, depending on the probability of occurrence and the impact associated with the identified risks. In essence, the possible risk treatments are: accepting, avoiding, sharing and controlling.
- Control activities: The best controls should be selected to deal adequately with the identified risks. With regard to the treatment of risks, there are three types of controls a business entity can implement: preventative controls,<sup>27</sup> detective controls<sup>28</sup> and corrective controls.<sup>29</sup> Furthermore, control activities are often in the form of segregation of duties, proper authorisation activities, adequate document usage and design, safeguarding of assets, and independent checks. Management should implement controls that ensure that the residual risk is within the business entity's risk appetite, and the cost is reasonable in relation to the benefit. These controls need to be communicated and understood for them to be effective. To ensure effectiveness and operating efficiency of established controls, they must be evaluated regularly through periodic assessments and monitoring.
- Information and communication: This relates to communication and consultation of ISO 31000 (see Section 2.4.1). Information and communication affect all other activities, as there should be constant interaction throughout the ERM process. Relevant information should be gathered, stored and disseminated to relevant people on time, allowing them to achieve their individual objectives. It is therefore important for a business entity to have a dedicated

<sup>&</sup>lt;sup>27</sup> Preventative controls are those controls which aim to ensure that a risk does not materialise (Hernandez, 2016:646).

<sup>&</sup>lt;sup>28</sup> Detective controls are those controls which ensure that risks are timeously noticed once they materialise (Hernandez, 2016:646).

<sup>&</sup>lt;sup>29</sup> Corrective controls are those controls which aim to remedy situations after risks have materialised (Whittington, 2015).

information system that can provide information timeously to all levels, in the format requested. For effective communication to exist there should be a free flow of information vertically,<sup>30</sup> and laterally.<sup>31</sup> Furthermore, constant communication with external parties is also vital. The quality of decisions made is greatly affected by the effectiveness of communication in a business. Thus, for an effective and quality decision-making process, a business entity should ensure that there is effective communication.

<u>Monitoring</u>: All the activities of ERM should be monitored through management activities – separate evaluations or both. This entails the measurement of a business entity's performance to determine if it meets the set standards. All the stages in the risk- management process, together with their respective outputs, should be monitored effectively to ensure that maximum value is derived from this process. The controls implemented also need to be monitored to determine if they are operating as intended. In addition, management should also ensure that adequate resources, as defined in the risk treatment plans, have been allocated. In essence, it is vital to monitor risk-management performance to ensure the overall effectiveness of implemented risk-management policies and processes.

If implemented correctly, the above framework will enhance the sustainability of a business entity, as it enhances the achievement of business objectives (McNally & Tophoff, 2015:30). In this study, the COSO ERM Framework was used as the foundation to address the identified research problem (see Section 1.3.1) for the following four reasons: 1) it is the most widely used ERM framework around the world (Hopkin, 2014:62), 2) it applies to businesses of all sizes, regardless of the industry in which they operate (Weller, 2015), 3) it is the most common ERM framework in South Africa (Institute of Risk Management South Africa, 2014), and 4) it was the first standardised framework which business entities used to benchmark their respective risk-management initiatives (Steinberg et al., 2004:v).

Despite the fact that COSO's ERM Framework is used by most businesses worldwide (Beasely et al., 2010; Moeller, 2013), research shows that South African SMMEs make limited use of it (Knox, 2012).

<sup>&</sup>lt;sup>30</sup> Vertical communication refers to the flow of information from senior management to lower-level employees (topdown), and from lower-level employees to senior management (bottom-up) (Lussier, 2008:344).

<sup>&</sup>lt;sup>31</sup> Lateral communication refers to interaction between employees of the same level within a business entity (Dwyer, 2012:150).

#### 2.7 RISK MANAGEMENT IN SOUTH AFRICAN SMMEs

Notwithstanding that risk-management processes should be thorough and holistic, previous research studies (Institute of Directors in Southern Africa, 2010; Smit & Watkins, 2012:6326; Vadiveloo & Parsa, 2012) show that risk management in South African SMMEs is generally carried out solely by owners/mangers of these entities owing to the limited resources at these business entities' disposal. This suggests that risk-management initiatives deployed by these business entities may not be effective and/or adequate as the owners/managers often lack the requisite risk management skills. (Janakiraman, 2015). Despite the above, previous studies (Verbano & Venturini, 2013:187; Aren & Sibindi, 2014:96; Ngary et al., 2014:919; Sunjka & Emwanu, 2015:1469; Siwangaza et al., 2014:170; Aziz & Yazid, 2015:297; Bruwer, 2015:58; Chakabva, 2015:53; Enow & Kamala, 2016:233) do however indicate that South African SMMEs make use of customised risk- management initiatives to manage operational risks, which include, inter alia, budget controls, code of ethics, cash counts, access controls to entrances, limited access to safes, CCTV camera usage, reconciliations, staff supervision, and segregation of duties.

A thorough investigation of various research-related databases revealed only a few previous studies conducted in relation to the utilisation of ERM in South African SMMEs. In general, South African SMMEs do not give ERM the attention it ought to be given (Payne, 2010:23). This view is supported by the fact that most South African SMMEs unknowingly engage in ERM initiatives, without having formally implemented any ERM framework (Ayinde et al., 2008:217; Masama et al., 2012:11786; Smit, 2012:20; Vadiveloo & Parsa, 2012; Chakabva, 2015:65; Bruwer & Siwangaza, 2016:112). In essence, the risk-management initiatives utilised by South African SMMEs are, most of the time, relevant to the COSO ERM Framework in the sense that they relate to the setting of clear objectives and the right tone at the top, the communication of risks, and the monitoring of risks (Masama et al., 2012:11785; Bruwer et al., 2013:1023; Chakabva, 2015:53).

With regard to the maturity of risk management, there are four levels which business entities pass through: 1) non-existent/ad hoc (only responds after something has gone wrong – known as reactive management), 2) internal control only (use of formal internal controls to manage risks), 3) stand-alone risk management and internal control (risk-management process does not function in conjunction with a business's management system), and 4) integrated risk- management process (risk management is an integral part of a business's management system) (McNally &

38

Tophoff, 2015:31). With regard to the risk-management levels above, in conjunction with the aforementioned risk-management initiatives used by South African SMMEs, the inference can be made that most South African SMMEs are at level two – the use of internal controls to manage risks. This view is affirmed by Smit and Watkins (2012:6328) when stating that risk-management initiatives of South African SMMEs tend to focus on the use of controls to minimise mostly financial losses. Although South African SMMEs have formal controls in place, these controls are perceived to be ineffective and/or inadequate to manage a plethora of risks hindering these business entities (Bruwer & Coetzee, 2016:207). In essence, it appears that South African SMMEs do not make proper use of ERM to identify and manage risks (Duong, 2009:22; Chakabva, 2015:59).

#### 2.8 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY

In this chapter, relevant concepts introduced in Chapter 1 were expanded upon. Furthermore, relevant discussion took place to guide this study, particularly surrounding: 1) the sustainability of South African SMMEs, 2) the economic factors affecting South African SMMEs, 3) risk and risk management, 4) risk-management approaches, 5) enterprise risk management, and 6) risk management in South African SMMEs. Moreover, relevant terms were also conceptualised within the ambit of this research study for the sake of clarity.

Despite their immense contribution to the national GDP and to employment creation, South African SMMEs' sustainability is considered among the worst in the world. Although sustainability is generally related to the attainment of three core objectives (economic, social, and environmental), in this chapter, SMME sustainability was conceptualised as a phenomenon relating to the long-term existence of these entities and their achievement of socio-economic objectives. Unfortunately, research shows that at least 75% of South African SMMEs fail in the first 42 months of operation, thereby implying that these business entities are not able to fulfil their socio-economic objectives. The weak sustainability of South African SMMEs is believed to be adversely influenced by a 'harsh' economic environment, as influenced by an array of macro-economic landscape of South African SMMEs was found to be 'harsh' – as supported by: 1) a GDP mediocre net increase of 3.70% (US\$12 billion) between 2013 and 2016, 2) a GDP per capita net decrease of 2.04% (US\$124), 3) an unemployment rate net increase of 4.11% between 2013 and 2016, 4) a deterioration in the Gini index, 5) an inflation rate net increase of 14.21%

between 2013 and 2016, and 6) a prime interest rate net increase of 23.53% between 2013 and 2016. In essence, this economic landscape can be referred to as the 'breeding ground' for risks that influence the sustainability of South African SMMEs.

Stemming from the abovementioned adverse economic landscape, South African SMMEs were perceived to be affected by a range of risks which pertained to operational risks, strategic risks, and compliance risks. In order to manage these risks adequately, the two ERM frameworks available to South African SMMEs are the SME Risk Architecture Model, and COSO ERM Framework. In order to understand the risk-management practices of South African SMMEs, the risk-management initiatives used by South African SMMEs were investigated. According to scholarly literature, implemented risk-management initiatives by these business entities are customised, ineffective and/or inadequate. Although these risk-management initiatives mostly relate to ERM initiatives, it is perceived that these initiatives are used by chance as opposed to choice.

In the next chapter, the research design, research methodology and research methods used in this research study are discussed.

## CHAPTER 3

# **RESEARCH DESIGN, RESEARCH METHODOLOGY AND RESEARCH METHODS**

## SYNOPSIS

For this research study, a methodical process was followed whereby each chapter built on the previous chapter. This process is depicted below.

Chapter 1: Introduction to the research study

Chapter 2: Literature review

**Chapter 3**: Research design, research methodolody and research methods

- Introduction
- •Research design and research methodology
- Research methods
- Ethical consideration
- Research limitations
- Data validity
- Data-collection tool
- ·Conclusion and relevance to the research study

Chapter 4: Data analysis, results and discussion

Chapter 5: Conclusion

### **3.1 INTRODUCTION**

In Chapter 1, the research problem, relevant research questions and research objectives pertaining to this research study were identified, while in Chapter 2, a thorough literature review was conducted in respect of the identified research problem. In this chapter, emphasis is placed on the research design, research methodology and research methods deployed in this research study. Hence, for the remainder of this chapter, discussion takes place under the following headings: 1) research design and research methodology, 2) research methods, 3) ethical considerations, 4) research limitations, 5) data validity, and 6) data-collection tool.

## 3.2 RESEARCH DESIGN AND RESEARCH METHODOLOGY

A research design can be deemed a 'building plan'; as it provides direction on how to conduct a research study (Mouton, 2001:55; Hair et al., 2003:57). Moreover, a research design should ensure that all relevant research questions are fully addressed (Hair et al., 2003:57; Durrheim, 2006:34; Saunders et al., 2012:159), thereby ensuring accurate testing of the research proposition. The research design of any research study can be categorised in terms of: 1) type of research study (empirical versus non-empirical), 2) source of data (primary data versus secondary data), 3) nature of data (numerical versus non-numerical), and 4) level of control (structuring of the data-collection tool) (Mouton, 2001:144). The research design used in this study is described below, under the aforementioned categories:

- <u>Type of study</u>: Although this study consisted of both empirical and non-empirical research, it was predominantly empirical in nature through the collection of primary data, with a small part being non-empirical, taking the form of a literature review (see Chapter 2).
- <u>Source of data</u>: Both primary and secondary data were obtained in this research study. The
  primary data obtained stemmed from responses gathered from respondents (SMME owners
  and/or managers) through survey research, while the secondary data were gleaned from the
  review of literature; applicable to the specific research problem (see Chapter 1).
- <u>Nature of data</u>: The primary data collected were predominantly numerical, and the survey responses were coded<sup>32</sup> in order to analyse them statistically. All secondary data were predominantly non-numerical in nature (see Chapter 2).

<sup>&</sup>lt;sup>32</sup> Coding refers to the assignment of numbers to individual responses in a dataset, to enable a computer to make sense of relevant collected data (Babbie & Mouton, 2001:412).

<u>Level of control</u>: Although there was no control exercised over secondary data; though a suitable level of control was exercised during the collection of primary data. The purpose of the control was to manage errors in the data collected; this took the form of using a structured questionnaire,<sup>33</sup> piloting the structured questionnaire, and validating responses provided by respondents.

Using the above as foundation, the research design deployed in this research study was that of survey research. Survey research pertains to the collection of data from a population or sample through means of a questionnaire, with the aim of describing, exploring, and/or explaining a construct of interest (Burton, 2007:4).

Notwithstanding the above, the research methodology refers to the steps which need to be followed to realise a particular research design (Saunders et al., 2012:160). In essence, there are three options available with regard to the research methodology of any study: 1) quantitative research methodology, 2) qualitative methodology, and 3) mixed-methods research methodology (i.e. both qualitative and quantitative) (Edmonds & Kennedy, 2012:3; Creswell, 2014:3). Considering that the researcher is a proponent of the positivistic research paradigm,<sup>34</sup> a quantitative research methodology was deemed appropriate for this research study (Leedy & Ormrod, 2005:179). According to McKimm et al. (2017:7), quantitative research involves the selection of research subjects (samples) and entails the collection of primary quantitative (numerical) data through means of using relevant data-collection tools. In particular, survey research was conducted (Saunders et al., 2007:139; Singh & Bajpai, 2007:169; Gropper & Smith, 2008:567; Saunders et al., 2012:177) through means of a questionnaire tool (Picciano, 2016:55).

### **3.3 RESEARCH METHODS**

Research methods refer to the various procedures performed by a researcher in order to collect, analyse and interpret data (Creswell, 2014:17), subsequently supporting the relevant research methodology (Collis & Hussey, 2009:73). Research methods can be grouped into four categories:

<sup>&</sup>lt;sup>33</sup> A structured questionnaire refers to a set of questions that has a limited number of possible responses (Zikmund et al., 2013:194).

<sup>&</sup>lt;sup>34</sup> Positivistic research entails the collection of data in order to gain greater insight into certain variables evident in society. The results from data analysis are generally used to support or refute the formulated hypothesis concerning such researched variables (Collis & Hussey, 2009:58; Creswell, 2014:7).

1) sampling methods, 2) data measurement, 3) data-collection methods, and 4) data-analysis methods (Babbie & Mouton, 2001:75). The research methods deployed in this research study, as categorised, are discussed below:

- <u>Sampling methods</u>: As previously mentioned (see Section 1.4), the target population of this research study constituted members of management of South African fast-food SMMEs. As the exact number of these business entities was unknown, non-probability sampling techniques were used, specifically a mixture of the purposive sampling method<sup>35</sup> and the convenience sampling method,<sup>36</sup> as supported by the following relevant delineation criteria:
  - All respondents had to be regarded as owners and/or managers of their respective SMMEs.
  - All SMMEs should have been regarded as fast-food<sup>37</sup> enterprises.
  - All SMMEs should have been regarded as non-franchised business entities.
  - All SMMEs should have adhered to the formal definition of an SMME as per the Small Business Act No. 102 of 1996.
  - All SMMEs should have been regarded as either micro enterprises or small enterprises.
  - All SMMEs should have employed fewer than 50 full-time employees.
  - o All SMMEs should have been deemed either sole traders or partnerships.
  - All SMMEs should have been in existence for at least two years.
  - All SMMEs should have operated within the perimeters of the Cape Peninsula.

The student selected a sample of SMMEs that were conveniently reachable, and they had to satisfy the above delineation criteria. The selected sample size was 150, of which 130 responded favourably, constituting a 86.66% response rate. However, of the 130 responses received, only 116 were regarded as valid.

 <u>Data measurement</u>: In order to address all posed research questions (see Section 1.3), the most suitable data measurements<sup>38</sup> were identified and implemented in the questionnaire tool. Since perceptions were obtained from South African SMME owners and/or managers, the

<sup>&</sup>lt;sup>35</sup> A sampling method where respondents are selected based on certain criteria to ensure better control of significant variables (Sharma, 1997:123; Burton, 2007:13).

<sup>&</sup>lt;sup>36</sup> A sampling method where respondents are selected based on their availability and/or proximity to a researcher (Ruel et al., 2016:150).

<sup>&</sup>lt;sup>37</sup> Only SMMEs operating in the fast-food industry were considered in this study.

<sup>&</sup>lt;sup>38</sup> There are four data types: nominal data (variables that can only be counted), 2) ordinal data (variables that can be ordered), 3) interval data (grouped variables that can be ranked), and 4) ratio data (variables where zero has an actual meaning) (Trochim, 2006; Phillips & Phillips, 2016:107).

data collected from respondents were mostly ordinal in nature, with a few responses providing ratio data and nominal data.

- <u>Data-collection methods</u>: Although questionnaires and/or interviews can be used to obtain data through survey research (Sincero, 2012), only a questionnaire consisting mainly of closed-ended questions was used for this research study. After the questionnaire was piloted, it was self-administered by the researcher to the respondents. The questionnaire comprised five sections (see Section 3.6), all with the main intent of addressing the research questions applicable to this research study (see Section 1.3). All collected data were captured and coded accordingly.
- <u>Data analysis method</u>: By using IBM<sup>®</sup> SPSS Statistics<sup>®</sup> software, relevant descriptive and inferential statistical analysis were performed (Trochim, 2006). In relation to descriptive statistics, frequency distribution tables and graphs were mostly generated from the collected data, while the inferential statistics pertained to that of exploratory factor analysis (see Chapter 4).

# 3.4 ETHICAL CONSIDERATIONS

Ethical considerations are a set of standards of behaviour which should provide guidance to a researcher in relation to his/her rights and those of his/her research subjects when conducting his/her research study (Saunders et al., 2012:226). In essence, ethical considerations should be considered throughout the duration of a research study (Anderson, 2009:79). For this research study, the following ethical considerations were taken into account (Anderson, 2009:79-85; Collis & Hussey, 2009:45-47; Cooper & Schindler, 2011:33-49; Zikmund et al., 2013; UK Data Archive, 2017):

 <u>Quality of data</u>: To obtain meaningful results from data analysis, the researcher should ensure that collected data are of good quality (in relation to accuracy, completeness, reliability, etc.). Firstly, the researcher ensured that the data gleaned were of good quality, by selfadministering the questionnaire. Since some of the questionnaires were completed in the absence of the researcher, data quality was further ensured by calling some SMMEs to confirm if indeed they had completed the questionnaire (validation). Lastly, the researcher reviewed each of the received questionnaires to determine if all questions were correctly and completely answered.

- <u>Deception</u>: Deception is when the truth is compromised during respondents' briefing on a study. In laypersons' terms, researchers should not withhold any important information or purposely lure potential respondents to participate in the study. In this study, the researcher did not withhold any relevant information (i.e. purpose of study, how the responses would be used) from respondents, while simultaneously not falsely promising anything in return for participating in the research study.
- <u>Voluntary participation</u>: All respondents should participate in a study voluntarily, without the use of undue influence from the researcher. All respondents were informed that partication in this research study was voluntary, and that they could withdraw from the research study at any time without recrimination. There also was no coercion of respondents to participate (see 'Deception', above).
- <u>Informed consent</u>: The researcher should explain the procedures of the study, thereby enabling respondents to make informed decisions. Before respondents voluntarily took part in this research study, they were informed of what the research study entailed. In addition, consent was provided by each respondent prior to participating in the research study.
- <u>Confidentiality and anonymity</u>: Anonymity entails the non-identification of respondents with their opinions on the questionnaire, while confidentiality entails the non-disclosure of respondents' sensitive information. Respondents were guaranteed that all information provided by them would only be used for research purposes, and would not be shared with anyone. In addition, the anonymity of individuals who completed the questionnaire was guaranteed by not requesting their personal details (i.e. name and surname).
- <u>Protection from harm</u>: If any research study has the possibility of causing harm to the researcher and/or respondents, relevant controls should put in place to manage it. In this study, since a questionnaire tool was used to obtain data, the researcher and all respondents were protected from harm. In addition, to ensure protection from harm for both researcher and research subjects, the research proposal was approved by the Research Ethics Committee of the Faculty of Business and Management Sciences at CPUT, and ratified by the Higher Degrees Committee.
- <u>Fair and accurate reporting</u>: Researchers should not tweak results in order to present a favourable outcome to identified research problems and/or developed hypotheses all results should be reported accurately. For this research study, after data were captured, they were analysed accordingly, while all relevant results were reported on as truthfully as possible.

### **3.5 RESEARCH LIMITATIONS**

Research limitations are defined by Baltimore County Public Schools (2015) as influences that restrict a research study, and are beyond the control of a researcher. The limitations pertaining to this study are briefly discussed below:

- Although the results were generalised to fast food South African SMMEs, only fast-food SMMEs in the Cape Peninsula were considered during data collection. Nevertheless, focusing on fast-food SMMEs is reasonable since the fast-food industry contributes immensely to the South African economy (Stofile et al., 2011:1; Maumbe, 2012:148), contributing approximately R170 billion to the national economy in 2015 (Murray, 2017).
- The SMMEs included in the study are from the Cape Peninsula, in the Western Cape. According to Statistics South Africa (2016), the Western Cape province is among the top three provinces that significantly contribute to the economy of South Africa. With the Cape Peninsula regarded as one of the major contributors to the economic growth of the Western Cape province, it stands to reason that this area also has a large number of SMMEs operating within its borders (Van Weele & Maree, 2013:8; Western Cape Government Provincial Treasury, 2015:114). In addition, the Cape Peninsula is a major tourist attraction, both provincially and nationally (Western Cape Government Provincial Treasury, 2015:67; SA Places, 2016), thereby translating to a significant number of fast-food outlets.
- The number of SMMEs in the Cape Peninsula was unknown, making it difficult to support the size of the sample used in this study statistically. However, with quantitative research it is generally agreed that basic statistical analysis requires a minimum of 30 responses to obtain results that are statistically significant (O'Leary, 2005:89). Therefore, a sample size of 116 respondents, all of whom adhered to strict delineation criteria, was regarded as sufficient to provide statistically significant results for this study.

## 3.6 DATA VALIDITY

According to Heale and Twycross (2015:66), validity refers to the extent to which a construct is correctly measured. In this study, two validity measures were used, and these are content validity and construct validity (Drost, 2011:114-117). Content validity relates to the appropriateness and adequacy of the content used in relation to the construct under study (Bernard, 2000:47; Krishnaswamy et al., 2009:265); while construct validity relates to whether the content truly

measures the construct under study (Goodwin & Goodwin, 2016:103). Content validity was partially enhanced by ensuring that research questions and objectives were clearly articulated, since these were the foundation of questions in the collection tool. In addition, a thorough literature review was conducted, and main concepts within the construct under study were conceptualised. Consequently, the literature review formed the basis for the variables used in the collection tool. Regarding the questions, the researcher ensured that all questions were clear, simple and unambiguous. Finally, the collection tool was piloted by the researcher to determine if respondents could easily answer the questions. Construct validity was enhanced by consulting with senior researchers in the field of risk management, and they ascertained that the content indeed measured the construct under study.

## 3.7 DATA-COLLECTION TOOL

As previously mentioned (see Section 3.3), a questionnaire tool was used to obtain data from respondents. The questionnaire tool used consisted of five sections (A to E), comprising mostly closed-ended questions. For the sake of clarity, each section in the questionnaire tool is discussed at greater length below:

### 3.7.1 Questionnaire tool: Section A

This section consisted of questions about the demographic characteristics of the SMMEs and the respondents. Basically, the questions in this section were used to validate each completed questionnaire (see Table 3.1).

Question number	Question type	Question				
A1	Fill in the blank (Nominal)	In what industry does your business operate?				
A2	Fill in the blank (Ratio)	How long has your business been in existence? (years)				
A3	Fill in the blank (Ratio)	How many employees do you employ? (number)				
A4	Multiple Choice	Are you part of a franchise?				
A5	Multiple choice	Are you the owner, manager, owner and manager, or other?				

**Table 3.1:** Questions in the questionnaire tool – Section A

### 3.7.2 Questionnaire tool: Section B

The objective of this section was to understand the economic factors that affect the sustainability of South African SMMEs. The questions in this section identified the various factors perceived to

affect the sustainability of South African SMMEs. A four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree) was used, and the questions started with the following base statement: *The following barriers affect my business's sustainability* (see Table 3.2).

Question number	Question type	Question					
B1	Likert scale	Food prices of suppliers affect my business's sustainability.					
B2	Likert scale	Unemployment rates affect my business's sustainability.					
B3	Likert scale	Inflation rates affect my business's sustainability.					
B4	Likert scale	Size of the business affects my business's sustainability.					
B5	Likert scale	Difficult access to finance (capital) affects my business's sustainability.					
B6	Likert scale	Stiff competition affects my business's sustainability.					
B7	Likert scale	Low demand for products and/or services affects my business's sustainability.					
	Likert scale	Location of the business affects my business's sustainability.					
_							
B9	Likert scale	Ineffective marketing affects my business's sustainability.					
B10	Likert scale	'Substitute' products and/or services at cheaper price affect my					
210		business's sustainability.					
B11	Likert scale	Unproductive employees affect my business's sustainability.					
B12	Likert scale	Poor service quality affects my business's sustainability.					
B13	Likert scale	Poor product quality affects my business's sustainability.					
B14	Likert scale	Cost of electricity affects my business's sustainability.					
B15	Likert scale	Cost of water affects my business's sustainability.					
B16	Likert scale	Lack of adequate credit facilities affects my business's sustainability.					
B17	Likert scale	Unreliable supply of raw materials affects my business's sustainability.					
B18	Likert scale	Interest rates affect my business's sustainability.					
B19	Likert scale	Legislation, rules and regulations affect my business's sustainability.					
B20	Likert scale	Labour costs affect my business's sustainability.					
B21	Likert scale	Crime rate affects my business's sustainability.					
B22	Likert scale	Bad debts (customers not pay debts) affect my business's					
DZZ		sustainability.					
B23	Likert scale	Taxation affects my business's sustainability.					
B24	Likert scale	Rapid change in technology affects my business's sustainability.					

Table 3.2: Questions in the questionnaire tool – Section B

### 3.7.3 Questionnaire tool: Section C

The objective of this section was to identify the risks that affect the sustainability of South African SMMEs. The questions identified the common risks perceived to affect the sustainability of South African SMMEs. A four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 =

strongly agree) was used, and the questions started with the following base statement: *My business is faced with the following problem*: (see Table 3.3).

Question number	Question type	Question
C1	Likert scale	My business is faced with the following problem: poor service quality.
C2	Likert scale	My business is faced with the following problem: employee theft.
C3	Likert scale	My business is faced with the following problem: large expenses (in relation to income).
C4	Likert scale	My business is faced with the following problem: increase in cost prices of products.
C5	Likert scale	My business is faced with the following problem: poor product quality.
C6	Likert scale	My business is faced with the following problem: weak business profitability (profits).
C7	Likert scale	My business is faced with the following problem: weak business liquidity (cash).
C8	Likert scale	My business is faced with the following problem: weak business solvency (more liabilities).
C9	Likert scale	My business is faced with the following problem: weak business efficiency (poor efficiency).
C10	Likert scale	My business is faced with the following problem: health and safety risks.
C11	Likert scale	My business is faced with the following problem: employee risks (immigration, etc.).
C12	Likert scale	My business is faced with the following problem: environmental risks (waste, etc.).
C13	Likert scale	My business is faced with the following problem: strong competition.
C14	Likert scale	My business is faced with the following problem: low demand for products.
C15	Likert scale	My business is faced with the following problem: delays in supply chain process.

Table3.3: Questions in the questionnaire tool - Section C

## 3.7.4 Questionnaire tool: Section D

The objective of this section was to gain insight into the risk-management initiatives utilised by South African SMMEs to combat the identified risks. A four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree) was used, and the questions started with the following base statement: *My business deals with the aforementioned risks by means of:* (see Table 3.4 overleaf).

Question number	Question type	Question				
		My business deals with the aforementioned risks by means of a				
D1	Likert scale	code of ethics (principles guiding employees).				
D2	Likert scale	My business deals with the aforementioned risks by means of risk				
02		policies.				
D3	Likert scale	My business deals with the aforementioned risks by means of				
		setting the right 'tone at the top'.				
D4	Likert scale	My business deals with the aforementioned risks by means of				
		setting of risk appetite(s).				
D5	Likert scale	My business deals with the aforementioned risks by means of an				
		adequate business strategy.				
D6	Likert scale	My business deals with the aforementioned risks by means of				
		setting clear objectives.				
D7	Likert scale	My business deals with the aforementioned risks by means of				
	Liliant a a sta	internal interviews and discussion (e.g SWOT).				
D8	Likert scale	My business deals with the aforementioned risks by means of				
	Likert scale	external sources (e.g. risk consultants).				
D9	Liken scale	My business deals with the aforementioned risks by means of tools				
	Likert scale	diagnostics and processes (e.g. checklists). My business deals with the aforementioned risks by means of risk				
D10	LIKEIT SCALE	analysis.				
	Likert scale	My business deals with the aforementioned risks by means of risk				
D11	Lintert obtaile	evaluation(s).				
<b>.</b>	Likert scale	My business deals with the aforementioned risks by means of risk				
D12		treatment(s) (reducing actual risks).				
D40	Likert scale	My business deals with the aforementioned risks by means of				
D13		implementing controls to reduce the effect of risks.				
D14	Likert scale	My business deals with the aforementioned risks by means of using				
D14		budgets.				
D15	Likert scale	My business deals with the aforementioned risks by means of using				
015		policies and procedures.				
D16	Likert scale	My business deals with the aforementioned risks by means of timely				
		communication of risks.				
D17	Likert scale	My business deals with the aforementioned risks by means of				
		consulting risk management experts.				
D18	Likert scale	My business deals with the aforementioned risks by means of				
		conducting performance appraisals regularly.				
D19	Likert scale	My business deals with the aforementioned risks by means of				
_		performing separate evaluations.				

Table 3.4: Questions in the questionnaire tool - Section D

## 3.7.5 Questionnaire tool: Section E

This purpose of this section was to determine the level of understanding and implementation of ERM by respondents in South African SMMEs. A six-point Likert scale (1= none, 2 = little, 3 = very little, 4 = average, 5 = some, 6 = a lot.) was used (see Table 3.5).

Question number Question type		Question			
E1	Likert scale	How well do you understand the ERM process?			
E2	Likert scale	How many of the ERM initiatives have you implemented in your business?			
E3	Likert scale	How much value does ERM add to your business?			

Table 3.5: Questions in the questionnaire tool – Section E

#### 3.8 CONCLUSION AND RELEVANCE TO THE RESEARCH STUDY

In this chapter, emphasis was placed on the research design, research methodology and research methods deployed in this research study to answer relevant research questions to, in turn, achieve the relevant research objectives (see Section 1.3). For this research study, quantitative methods were used to implement the survey research design. In order to answer the research questions adequately, the researcher conducted both non-empirical research (by means of a literature review) (see Chapter 2) and empirical research (by deploying survey research).

To select the envisaged sample size of 150, a combination of purposive sampling and convenience sampling in conjunction with strict delineation criteria were used. Since most primary data collected were numerical, a structured questionnaire was deemed the appropriate tool to glean data. After the piloting of the questionnaire, the researcher self-administered the questionnaires to the envisaged sample. Only 130 respondents responded favourably, translating to a response rate of 86.67%; however, only 116 responses were regarded as valid. The data collected were captured and coded with the assistance of IBM<sup>®</sup> SPSS Statistics<sup>®</sup> software, and thereafter analysed using descriptive and inferential statistics (exploratory factor analysis) (see Chapter 4) . In addition, relevant ethical considerations were taken into account, while the limitations of the research study were also addressed.

Lastly, the questions posed in the questionnaire were addressed. The structured questionnaire, comprising mostly closed-ended questions, contained five sections, namely: 1) Section A (biography of SMMEs), 2) Section B (economic factors affecting South African SMME sustainability), 3) Section C (risks influencing South African SMME sustainability), 4) Section D (risk-management initiatives used by South African SMMEs), and 5) Section E (understanding and implementation of ERM by South African SMMEs' owners and/or managers). With most of the data being ordinal, a four-point Likert scale was used for Section B to D, while a six-point Likert scale was used for Section E. In Section A, there was a mixture of ratio questions and multiple-choice questions.

In the next chapter, results from the analysis are presented and discussed accordingly.

52

## **CHAPTER 4**

### DATA ANALYSIS, RESULTS AND DISCUSSION

#### **SYNOPSIS**

For this research study, a methodical process was followed whereby each chapter built on the previous chapter. This process is depicted below.

**Chapter 1**: Introduction to the research study

Chapter 2: Literature review

Chapter 3: Research design, research methodolody and research methods

Chapter 4: Data analysis, results and discussion

#### Introduction

- Data validity
- •Data reliability
- SMME demographic information
- Economic factors affecting South African SMMEs
- Risks faced by South African SMMEs
- Risk-management initiatives used by South African SMMEs
- Alignment of risk-management initiatives used with COSO ERM
- Perceptions of respondents regarding ERM
- •Conclusion and relevance to the study

## **Chapter 5: Conclusion**

## **4.1 INTRODUCTION**

For this research study, the following primary research question was asked (see Section 1.3.1):

To what extent do South African SMMEs utilise ERM to help enhance their overall sustainability?

In order to address this research question, the following investigative research questions were asked (see Section 1.3.2):

- What are the risks which South African fast-food SMMEs in the Cape Peninsula face?
- How are these risks treated by these business entities?
- Are these risk treatments aligned to ERM?

Although the first two investigative questions were theoretically investigated by conducting a literature review (see Chapter 2), all of the three investigative questions were empirically investigated by means of survey research (see Section 3.2). In this chapter, results from the data analysis are both provided and discussed under the following headings: 1) data validity, 2) data reliability, 3) demographic information of SMMEs, 4) economic factors affecting South African SMMEs, 5) risks faced by South African SMMEs, 6) risk-management initiatives used by South African SMMEs, 7) alignment of risk-management initiatives used with the COSO ERM, and 8) perceptions of respondents regarding ERM.

## 4.2 DATA VALIDITY

In any study, it is important to ensure that the collected data are valid (Bernard, 2000:46); otherwise incorrect conclusions can be made. Data validity can be regarded as the suitability of collected data with regard to the construct under study (Leung, 2015:324). Since this study was quantitative in nature, data validity was enhanced by ensuring the validity of the data-collection tool used (Sullivan, 2011:119). In essence, two types of validity measures were used to enhance the validity of the data-collection tool, namely content validity and construct validity (see Section 3.6).

Notwithstanding the above, the validity of data was also subject to a strict list of delineation criteria (see Section 3.3), where each respondent had to adhere to certain core characteristics before their responses would be regarded as valid. Out of 130 responses received, a total of 116 (89%) were regarded as valid.

# 4.3 DATA RELIABILITY

The data-collection tool used to collect data in any research study should be reliable in order to provide credible data. Data reliability entails that collected data are adequate, complete and accurate, with the intent to measure the relevant construct under investigation (Morgan & Waring, 2004:2). This was of particular importance for this study since collected data had to shed empirical light on the first two investigative questions, while answering the final investigative question (see Section 1.3.2). For this study, a questionnaire comprising nine questions was used to collect data from respondents (see Annexure A). In this data-collection tool, four Likert-scale questions were asked.

In order to determine the reliability of the data obtained through the questionnaire, relevant Cronbach's Apha<sup>39</sup> values were calculated to determine internal consistency,<sup>40</sup> particularly for the four Likert-scale questions. According to the Institute for Digital Research and Education (2017) at UCLA, a Cronbach's Alpha value of 0.700, for internal consistency testing, is deemed acceptable. The relevant Cronbach's Alpha values for the four Likert-scale sections, as analysed, are presented in Table 4.1.

Section	Items tested	Cases	Cronbach's Alpha		
В	24	116	0.822		
С	17	116	0.868		
D	19	116	0.907		
E	3	116	0.934		

Table 4.1: Cronbach's Alpha values for the four Likert-scale questions in the questionnaire

<sup>&</sup>lt;sup>39</sup> Cronbach's Alpha is a statistical measure that examines how closely related a set of items are (Institute for Digital Research and Education, 2017).

<sup>&</sup>lt;sup>40</sup> Internal consistency refers to the extent to which a number of items (variables) in a questionnaire measure the same general construct (Tang et al., 2014:205).

From the above, the inference can be made that the data collected for this research study, particularly from the four Likert-scale questions, were deemed reliable. This view is supported by the fact that all Cronbach's Alpha values were significantly larger than the 0.700 threshold.

## 4.4 DEMOGRAPHIC INFORMATION

For respondents to be regarded as valid respondents, they had to satisfy a range of delineation criteria (see Section 3.3). In particular, respondents had to be owners and/or managers of non-franchised fast-food SMMEs which had been in operation in the Cape Peninsula for at least two years. Of the targeted 150 respondents, a total of 116 (77%) were deemed valid respondents; only their provided data were used to perform relevant data analysis.

When respondents were asked about the positions they fulfilled in their respective SMMEs, 67.25% indicated that they were managers, 26.72% indicated that they were owners, while 6.03% indicated that they were both owner and manager of their respective enterprises. A summary of these results are depicted in Figure 4.1.



Figure 4.1: Position of sampled respondents in their respective SMMEs

Stemming from the above, the inference can be made that the respondents had the necessary authority to make business decisions in their respective SMMEs.

Respondents were also asked to indicate the number of years their businesses had been in existence. On average, it was found that these businesses had been in existence for 7.18 years, with a modal<sup>41</sup> and median<sup>42</sup> duration of 3 years and 5 years respectively. Although the aboveaverage duration gave the impression that sampled SMMEs had been in existence for longer periods than generally found in the literature (see Section 2.1), only 58.5% of sampled SMMEs had existed for four years or longer. Moreover, the modal duration of 3 years and a median duration of 5 years suggest that the number of years that these sampled SMMEs had been in existence is not as remarkable as portrayed by the average duration of 7.18 years. This view is substantiated by the fact that the calculated average is significantly higher than both the mode and median. A probable explanation for the latter dispensation is that there were a few businesses (5.2% of respondents) that had been in existence for more than 15 years. Thus, the inference can be made that the existence rate of sampled SMMEs leaves much to be desired, since it is almost identical to the theoretical depiction of these businesses' existence rate as per scholarly literature (see Section 2.1). A summary of these results is presented in Figure 4.2.



Figure 4.2: Number of years which sampled SMMEs had been in existence

<sup>&</sup>lt;sup>41</sup> Mode refers to the value that has the highest frequency (Gravetter & Wallnau, 2013:73).

<sup>&</sup>lt;sup>42</sup> Median is the value that is in the middle of a distribution. Stated otherwise, it is a value where 50% of the observations fall below it and the other 50% fall above it (Gravetter & Wallnau, 2013:72).

When respondents were asked to indicate their number of full-time employees, the results show that 51.7% of sampled SMMEs employed between 0 and 5 full-time employees, 20.7% employed between 6 and 10 full-time employees, while 27.6% employed between 11 and 50 full-time employees. Using the classification of SMME sizes as foundation (see Section 1.1), the inference can be made that 51.7% of respondents were owners and/or managers of micro enterprises, 20.7% of respondents were owners and/or managers of very small enterprises, and 27.6% of respondents were owners and/or managers of very small enterprises, and 27.6% of employees was calculated at 9.77, with a calculated median value of 5 and a calculated modal value of 3. Although typical enterprises of respondents could be regarded as very small enterprises (using an average of 9.77 employees), the majority of enterprises (51.7%) were actually regarded as micro enterprises. This view is supported by the median and modal values noted above. The above results are similar to the ones from a recent study by Bruwer (2016:188). In essence, the number of employees of these SMMEs do not correspond with the years these entities has been in existence for. Thus, the inference can be made that the job-creation ability of the sampled SMMEs is undesirable. A summary of the results is depicted in Figure 4.3.



Figure 4.3: Number of full-time employees by sampled SMMEs

In order to understand the financial sustainability of sampled SMMEs, respondents were asked to provide their agreement in respect of statements which related to their respective SMMEs' profitability, liquidity, solvency and efficiency. To achieve this, a four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree) was used, and each statement

started with the following base sentence: *My business is faced with the following problem* .... A summary of the results is presented in Table 4.2.

My business is faced with the following problem:		Valid				Median	Mean	Standard
	Strongly disagree	Disagree	Agree	Strongly agree	Total Agreement	value	value	Deviation
Weak business profitability (weak profits)	11.20%	33.60%	42.20%	12.90%	55.10%	3	2.6	0.86
Weak business efficiency (takes long to make money)	11.20%	44.80%	35.30%	8.60%	43.90%	2	2.4	0.80
Weak business liquidity (little cash on hand)	11.20%	48.30%	30.20%	10.30%	40.50%	2	2.4	0.82
Weak business solvency (more liabilities than assets)	19.00%	44.80%	29.30%	6.90%	36.20%	2	2.2	0.84

 Table 4.2: Perceived financial sustainability of sampled SMMEs

The two central measures used to represent the perception of the sampled SMMEs were mean and median. This was specifically determined as mean values can be used to rank and/or prioritise items, while median values provide insight into the absolute middle average (Gravetter & Wallnau, 2013:72). Using the above as a basis, a four-point Likert scale will have a point of neutrality when a mean value is 2.5. In the same vein, a mean greater than 2.5 will entail agreement, while a mean less than 2.5 will entail disagreement. Thus, the further the mean is from the neutrality point (above 2.5), the stronger the level of agreement, and the further the mean is from the neutrality point (below 2.5), the stronger the level of disagreement.

From the results in Table 4.2, it is apparent that the majority (at least 56.1%) of respondents did not perceive their respective SMMEs as having weak efficiency, solvency or liquidity. Nonetheless, the perceived efficiency and liquidity of sampled SMMEs were not perceived as very good either. This view is supported by the calculated mean values which were very close to 2.5 (neutrality point). Thus, the inference can be made that sampled SMMEs had close-to-average business efficiency (56.10% of the time), close-to-average business liquidity (59.50% of the time) and an above-average business solvency (63.80% of the time). This is strengthened by the modal value of 2 for each of these three elements – indicating that respondents' absolute balanced perception was to disagree with the respective statements that their respective SMMEs had weak business efficiency, weak business liquidity, and weak business solvency.
A probable reason for these business entities experiencing only moderate business efficiency, moderate business liquidity, and moderate business solvency may be attributed to their operating in a harsh economic environment (see Section 2.3). This view is particularly justified when emphasis is placed on the results surrounding business profitability. The inference can be made that sampled SMMEs experienced weak profitability (perceived), since the calculated mean amounted to 2.6, with the calculated median amounting to 3. Otherwise stated, respondents leaned slightly towards agreeing with the statement (55.10% of the time) that their respective SMMEs experienced weak profitability. In essence, this affirms what was found in a recent GEM report that 41.2% of the SMMEs that discontinued their operations in 2016, did so because of weak profitability (Herrington & Kew, 2017:112).

In summary, while taking all applicable demographical information into consideration, the average respondent for this research study was a manager of a very small non-franchised business entity, with 9.77 full-time employees, while being in existence for 7.18 years, and that possessed a relatively moderate financial sustainability.

#### **4.5 ECONOMIC FACTORS AFFECTING SOUTH AFRICAN SMMEs**

In order to understand the economic environment in which sampled SMMEs operated, respondents were asked to identify the various economic factors which influenced their respective SMMEs' sustainability (including their financial sustainability). To achieve this, a four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree) was used, and each statement started with the following base sentence: *The following barriers affect my business's sustainability ....* A summary of the results is presented in Table 4.3 overleaf.

#### **Table 4.3:** Economic factors affecting the sustainability of sampled SMMEs

Table 4.3. Economic factors anceting the sustain	Valid					Median value	Mean value	Standard Deviation
	Strongly disagree	Disagree	Agree	Strongly agree	Total agreement			
The following barriers affect my business's sustainability:								
Cost of electricity	4.3%	12.9%	28.4%	54.4%	82.8%	4	3.3	0.86
Inflation rates	5.2%	11.2%	49.1%	34.5%	83.6%	3	3.1	0.81
Food prices of suppliers	6.9%	13.8%	38.8%	40.5%	79.3%	3	3.1	0.90
Stiff competition	6.9%	16.4%	31.9%	44.8%	76.7%	3	3.1	0.93
Difficult access to finance (external funding)	12.1%	17.2%	28.4%	42.3%	70.7%	3	3.0	1.04
Unemployment rates	6.0%	31.0%	34.5%	28.5%	63.0%	3	2.9	0.91
Labour costs	12.1%	20.7%	40.5%	26.7%	67.2%	3	2.8	0.97
Interest rates	11.2%	23.3%	39.7%	25.8%	65.5%	3	2.8	0.95
Crime rate	15.5%	20.7%	31.9%	31.9%	63.8%	3	2.8	1.06
Cost of water	18.1%	19.0%	28.4%	34.5%	62.9%	3	2.8	1.11
Legislation, rules and regulations	12.9%	28.4%	36.2%	22.5%	58.7%	3	2.7	0.97
Size of business (capacity to meet demand)	14.7%	27.6%	33.6%	24.1%	57.7%	3	2.7	1.00
Substitute products and/or services at cheaper prices	8.6%	34.5%	44.8%	12.1%	56.9%	3	2.6	0.81
Lack of adequate credit facilities	12.1%	38.8%	35.3%	13.8%	49.1%	2	2.5	0.88
Low demand for products and/or services	17.2%	37.9%	34.5%	10.4%	44.9%	2	2.4	0.89
Taxation	31.9%	20.7%	32.8%	14.6%	47.4%	2	2.3	1.07
Unreliable supply of raw materials	22.4%	36.2%	31.9%	9.5%	41.4%	2	2.3	0.92
Ineffective marketing	24.1%	34.5%	26.7%	14.7%	41.4%	2	2.3	1.00
Rapid change in technology	25.0%	37.1%	25.0%	12.9%	37.9%	2	2.3	0.98
Unproductive employees	37.9%	19.8%	25.9%	16.4%	42.3%	2	2.2	1.12
Poor service quality	35.3%	25.9%	25.9%	12.9%	38.8%	2	2.2	1.05
Location of business	28.4%	34.5%	21.6%	15.5%	37.1%	2	2.2	1.04
Bad debts (customers not paying)	44.0%	18.1%	24.1%	13.8%	37.9%	2	2.1	1.11
Poor product quality	36.2%	31.9%	17.2%	14.7%	31.9%	2	2.1	1.05

On average, all the economic factors with a median value of 3 and higher were regarded by respondents as adversely affecting their business sustainability. Using the calculated mean values as a basis, the most predominant economic factors were the cost of electricity (mean of 3.3), inflation rates (mean of 3.1), food prices of suppliers (mean of 3.1), stiff competition (mean of 3.1), and difficult access to finance (external funding) (mean of 3.1). Other economic factors also perceived to be negatively impacting the sampled SMMEs' sustainability were unemployment rates (mean of 2.9), labour costs (mean of 2.8), interest rates (mean of 2.8), crime rate (mean of 2.8), cost of water (mean of 2.8), legislation, rules and regulations (mean of 2.7), size of business (capacity to meet demand) (mean of 2.7), and substitute products and/or services at cheaper prices (mean of 2.6). All the economic factors with a median value of less than 3 were regarded by respondents as not adversely affecting their relevant businesses' sustainability. These economic factors were lack of adequate credit facilities (mean of 2.5), low demand for products and/or services (mean of 2.4), taxation (mean of 2.3), unreliable supply of raw materials (mean of 2.3), ineffective marketing (mean of 2.3), rapid change in technology (mean of 2.3), unproductive employees (mean of 2.3), poor service quality (mean of 2.2), location of business (mean of 2.2), bad debts (customers not paying) (mean of 2.1), and poor product quality (mean of 2.1). Although there are several other economic factors which may influence the sustainability of SMMEs, only the above economic factors were covered in this study as they were regarded by scholarly literature as being the most influential on business sustainability (see Section 2.3).

Taking into account the above, it becomes apparent that sampled SMMEs were affected by mostly macro-economic factors (see Section 2.3), particularly those of cost of electricity, stiff competition, and inflation. Hence, the inference can be made that the economic environment in which these SMMEs had to operate was undesirable and/or harsh. Therefore, it is highly probable that sampled SMMEs may have been adversely affected by a plethora of risks which could emanate from the harsh economic environment in which they operate.

#### 4.6 RISKS FACED BY SOUTH AFRICAN SMMEs

Taking into account that sampled SMMEs operate in a harsh economic environment (see Section 4.5), which is regarded by scholarly literature as a type of 'breeding ground' for risks (see Section 2.3), it is highly likely that these entities' sustainability was influenced by an array of risks. To determine whether this was the case, respondents were asked to provide their perceptions on relevant risks which affect their respective SMMEs' sustainability. This was done by using a four-

point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree), with each statement starting with the following base sentence: *My business is faced with the following problem* .... A summary of the results is presented in Table 4.4.

All the risks with a median value of 3 or more were the ones which respondents perceived as influencing their respective businesses' sustainability. Considering the mean values, the risks that influenced these businesses the most (perceived) were those of strong competition (mean of 3.4) and increase in cost prices of products (mean of 3.0). Other risks that were also perceived as influencing the business sustainability of sampled SMMEs were employee theft (mean of 2.6), large expenses (in relation to income) (mean of 2.6), unstable business conditions (mean of 2.6), low demand for products (mean of 2.6), weak business growth (mean of 2.6), low demand for products (mean of 2.5), and environmental risks (waste, etc.) (mean of 2.5). Although the mean values for low demand for products and environmental risks (waste, etc.) suggested that respondents were neutral (mean of 2.5), the median values (3) and agreement percentages (52.6%) serve as justification that respondents leaned towards agreement on the relevant statements. Respondents regarded the following risks as not influencing their businesses' sustainability: employee risks (immigration, etc.) (mean of 2.3), delays in supply chain process (mean of 2.3), poor product quality (mean of 2.2), and poor service quality (mean of 2.0). The abovementioned four risks all had a median value of 2, which indicates that respondents disagreed that these risks did influence their SMMEs' sustainability to a large extent. As indicated by the agreement level of 50% and a median value of 2.5, the risk which SMMEs were undecided on was health and safety.

In essence, stemming from the results above, these businesses were mostly affected by strategic risks (strong competition) and operational risks (employee theft and increases in cost prices of products and services), most of which may have stemmed from economic factors (e.g. stiff competition, inflation rates, and cost of electricity). Nevertheless, without sound risk- management initiatives, these risks may realise and result in economic losses (e.g. weaker profitability, weaker liquidity, weaker efficiency and/or weaker solvency) and/or non-economic losses (e.g. lost customer loyalty). The potential impact of these risks may be greater for these business entities, taking into account that scholarly literature suggests that they do not make use of sound risk-management initiatives (see Section 2.7).

## Table 4.4: Risks which affected sampled SMMEs' sustainability

		Valid					Mean value	Standard Deviation
My business is faced with the following problem:	Strongly disagree	Disagree	Agree	Strongly agree	Total agreement			
Strong competition	5.2%	10.3%	27.6%	56.9%	84.5%	4	3.4	0.87
Increase in cost prices of products	4.3%	19.8%	50.9%	25.0%	75.9%	3	3.0	0.79
Employee theft	27.5%	12.1%	37.1%	23.3%	60.4%	3	2.6	1.13
Large expenses (in relation to income)	19.0%	23.3%	37.9%	19.8%	57.7%	3	2.6	1.01
Unstable business conditions	7.8%	37.1%	44.0%	11.1%	55.1%	3	2.6	0.79
Weak business growth	10.3%	37.1%	38.8%	13.8%	52.6%	3	2.6	0.86
Low demand for products	18.1%	29.3%	41.4%	11.2%	52.6%	3	2.5	0.92
Environmental risks (waste, etc.)	18.1%	29.3%	33.6%	19.0%	52.6%	3	2.5	1.00
Health and safety risks	23.3%	26.7%	38.8%	11.2%	50.0%	2.5	2.4	0.97
Employee risks (immigration, etc.)	23.3%	34.5%	28.4%	13.8%	42.2%	2	2.3	0.98
Delays in supply chain process	18.1%	46.6%	25.0%	10.3%	35.3%	2	2.3	0.88
Poor product quality	32.8%	25.0%	32.8%	9.4%	42.2%	2	2.2	1.00
Poor service quality	39.7%	31.9%	18.9%	9.5%	28.4%	2	2.0	0.99

#### 4.7 RISK-MANAGEMENT INITIATIVES USED BY FAST-FOOD SMMEs

In order to determine whether sampled SMMEs made use of effective and/or adequate riskmanagement initiatives, respondents were asked to identify the risk-management initiatives they used in dealing with the risks they faced (see Section 4.6). To achieve this, a four-point Likert scale (1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree) was used, with each statement starting with the following base sentence: *My business deals with the aforementioned problems by means of ....* A summary of the results is presented in Table 4.5.

The median values of almost all the risk-management initiatives listed are greater than 3, implying that respondents made use of almost all the listed initiatives. Furthermore, the mean values in Table 4.5 are significantly above the point of neutrality (mean of 2.5), thereby suggesting that there was a moderate to strong agreement among respondents concerning the use of the identified risk-management initiatives. Stated otherwise, these SMMEs used these risk-management initiatives most of the time. Considering the agreement percentages as basis, the risk-management initiatives mostly used by sampled SMMEs were those of budgets (91.4% agreement), setting of clear objectives (88.8% agreement), using policies and procedures (85.4% agreement), using a code of ethics (82.7% agreement), using controls to reduce the effects of risk (85.4% agreement), and using an adequate business strategy (82.8% agreement). In essence, the above echo what is noted in literature, that South African SMMEs mostly use internal controls to manage risks (see Section 2.9). The only initiatives that the respondents did not use were risk-management experts (38% agreement) and risk- management consultants (33.2% agreement).

Since the respondents made use of almost all the listed risk-management initiatives (see Table 4.5), the inference can be made that the risk-management initiatives used by these businesses may have been effective and/or adequate. Taking into account that these business entities were still to a large extent influenced by risks which may have stemmed from economic factors (see Section 4.6), the effectiveness and/or adequacy of these risk-management initiatives should be regarded as an avenue for further research.

		Val				Median		Standard Deviation
My business deals with the aforementioned problems by means of:	Strongly disagree	Disagree	Agree	Strongly agree	Total agreement	value		
Using budgets	1.7%	6.9%	35.3%	56.1%	91.4%	4	3.5	0.70
Setting clear objectives	2.6%	8.6%	34.5%	54.3%	88.8%	4	3.4	0.76
Using policies and procedures	1.7%	12.9%	44.8%	40.6%	85.4%	3	3.2	0.74
Code of ethics (principles guiding employees	5.2%	12.1%	44.0%	38.7%	82.7%	3	3.2	0.83
Implementing controls to reduce the effect of risks	3.4%	11.2%	52.6%	32.8%	85.4%	3	3.1	0.75
An adequate business strategy	4.3%	12.9%	52.6%	30.2%	82.8%	3	3.1	0.78
Internal interviews and discussion (e.g. SWOT)	6.0%	17.2%	36.2%	40.6%	76.8%	3	3.1	0.90
Performing separate evaluations	6.0%	16.4%	51.7%	25.9%	77.6%	3	3.0	0.82
Setting the right tone at the top	7.8%	14.7%	43.1%	34.4%	77.5%	3	3.0	0.90
Tools diagnostics and processes (e.g. checklists)	7.8%	15.5%	43.1%	33.6%	76.7%	3	3.0	0.90
Timely communication of risks	4.3%	21.6%	47.4%	26.7%	74.1%	3	3.0	0.81
Risk evaluation(s) (prioritising or ranking the identified risks)	4.3%	22.4%	43.1%	30.2%	73.3%	3	3.0	0.84
Risk analysis	3.4%	25.0%	44.0%	27.6%	71.6%	3	3.0	0.82
Conducting performance appraisals regularly	5.2%	17.2%	55.2%	22.4%	77.6%	3	2.9	0.77
Setting of risk appetite(s)	6.0%	22.4%	49.1%	22.5%	71.6%	3	2.9	0.82
Risk treatment(s) (reducing actual risks)	4.3%	24.1%	49.1%	22.5%	71.6%	3	2.9	0.80
Risk policies	5.2%	26.7%	46.6%	21.5%	68.1%	3	2.8	0.82
External sources (using risk-management consultants)	31.0%	31.0%	26.7%	11.3%	38.0%	2	2.2	1.00
Consulting experts (risk-management experts)	29.3%	34.5%	22.4%	13.8%	36.2%	2	2.2	1.01

### Table 4.5: Risk-management initiatives used by sampled SMMEs to manage risks

## 4.8 ALIGNMENT OF SMME RISK-MANAGEMENT INITIATIVES WITH THE COSO ERM FRAMEWORK

Stemming from the results above, it appears that sampled SMMEs made use of an array of risk-management initiatives to identify, analyse and treat risks. Therefore, in order to determine the extent to which the risk-management initiatives utilised by sampled SMMEs held relevancy to the COSO ERM Framework, relevant inferential statistical analysis was conducted. Specifically, all variables as shown in Table 4.5 were analysed through the use of principle component analysis and principle axis factoring, otherwise known as exploratory factor analysis.<sup>43</sup> This was done with the main intent to reduce the number of variables (items) from 19 into a number of factors as close to 1 as possible (Fabrigar & Wegener, 2011:1). A summary of the exploratory factor analysis results is shown in Table 4.6.

	Component					
	1	2	3	4		
8.1) A code of ethics (principles guiding	.623	323	.084	.250		
employees)						
8.2) Risk policies	.638	031	.060	207		
8.3) Setting the right 'tone at the top'	.416	393	.404	.500		
8.4) Setting of risk appetite(s)	.585	045	.241	352		
8.5) An adequate business strategy	.714	171	221	.009		
8.6) Setting clear objectives	.627	445	.019	.129		
8.7) Internal interviews and discussion (e.g. SWOT)	.682	.078	.288	.090		
8.8) External sources (e.g. risk consultants)	.398	.691	.240	.310		
8.9) Tools diagnostics and processes (e.g. checklists)	.578	.038	.328	.438		
8.10) Risk analysis	.739	.084	.255	396		
8.11) Risk evaluation(s)	.809	.160	.203	354		
8.12) Risk treatment(s) (reducing actual risks)	.731	.086	.186	277		
8.13) Implementing controls to manage risk	.657	.066	.020	.021		
8.14) Using budgets	.687	313	202	006		
8.15) Using policies and procedures	.626	166	451	.145		
8.16) Timely communication of risks	.676	.053	424	.203		
8.17) Consulting experts	.250	.808	104	.285		
8.18) Conducting performance appraisals regularly	.683	.144	447	.030		
8.19) Performing separate evaluations	.642	.230	308	152		

**Table 4.6:** Summary of exploratory factor analysis results

Extraction Method: Principal Component Analysis. 4 components extracted.

Stemming from the results in Table 4.6, the inference can be made that of the 17 different riskmanagement initiatives used by SMMEs, a total of 16 (84.21%) related strongly to the COSO ERM Framework (see Section 2.6.2), forming a single factor in this regard. According to Rossoni et al. (2016:200), the minimum acceptable values for exploratory factor analysis in

 <sup>&</sup>lt;sup>43</sup> Factor analysis refers to the reduction of many factors into a lesser number of factors (Dictionary.com, 2017). Essentially, these new factors will constitute one or more of the initial individual factors.

relation to the Kaiser-Meyer-Olkin (KMO) test<sup>44</sup> and Cronbach's Alpha are 0.500 and 0.700 respectively. This single factor showed a KMO test value of 0.876 (a chi-squared value based on Bartlett's test of sphericity of 990.506) and a Cronbach's Alpha of 0.918 – justifying the legitimacy of the calculated factor. For all intents and purposes, the factor will be referred to as *COSOERM* for the remainder of this study. The mean (3.07), median (3), and standard deviation (0.54) for *COSOERM* indicated that the sampled SMMEs did make use of the factor. Stated otherwise, the risk-management initiatives which were used by these businesses held strong relevancy to the COSO ERM Framework.

#### 4.9 PERCEPTIONS OF RESPONDENTS REGARDING ERM

Using the above as a basis, clear tangent planes emerge that SMMEs made use of riskmanagement initiatives which were directly related to the COSO ERM Framework; however, the question remains whether respondents implemented *COSOERM* by choice or by chance. This is particularly important since it is claimed that South African SMMEs unknowingly make use of ERM initiatives (see Section 2.7). To help shed light on this, respondents' perceptions on ERM were gleaned using a six-point Likert scale (1 = none, 2 = very little, 3 = little, 4 = average, 5 = some, 6 = a lot) on three individual questions. The results are presented in Table 4.7

	Valid						Median value	Mean value	Standard Deviation
	None	Very little	Little	Average	Some	A lot			
How well do you understand the ERM process?	21.60%	20.70%	10.30%	24.10%	15.50%	7.80%	3.0	3.15	1.62
How many of the ERM initiatives have you implemented in your business?	21.60%	21.50%	18.10%	23.30%	11.20%	4.30%	3.0	2.94	1.47
How much value does ERM add to your business?	19.80%	19.80%	19.80%	15.50%	17.30%	7.80%	3.0	3.14	1.59

Table 4.7: Perceptions of respondents regarding ERM

The first question was intended to better understand how familiar respondents were with the ERM process. From the table above, it is clear that the majority (52.60%) had virtually no understanding thereof. This is of concern, since without sufficient understanding of the ERM process it is impossible to implement effective and/or adequate ERM initiatives. Consequently,

<sup>&</sup>lt;sup>44</sup> The KMO test measures if the sample used in a study is adequate for factor analysis to be performed (Tabussum & Mahmood, 2015:61).

taking into account that most respondents had no understanding of the ERM process, it is highly probable that it may have adversely influenced the effectiveness and/or adequacy of implemented ERM initiatives. Thus, the inference can be made that the implemented ERM initiatives may have been implemented by chance. Stated otherwise, respondents may have unknowingly implemented ERM initiatives, without understanding the technicalities surrounding the ERM process.

Respondents were also asked to indicate the extent to which they had gone to implement ERM initiatives (as built on the COSO ERM Framework) in their respective businesses. As expected, the results of the second question were similar to those of the first question. A total of 61.20% of respondents indicated that they had implemented virtually none of the initiatives of the COSO ERM Framework. These results contradict the results stemming from Table 4.6, which suggested that respondents made use of *COSOERM*. This inconsistency may be due to respondents' unknowingly use of risk-management initiatives that are in line with the COSO ERM Framework, supporting the perception that respondents may have unknowingly implemented ERM initiatives without having any technical insight into the ERM process. In the same vein, it may also have been the case that respondents took a chance in relation to the implementation of ERM initiatives.

Furthermore, respondents were asked to indicate their perceptions on value added by ERM to their respective businesses. Considering that value added by ERM is most likely influenced by the ERM initiatives implemented, it is unsurprising that the results of this question corresponded to the results of the second question. From Table 4.6 it was established that the majority of respondents (59.4%) regarded the value added by ERM to their businesses as negligible. Hence, it was not surprising that more than half of respondents (61.20%) had not yet made significant efforts to implement ERM. Besides the ERM initiatives implemented, a lack of knowledge about ERM could have also contributed to the respondents' perceptions on the value added by ERM.

From the inferential statistics above, it is highly probable that the implemented ERM initiatives in sampled SMMEs were more likely implemented by chance than by choice. This view is underscored by the fact that the average respondent had virtually no understanding of ERM. Furthermore, this concurs with the results from recent studies, where it was established that most SMMEs do not have a formal risk-management framework in place (Chakabva, 2015:65; Bruwer & Siwangaza, 2016:112). This is a phenomenon which should be further investigated in future studies.

#### 4.10 CONCLUSION AND RELEVANCE TO THE STUDY

The purpose of this chapter was to present the analysed data and subsequent discussion to adequately address the primary research question and relevant investigative research questions (see Section 1.3.2). Before the analysis and presentation of results, relevant issues of data validity and data reliability were addressed. In essence, the internal consistency of the variables in the questionnaire was analysed using computed Cronbach's Alpha values. Stemming from the calculated Cronbach's Alpha values, it was concluded that collected data were reliable.

The demographic information pertaining to the position of respondents in their respective businesses, the number of years their businesses had existed, and the number of employees, was analysed by means of descriptive statistics. This was done to provide background information on sampled SMMEs. Furthermore, the perceived financial sustainability of these SMMEs was analysed, and it was considered to be moderate. Results on the economic factors affecting these entities were also presented and discussed. Sampled SMMEs were mostly affected by macro-economic factors such as stiff competition, cost of electricity, and inflation rates. Regarding the risks affecting the sustainability of these businesses, the results showed that the most predominant risks were operational risks (employee theft, and increases in cost prices of products and services) and strategic risks (strong competition). The inference was made that the risk-management initiatives of these entities were ineffective and/or inadequate.

The risk-management initiatives utilised by the sampled SMMEs were analysed, and the results showed that these businesses used almost all the listed initiatives (see Section 4.7). However, the predominant risk-management initiatives used were budgets, setting of clear objectives, using policies and procedures, using a code of ethics, using controls to reduce the effects of risk, and using an adequate business strategy. In order to conclusively answer the third investigative research question (see Section 1.3.2), relevant inferential statistical analysis was performed in the form of exploratory factor analysis. The results showed that the risk-management initiatives used by sampled SMMEs had strong relevance to the COSO ERM Framework. Taking into account the results from respondents' perceptions on ERM, evidence was provided that these business entities most likely made use of such initiatives by chance as opposed to choice.

In the next chapter, the research problem, relevant research questions and research objectives are revisited, while relevant conclusions are drawn and recommendations made.

### **CHAPTER 5**

### CONCLUSION

#### SYNOPSIS

For this research study, a methodical process was followed whereby each chapter built on the previous chapter. This process is depicted below.

Chapter 1: Introduction to the research study

Chapter 2: Literature review

Chapter 3: Research design, research methodolody and research methods

Chapter 4: Data analysis, results and discussion

### Chapter 5: Conclusion

- Introduction
- •Research problem revisited
- Primary research question and primary research objective revisited
- Investigative research questions and secondary research objectives revisited
- Conclusion of the research study
- Recommendations
- Avenues for further research

#### **5.1 INTRODUCTION**

The main purpose of this study was to address the identified research problem (see Section 1.2) by means of addressing relevant research questions (both primary and investigative), together with their respective research objectives (both primary and secondary) (see Section 1.3). The first two investigative questions were predominantly investigated through an extensive literature review (see Chapter 2). Related secondary research objectives were achieved by providing answers to the posed investigative research questions, from where certain concepts were conceptualised:

- Sustainability (in general) (see Section 2.2).
- Sustainability (in a South African SMME dispensation) (see Section 2.2).
- Risk (see Section 2.4).
- Risk management (see Section 2.4)

Although the first two investigative research questions were also partially addressed by empirical research, the final investigative research question predominantly relied on the latter in order to achieve the related secondary research objective. The questionnaire used in this study was developed from the literature review conducted in Chapter 2, and data were collected from owners and/or managers of fast-food SMMEs operating in the Cape Peninsula.

In order to select the units of analysis, a combination of purposive sampling and convenience sampling was used. Furthermore, these sampling methods were supported by strict delineation criteria which the units of analysis had to adhere to in order to be considered valid respondents. The gleaned data were analysed using predominantly descriptive statistics and partially inferential statistics, and the results were presented and discussed in Chapter 4. In essence, the data were analysed under the following headings:

- SMME demographic information (position of respondents in their respective businesses, number of employees, and years their businesses have been in existence) (see Section 4.4).
- Economic factors affecting South African SMMEs (see Section 4.5).
- Risks faced by South African SMMEs (see Section 4.6).
- Risk-management initiatives used by South African SMMEs (see Section 4.7).
- Alignment of risk-management initiatives used with the COSO ERM (see Section 4.8).
- Perceptions of respondents regarding ERM (see Section 4.9).

The purpose of this chapter is to draw final conclusions and make recommendations by revisiting the research problem, the primary research question, the three investigative questions, the primary research objectives, and the secondary research objectives.

#### 5.2 RESEARCH PROBLEM REVISITED

Following the background to the study in Chapter 1 (see Section 1.1), the study posited that South African SMMEs had weak business sustainability, mainly owing to their negligence in deploying proper risk-management initiatives. The term 'sustainability', in a South African SMME dispensation, was conceptualised in Chapter 2 (see Section 2.2). Despite the general consensus in literature regarding the neglect of sound risk-management initiatives by South African SMMEs in general, little was known about the risk-management initiatives used and the extent to which they were in accordance with ERM, particularly the renowned COSO ERM Framework. In essence, the problem statement that was identified in this study read as follows:

South African SMMEs have a weak sustainability 'track record', owing to the insufficient utilisation of sound risk-management initiatives – particularly that of enterprise risk management (ERM).

In order to address the above research problem, the relevant primary research question together with the related research objective was formulated.

# 5.3 PRIMARY RESEARCH QUESTION AND PRIMARY RESEARCH OBJECTIVE REVISITED

To adequately address the identified research problem, the following primary research question was posed:

Do South African fast-food SMMEs operating in the Cape Peninsula utilise ERM to help enhance their overall sustainability?

The primary research objective stemming from the above primary research question, read:

To ascertain whether South African fast-food SMMEs operating in the Cape Peninsula utilise ERM to help enhance their overall sustainability. In order to comprehensively answer the identified primary research question, with the main intent of attaining the primary research objective, three investigative research questions and the related secondary research objectives were formulated.

# 5.4 INVESTIGATIVE RESEARCH QUESTIONS AND SECONDARY RESEARCH OBJECTIVES REVISITED

All of the three developed investigative research questions, together with the secondary research objectives, were closely linked to one another, including the primary research question and primary research objective. These investigative questions and their respective secondary objectives are revisited below.

# 5.4.1 First investigative research question and its relevant secondary research objective revisited

In an attempt to answer the primary research question, the first investigative question read as follows:

What are the risks which South African fast food SMMEs in the Cape Peninsula face?

The aim of this investigative research question was to attain the following secondary research objective:

To determine the risks that South African fast food SMMEs in the Cape Peninsula face.

Considering that the risks affecting businesses emanate from the economic environment of the economies in which these businesses operate, it was deemed necessary first to consider the South African economic environment in conjunction with the economic factors affecting SMMEs. According to literature, the economic factors affecting South African SMMEs comprise two main categories, namely, macro-economic factors (e.g. high unemployment, high interest rates, high inflation rates, and expensive and unreliable electricity supply), and micro-economic factors (e.g. lack of access to capital, unreliable suppliers, availability of substitute products, and poor management skills) (see Section 2.3). Based on the literature review, the South African economic landscape was found to be undesirable for SMMEs to

operate in. In core, this environment was regarded as 'harsh' – a type of 'breeding ground' for a plethora of risks (see Section 2.3).

When focus is shifted to the sampled SMMEs in this study, based on the descriptive statistics performed (see Section 4.4), the typical respondent was revealed as a manager of a very small non-franchised business entity, with 9.77 employees, and having been in existence for 7.18 years with a relatively moderate financial sustainability (see Section 4.4). Most respondents indicated that their respective SMMEs' sustainability was predominantly affected by cost of electricity, inflation rates, food prices of suppliers, stiff competition, and difficult access to finance (external funding) (see Section 4.5).

Considering the above results, it became apparent that sampled SMMEs were mostly affected by macro-economic factors – those factors which to a large extent cannot be controlled by management. With this in mind, the finding from literature that the South African economic environment can be described as 'harsh' (see Section 2.3) is supported to a large extent. Thus the inference was drawn that sampled SMMEs faced numerous risks at a given time.

In order to adequately answer the first investigative research question, the term 'risk' was conceptualised (see Section 2.4), and this was achieved by examining a non-exhaustive list of definitions of the term 'risk' (see Table 2.3). In essence, the term 'risk' was conceptualised as the chances of threats and/or opportunities occurring, materialising and/or impacting, either positively or negatively, on the attainment of a business's overall sustainability (see Section 2.4). Furthermore risks were also demarcated into four broad categories: 1) strategic risks, 2) operational risks, 3) reporting risks, and 4) compliance risks (see Section 2.4).

Stemming from the results, sampled SMMEs were found to be affected by various risks, predominantly those of operational risks (i.e. employee theft, and increases in cost prices of products and services) and strategic risks (i.e. stiff competition). The literature review noted that the abovementioned risks can negatively impact a business's sustainability, especially if sound risk-management initiatives are not in place (see Section 2.4).

Hence, it was concluded that sampled SMMEs faced a range of risks which had the potential to negatively impact on their overall sustainability.

# 5.4.2 Second investigative research question and its relevant secondary research objective revisited

Stemming from the primary research question, the second investigative research question read as follows:

How are these risks treated by South African fast-food SMMEs in the Cape Peninsula?

The aim of this investigative research question was to attain the following secondary research objective:

# To understand the risk-management initiatives used by South African fast food SMMEs in the Cape Peninsula, in dealing with the identified risks.

As previously mentioned, this question was interrogated throuh literature review, and before the question could be answered, the term 'risk management' was conceptualised. Stemming from the non-exhaustive list of definitions of risk management (see Table 2.4), the term 'risk management' was conceptualised as the step-by-step process whereby potential threats (and opportunities) are identified and analysed in order to mitigate their realisation and potential impact (spur on their realisation and potential impact), with the main intent to provide reasonable assurance surrounding the attainment of relevant business objectives in the foreseeable future (see Section 2.4). Despite the paucity of literature on risk- management initiatives utilised by South African SMMEs operating in the fast-food industry, it was inferred from the available literature that these business entities mostly use customised risk-management initiatives which generally take the form of internal controls which include, inter alia, budget controls, having a code of ethics, performing cash counts, and having access controls to entrances (see Section 2.7).

Stemming from the results it was found that sampled SMMEs made significant use of almost all listed risk-management initiatives in the questionnaire (see Section 4.4). The risk management initiatives that were used most frequently, compared with others, were: 1) using budgets, 2) setting clear objectives, 3) using policies and procedures, 4) having a code of ethics, 5) using controls to reduce the effects of risk, and 6) using an adequate business strategy. In essence, the abovementioned risk-management initiatives affirm the claims of literature in the sense that South African SMMEs mostly use customised risk-management initiatives which take the form of internal control activities (see Section 2.7). However, considering that the sampled SMMEs made use of almost all the listed risk-management initiatives, it was inferred that these entities' risk management may have been effective and/or adequate.

Therefore it was concluded that the risk-management initiatives deployed by sampled SMMEs could possibly enable them to manage the risks they faced.

# 5.4.3 Third investigative research question and its relevant secondary research objective revisited

Also stemming from the primary research question, the third investigative research question read as follows:

# Are the risk treatments used by South African fast-food SMMEs in the Cape Peninsula aligned to ERM?

The aim of this investigative research question was to attain the following secondary research objective:

To compare the risk-management initiatives used by South African fast food SMMEs in the Cape Peninsula, against the COSO ERM framework, and determine if the initiatives are in accordance with this framework.

Although this question was solely answered by conducting empirical research, the literature review enabled the researcher to elaborate on the COSO ERM Framework (see Section 2.6.2). It was against this background by which targeted participants' responses (in relation to risk-management initiatives they used) were 'benchmarked'. In essence, this framework comprises eight components: 1) internal environment, 2) objective setting, 3) event identification, 4) risk assessment, 5) risk response, 6) control activities, 7) information and communication, and 8) monitoring (see Section 2.6.2).

In order to determine whether sampled SMMEs deployed risk-management initiatives that conformed to the COSO ERM Framework, inferential statistical analysis, in the form of exploratory factor analysis, was performed to group the listed 19 risk-management initiatives (see Section 4.5). Of the 19 risk management initiatives, a total of 16 initiatives were found to conform to *COSOERM* (see Table 4.6), with a calculated KMO value of 0.876 and a Cronbach's Alpha of 0.918. These results suggested that sampled SMMEs were actually using

risk-management initiatives that were in line with the COSO ERM Framework (see Section 4.8).

Notwithstanding the above, before any final conclusions were drawn, it was important to determine whether respondents were using ERM-related initiatives by choice or by chance (shedding light on how appropriately risks are treated in relation to ERM). Three questions were posed to respondents where they had to share their perceptions on ERM from where it was found that most respondents had virtually no understanding of ERM.

Therefore, it was concluded that these entities were not using ERM in a formal manner (choice), but were merely using ERM-related initiatives by chance.

#### 5.5 CONCLUSION OF THE RESEARCH STUDY

Despite their undoubted paramount importance, South African SMMEs have the worst business sustainability in the world. Although their weak sustainability could be attributed to many factors, the researcher was of the opinion that it is mainly due to the insufficient utilisation of risk-management initiatives, particularly ERM.

The seven key economic indicators examined in the literature review revealed that the South African economic landscape is undesirable for businesses to operate and be sustainable in. In essence, this economic landscape was considered to be 'harsh' - a 'breeding ground' for a plethora of risks. This 'harsh' economic landscape is responsible for the various risks affecting South African SMMEs, and these risks are demarcated into four categories: 1) operational risks, 2) strategic risks, 3) compliance risks, and 4) reporting risks. In spite of the above, it was deduced that South African SMMEs use customised risk-management initiatives in the form of internal controls. Thus, the risk-management initiatives used by South African SMMEs were considered to be ineffective and/or inadequate. According to the analysed results, the overall financial sustainability of the sampled SMMEs was perceived to be moderate (above-average solvency, efficiency and liquidity; below-average profitability). This was not surprising, considering the numerous factors and risks which these entities identified as affecting their operations. The predominant economic factors which influenced sampled SMMEs' sustainability included the cost of electricity, inflation rates, food prices of suppliers, stiff competition, difficult access to finance (external funding), and increases in cost prices of products. Furthermore, it was found that the sustainability of these SMMEs was mostly influenced by operational risks (i.e. employee theft, and increases in cost prices of products and services) and strategic risks (i.e. stiff competition). The above results support the notion stemming from literature that the South African economic landscape is 'harsh'. In essence, without sound risk-management initiatives in place, the identified economic factors and risks have the capacity to influence the sustainability of these SMMEs adversely.

Initially, the risk-management initiatives used by sampled SMMEs suggested that these business entities made use of ERM. However, an examination of the respondents' perceptions regarding ERM revealed that these ERM-related initiatives were used by chance as opposed to choice. Thus, it was concluded that the risk-management initiatives used by sampled SMMEs were ineffective and/or inadequate, considering that these SMMEs had virtually no understanding of ERM. This supports the views from literature that South African SMMEs unknowingly engage in ERM initiatives.

Since a typical respondent had virtually no understanding of ERM, it can be concluded that these SMMEs, in effect, did not make proper use of ERM. This could be a plausible reason why these business entities had only moderate financial sustainability, since they could not effectively and adequately manage the economic factors and risks that negatively influenced their business operations and overall sustainability.

### 5.6 RECOMMENDATIONS

Taking into account the research conducted, the following propositions are recommended:

- Management of SMMEs should aim to acquire more knowledge and understanding of ERM. This can be achieved by attending risk-management training provided by reputable organisations like the Institute of Risk Management South Africa (IRMSA).
- Management of SMMEs should consult risk-management experts on an ongoing basis, particularly during the initial stages of ERM implementation.
- The government of South Africa should set up risk-management training institutions, exclusively for SMMEs. These institutions should conduct workshops on a regular basis, thereby assisting SMMEs to have sound risk-management initiatives in place.
- The government of South Africa should try to improve the economic landscape in which SMMEs operate, specifically in relation to inflation, interest rates, cost of electricity, crime rates, labour costs, and unemployment rates, to mention but few.
- The government of South Africa should offer financial support to those SMMEs intending to fund the implementation of ERM in their businesses.

### 5.7 AVENUES FOR FURTHER RESEARCH

Avenues for further research, inter alia, are listed below:

- To determine the impact of utilising ERM on the sustainability of South African SMMEs.
- To ascertain whether risk management education has a positive influence on the sustainability of South African SMMEs.
- To determine the effectiveness and/or adequacy of risk-management initiatives utilised by South African SMMEs.

#### REFERENCES

Abor, J. & Quartey, P. 2010. Issues in SME development in Ghana and South Africa. *International Research Journal of Finance and Economics*, 39:218-228.

Advisory, Conciliation and Arbitration Service. 2005. Employee communications and consultation. http://www.acas.org.uk/media/pdf/8/d/B06\_1.pdf [Accessed 22/072016].

Africagrowth Institute. 2012. South African SMME business confidence index report: 3<sup>rd</sup> quarter 2012. http://www.africagrowth.com/oct\_report\_12.pdf [Accessed 10/04/2014].

Aggaarwal, S. & Vijay, Y. 1998. Together with economics. New Delhi: Rachna Sagar.

Agrawal, R. 2016. 'Enterprise risk management' essential for survival and sustainable development of micro, small and medium enterprises. *International Review*, (1-2):117-124.

Agwu, M.O. & Emeti, C.I. 2014. Issues, challenges and prospects of small and medium scale enterprises (SMEs) in Port-Harcourt City, Nigeria. *European Journal of Sustainable Development*, 3(1):101-114.

Akintoye, A., Goulding, J. & Zawdie, G. 2012. *Construction innovation and process improvement*. Chichester: John Wiley.

Allan, N. & Beer, L. 2006. Strategic risks: it's all in your head. Working Paper Series 2006.01. School of Management, University of Bath. http://www.bath.ac.uk/management/research/pdf/2006-01.pdf [Accessed 14/02/2017].

Anane, G.K., Kobina, P.B. & Manu, J.K. 2013. Sustainability of small and medium scale enterprises in rural Ghana: the role of microfinance institutions. *Asian Economic and Financial Review*, 3(8):1003-1017.

Anderson, V. 2009. *Research methods in human resource management*, 2<sup>nd</sup> ed. London: Chartered Institute of Personnel and Development.

ACI Worldwide, Inc. 2011. Taking risk management from the silo across the enterprise. https://www.aciworldwide.com/-/media/files/collateral/trends/aci-taking-risk-mgmt-from-siloacross-enterprise-tl-us-0211-4572.pdf [Accessed 16/01/2015].

Aren, A.O. & Sibindi, A.B. 2014. Cash flow management practices: an empirical study of small businesses operating in the South African retail sector. *Risk Governance & Control: Financial Markets & Institutions*, 4(2):87-100.

Ashkenas, R. 2015. Jack Welch's approach to breaking down silos still works. *Harvard Business Review*. https://hbr.org/2015/09/jack-welchs-approach-to-breaking-down-silos-still-works [Accessed 10/08/2016].

Association of Chartered Certified Accountants. 2010. Small business: a global agenda. http://www.accaglobal.com/content/dam/acca/global/PDF-technical/small-business/pol-afb-sbaga.pdf [Accessed 10/08/2013].

Association of Chartered Certified Accountants. 2015. Strategic and operational risks. http://www.accaglobal.com/ie/en/student/exam-support-resources/professional-exams-study-resources/p1/technical-articles/strategic-operational-risks.html [Accessed 17/04/2017].

Aven, T. 2012. Foundations of risk analysis. Chichester: John Wiley.

Aven, T. 2016. Risk assessment and risk management: review of recent advances on their foundation. *European Journal of Operational Research*, 253(1):1-13.

Ayandibu, A.O. & Houghton, J. 2017. The role of small and medium scale enterprise in local economic development (LED). *Journal of Business and Retail Management Research*, 11(2):133-139.

Ayinde O.E., Omotesho, O.A. & Adewumi M.O. 2008. Risk attitudes and management strategies of small-scale crop producer in Kwara State, Nigeria: a ranking approach. *African Journal of Business Management*, 2(12):217-221, December.

Aziz, K.A. & Yazid, A.S. 2015. Enterprise risk management (ERM) practices among SMESs in Malaysia: a conceptual framework. *Proceedings of ICIC2015 – International Conference on Empowering Islamic Civilization in the 21st Century, Kuala Terengganu, Malaysia, 6–7 September 2015.* Kuala Terengganu, Malaysia: University Sultan Zainal Abidin: 296-300

Babbie, E.R. & Mouton, J. 2001. *The practice of social research*. Cape Town: Oxford University Press Southern Africa.

Baltimore County Public Schools. 2015. Develop a research proposal: planning the methodology – limitations and delimitations.

https://www.bcps.org/offices/lis/researchcourse/develop\_writing\_methodology\_limitations.html [Accessed 20/08/2016].

Bank for International Settlements. 2011. Principles for the sound management of operational risk. http://www.bis.org/publ/bcbs195.pdf [Accessed 19/02/2016].

Banking Association South Africa. 2017. SME enterprise. http://www.banking.org.za/what-we-do/sme-enterprise [Accessed 30/03/2017].

Bateh, J., Heaton, C., Arbogast, G.W. & Broadbent, A. 2013. Defining sustainability in the business setting. *Journal of Sustainability Management*, 1(1):1-4.

BBC News. 2015. Why is South Africa still so unequal? 13 May. http://www.bbc.com/news/world-africa-32623496 [Accessed 10/02/2016].

Beasley, M.S., Branson, B.C. & Hancock, B.V. 2010. Current state of enterprise risk oversight and market perceptions of COSO's ERM framework. https://www.coso.org/Documents/COSO-Survey-Report-FULL-Web-R6-FINAL-for-WEB-POSTING-111710.pdf [Accessed 31 August 2016].

Beasley, M.S., Branson, B.C. & Hancock, B.V. 2017. The state of risk oversight: an overview of enterprise risk management practices, 8<sup>th</sup> ed. https://erm.ncsu.edu/az/erm/i/chan/library/AICPA\_ERM\_INITIATIVE\_Research\_Study\_2017.pdf [Accessed 30/03/2017].

Beaumier, C & DeLoach, J. 2011. Ten common risk management failures and how to avoid them. *Business Credit*, 113(8):46-52, September–October.

Beck, M. & Kewell, B. 2014. *Risk: a study of its origins, history and politics*. Singapore: World Scientific.

Bernard, H.R. 2000. Social research methods: qualitative and quantitative approaches. Thousand Oaks, CA: Sage.

Bird, K. 2016. New handbook helps SMEs better manage risks. 18 January. http://www.iso.org/iso/home/news\_index/news\_archive/news.htm?Refid=Ref2034 [Accessed 12/07/2016].

Bond, P. 2016. Why South Africa should undo Mandela's economic deals. *Mail & Guardian*, 12 January. https://mg.co.za/article/2016-01-12-why-south-africa-should-undo-mandelas-economic-deals [Accessed 28/03/2017].

Braine, J.W. 2015. Why enterprise risk management is critical for small & mid-sized businesses. http://www.cfoedge.com/resources/articles/cfo-edge-enterprise-risk-management.pdf [Accessed 05/03/2016].

Brandt, R. 2010. A practical look at ERM programme development: RM strategy, risk management. *Enterprise Risk*, 4(10):26-27, November.

Brink, A., Cant, M. & Ligthelm, A. 2003. Problems experienced by small businesses in South Africa. Paper presented at the 16<sup>th</sup> Annual Conference of the Small Enterprise Association Australia and New Zealand (SEAANZ), Ballarat, Victoria, Australia, 28 September–1 October.

Broadleaf Capital International. 2007. Tutorial notes: The Australian and New Zealand standard on risk management, AS/NZS 4360:2004. http://www.ucop.edu/enterprise-risk-management/\_files/asnzs4360\_2004\_tut\_notes.pdf [Accessed 01/03/2016].

Broadleaf Capital International. 2014. Risk assessment and risk treatment. http://broadleaf.com.au/resource-material/risk-assessment-and-risk-treatment/ Accessed 01/03/2016.

Brockett, A.M. & Rezaee, Z. 2012. Corporate sustainability: integrating performance and reporting. Hoboken, NJ: John Wiley.

Bromiley, P., McShane, M., Nair, A. & Rustambekov, E. 2015. Enterprise risk management: review, critique, and research directions. *Long Range Planning*, 48(4):265-276.

Bruwer, J.P., Masama, B., Mgidi, A., Myezo, M., Nqayi, P., Nzuza, N., Phangwa, M., Sibanyoni, S. & Va, N. 2013. The need for a customised risk management framework for small enterprises. *Proceedings of the Southern African Accounting Association Biennial Conference, Somerset West, South Africa, 26–28 June 2013.* Cape Town: Southern African Accounting Association: 999-1030.

Bruwer, J.P. 2015. The influence of the control environment on the sustainability of fast food micro and very small enterprises operating in the northern suburbs. *Journal of Leadership and Management Studies*, 2(1):50-63, March.

Bruwer, J.P. 2016. The relationship(s) between the managerial conduct and the internal control activities of South African fast moving consumer goods SMMEs. Unpublished DTech (Internal Auditing) thesis, Cape Peninsula University of Technology, Cape Town, South Africa.

Bruwer, J.P. & Coetzee, P. 2016. A literature review of the sustainability, the managerial conduct of management and the internal control systems evident in South African small, medium and micro enterprises. *Problems and Perspectives in Management*, 14(2):201-211.

Bruwer, J.P. & Siwangaza, L. 2016. Is the control environment a basis for customised risk management initiatives in South African small, medium and micro enterprises? *Expert Journal of Business and Management*, 4(2):105-117.

Bruwer, J.P. & Van den Berg, A. 2017. The conduciveness of the South African economic environment and small, medium and micro enterprise sustainability: a literature review. *Expert Journal of Business and Management*, 5(1):1-12.

Burda, M. & Wyplosz, C. 2013. *Macroeconomics: a European text,* 6<sup>th</sup> ed. Oxford: Oxford University Press.

Bureau for Economic Research. 2016. The small, medium and micro enterprise sector of South Africa. Research note 2016/No.1.

http://www.seda.org.za/Publications/Publications/The%20Small,%20Medium%20and%20Micro%20E nterprise%20Sector%20of%20South%20Africa%20Commissioned%20by%20Seda.pdf [Accessed 13/08/2016].

Burton, L. 2007. Survey research: Choice of instrument, sample. http://ocw.jhsph.edu/courses/HSRE/PDFs/HSRE\_lect11\_burton.pdf [Accessed 29/03/2017].

*Business Dictionary.* 2016a. Definition of risk management. [http://www.businessdictionary.com/definition/risk-management.html [Accessed 23/02/2016].

*Business Dictionary*. 2016b. Definition of monitoring. http://www.businessdictionary.com/definition/monitoring.html [Accessed 08/02/2016].

Cagnin, F., Oliveira, M.C., Simon, A.T., Helleno, A.L. & Vendramini, M.P. 2016. Proposal of a method for selecting suppliers considering risk management: an application at the automotive industry. *International Journal of Quality & Reliability Management*, 33(4):488-498.

Cain, P., Brech, D., Barata, K., Reed, B., & Walford, J. 2014. Managing financial records. http://www.irmt.org/documents/educ\_training/public\_sector\_rec/IRMT\_principles.doc [Accessed 11/07/2016].

Callen, T. 2012. Gross domestic product: an economy's all. http://www.imf.org/external/pubs/ft/fandd/basics/gdp.htm [Accessed 16 May 2016].

*Cambridge Dictionary.* 2016. Economic environment. http://dictionary.cambridge.org/dictionary/english/economic-environment [Accessed 07/08/2016].

Cant, M.C. & Wiid, J.A. 2013. Establishing the challenges affecting South African SMEs. *International Business & Economics Research Journal*, 12(6):707-716, June.

Carroll, T. 2010. A holistic approach to business risk management. In *Approaches to Enterprise Risk Management*. London: Bloomsbury: 9-12.

Center for Audit Quality. 2012. Guide to internal control over financial reporting. Washington, DC: CAQ. http://www.thecaq.org/sites/default/files/caq\_icfr\_042513.pdf [Accessed 20/11/2016].

Chakabva, O. 2015. The effectiveness of risk management practices of small, medium and micro enterprises (SMMEs) which provide microfinance in the Cape Metropole, South Africa. Unpublished MTech (Internal Auditing) thesis, Cape Peninsula University of Technology, Cape Town, South Africa.

Chakraborty, A. 2015. Impact of poor accounting practices on the growth and sustainability of SMEs. *International Journal of Business & Management*, 3(5):227-231, May.

Chew, D.H. (ed.). 2012. *Corporate risk management: theory and practice*. New York, NY: Columbia University Press.

Chiu, I.H.Y. 2015. *Regulating (from) the inside: the legal framework for internal control in banks and financial institutions*. Oxford: Hart.

Chowdhury, A. 2008. Methods explained: the GDP implied deflator. *Economic & Labour Market Review*, 2(6): 53-56, June.

Clarno, A. 2017. *Neoliberal apartheid: Palestine/Israel and South Africa after 1994*. Chicago, IL: University of Chicago Press.

Collis, J. & Hussey, R. 2009. *Business research: a practical guide for undergraduate and postgraduate students*, 3<sup>rd</sup> ed. New York, NY: Palgrave Macmillan.

Conservation Gateway. 2017. Developing social objectives and indicators. https://www.conservationgateway.org/ConservationPractices/PeopleConservation/SocialScience/Mea suringSocialImpact/SocialObjectivesIndicators/Pages/developing-social-objecti.aspx [Accessed 17/04/2017].

Cooper, D.R. & Schindler, P.S. 2011. Business research methods, 11th ed. London: McGraw-Hill.

Cox, D.W. 2007. The cultural frontiers of total risk management. In Cox, D.W. (ed). *Frontiers of risk management: key issues and solutions*. London: Euromoney Books: 1-10.

Coyle, D. 2015. GDP: A brief but affectionate history. Princeton, NJ: Princeton University Press.

Creswell, J.W. 2014. *Research design: qualitative, quantitative, and mixed approaches*, 4<sup>th</sup> ed. London: Sage.

Cumming, K. 2012. Common information risks and why they could be happening to you. http://futureproof.records.nsw.gov.au/common-information-risks-and-why-they-could-be-happening-toyou/ [Accessed 19/02/2016].

Curkovic, S., Scannell, T., Wagner, B. & Vitek, M. 2013. A longitudinal study of supply chain risk management relative to COSO's enterprise risk management framework. *Modern Management Science & Engineering*, 1(1):13-36, February.

Curtis, P. & Carey, M. 2012. Risk assessment in practice. https://www2.deloitte.com/content/dam/Deloitte/global/Documents/Governance-Risk-Compliance/dttlgrc-riskassessmentinpractice.pdf [Accessed 19/02/2016].

Dabari, I.J & Saidin, S.Z. 2014. A theoretical framework on the level of risk management implementation in the Nigerian banking sector: the moderating effect of top management support. *Procedia – Social and Behavioral Sciences*, 164:627-634.

Davies, M. 2015. Making a risk matrix useful. http://causalcapital.blogspot.co.za/2015/08/making-risk-matrix-useful.html [Accessed 22/08/2016].

De Beer, A. 2008. What is business management? In Nieuwenhuizen, C. & Rossouw, D. (eds). *Business management: a contemporary approach*. Cape Town: Juta:1-38.

Delgado, M., Ketels, C., Porter, M.E. & Stern, S. 2012. The determinants of national competitiveness. Working Paper 18249. Washington, DC: National Bureau of Economic Research: 1-47.

Deloitte. 2013. Exploring strategic risk

https://www2.deloitte.com/content/dam/Deloitte/global/Documents/Governance-Risk-Compliance/dttl-grc-exploring-strategic-risk.pdf [Accessed 19/02/2016].

Department of Trade and Industry. 2013. SA Government is committed to supporting SMME development – Deputy Minister Thabethe. https://www.thedti.gov.za/editmedia.jsp?id=2681 [Accessed 25/01/16].

Diaz-Garrastacho, D. & Dickins, D.2011. ERM in a CPA practice. CPA Journal: 58-61, July.

Dictionary.com. 2017. Definition of factor analysis. http://www.dictionary.com/browse/factor-analysis [Accessed: 13/04/2017].

Dornberger, K., Oberlehner, S. & Zadrazil, N. 2014. Challenges in implementing enterprise risk management. *ACRN Journal of Finance and Risk Perspectives*, 3(3):1-14, November.

Drost, E.A. 2011. Validity and reliability in social science research. *Education Research and Perspectives*, (38)1:105-123, June.

Dumbravă, V. & Iacob, V.S. 2013. Using probability – impact matrix in analysis and risk assessment projects. *Journal of Knowledge Management, Economics and Information Technology (Special Issue)*, 3(6):76-96, December.

Duong, L. 2009. Effective risk management strategies for small-medium enterprises and micro companies: a case study for Viope Solutions Ltd. Unpublished thesis for degree in International Business, Arcada University of Applied Sciences, Helsinki, Finland.

Duren, C. 2016. Identify emerging risks that can affect the sustainability of your organization. http://riskarticles.com/identify-emerging-risks-that-can-affect-the-sustainability-of-your-organization/ [Accessed 28/04/2016].

Durrheim, K. 2006. Research design. In Terre Blanche, M.T., Durrheim, K. & Painter D. (eds). *Research in practice: applied methods for the social sciences*, 2<sup>nd</sup> rev. ed. Cape Town: University of Cape Town Press: 33-59.

Dwivedi, D.N. 2002. *Microeconomics: theory and applications*. New Delhi: Pearson Education.

Dwyer, J. 2012. *Communication for business and the professions: strategies and skills*. Sydney: Pearson Higher Education Australia.

Dzansi, D.Y. 2004. Social responsibility of SMMEs in rural communities. Unpublished PhD (Entrepreneurship) thesis, University of Pretoria, South Africa.

Economic Cooperation Organization Trade and Development Bank. 2007. Operational and compliance risk management. http://www.etdb.org/content/operationalandcomplianceriskmanagement [Accessed 11/07/2016].

Edmonds, W.A. & Kennedy, T.D. 2012. An applied reference guide to research designs: quantitative, qualitative, and mixed methods: quantitative, qualitative, and mixed methods. Thousand Oaks, CA: Sage.

Endicott, M. & Gardiner, T. 2011. ERM and sustainability: beyond buzzwords. *Canadian Underwriter*, 78(3):20-24, March.

Enow, S.T. & Brijlal, P. 2014. The effect of working capital management on profitability: the case of small medium and micro enterprises in South Africa. *Journal of Accounting and Management*, 4(2):7-15.

Enow, S.T. & Kamala, P. 2016. Cash management practices of small, medium and micro enterprises in the Cape Metropolis, South Africa. *Investment Management and Financial Innovations*, 13(1):230-236.

*Entrepreneur Magazine*. 2009. Insatiable appetite for convenience – be your own boss annual: 37-39, January.

Epstein, M.J. & Buhovac, A.R. 2006. Management accounting guideline: the reporting of organizational risks for internal and external decision-making. http://www.cimaglobal.com/Documents/ImportedDocuments/Tech\_MAG\_Reporting\_Organisational\_R isks\_for\_Decision\_Making\_Sept06.pdf [Accessed 19/02/2016].

Erard, D. 2010. Understanding & implementation of ISO 31000:2009. http://www.iso.org/iso/livelinkgetfile?IINodeId=327278&IIVoIId=-2000 [Accessed 24/02/2016].

Fabrigar, L.R. & Wegener, D.T. 2011. *Exploratory factor analysis*. New York, NY: Oxford University Press.

Fagerberg, J. 2013. The changing global economic landscape: the factors that matter. In Solow, R.M. & Touffut, J.P. *The shape of the division of labour: nations, industries and households*. Cheltenham: Edward Elgar: 6-31. http://www.sv.uio.no/tik/InnoWP/20130201\_fagerberg.pdf [Accessed 17 August 2016].

Fatoki, O. 2014. The causes of the failure of new small and medium enterprises in South Africa. *Mediterranean Journal of Social Sciences*, 5(20):922-927, September.

Fernando, A.C. 2011. *Business Environment*. New Delhi: Pearson Education.

*Financial Dictionary*. 2016. Prime rate. http://financial-dictionary.thefreedictionary.com/prime+rate [Accessed 17/02/2016].

*Financial Dictionary*. 2017. Business failure. http://financialdictionary.thefreedictionary.com/Business+failure [Accessed 22/04/2017].

FinScope. 2010. FinScope South Africa small business survey 2010. http://led.co.za/sites/default/files/Brochure FS SMME 10.pdf [Accessed 23/01/2016].

Firoozye, N.B. & Ariff, F. 2015. *Managing uncertainty, mitigating risk: tackling the unknown in financial risk assessment and decision making*. London: Palgrave Macmillan.

Flouris, T.G. & Yilmaz, A.K. 2016. *Risk management and corporate sustainability in aviation*. Abingdon: Routledge.

Forslund, D. 2016. The World Bank and inequality in South Africa. http://roape.net/2016/04/19/world-bank-inequality-south-africa/ [Accessed 08/08/2016].

Friedrich, C. 2016. Why do 70% to 80 % of small businesses fail within five years? https://www.moneyweb.co.za/mybusiness/why-do-70-to-80-of-small-businesses-fail-within-five-years/ [Accessed 30/03/2017].

Frigo, M.L. & Anderson, R.S. 2011. Embracing enterprise risk management. http://www.coso.org/documents/EmbracingERMGettingStartedforWebPostingDec110\_000.pdf [Accessed 10/01/2014].

Garrido, M.C., Ruotolo, M.C.A., Ribeiro, F.M.L. & Naked, H.A. 2011. Risk identification techniques knowledge and application in the Brazilian construction. *Journal of Civil Engineering and Construction Technology*, 2(11):242-252, November.

Gates, S., Nicolas, J.L. & Walker, P.L. 2012. Enterprise risk management: a process for enhanced management and improved performance. *Management Accounting Quarterly*, 13(3):28-38, Spring.

Gelinas, U.J., Dull, R.B. & Wheeler, P. 2014. *Accounting information systems*, 10<sup>th</sup> ed. Melbourne: Cengage Learning Australia.

Gilbert, J.B. 2007.Enterprise risk management: the new imperative. An executive White Paper. www.lexicon-systems.com/.../EnterpriseRiskManagementNew... -United States [Accessed 28/01/2013].

Global Entrepreneurship Monitor. 2017. What is the National Expert Survey (NES)? http://www.gemconsortium.org/wiki/1142 [Accessed 17/04/2017].

Glowik, M. & Smyczek, S. 2015. *Healthcare: market dynamics, policies and strategies in Europe.* Berlin: Walter de Gruyter Oldenbourg.

Godbole, P. 2012. Effective risk management is crucial for small businesses: Mritunjay Kapur. *Business Standard*, January 17. http://www.business-standard.com/article/sme/effective-risk-management-is-crucial-for-small-businesses-mritunjay-kapur-112011700102\_1.html [Accessed 10/02/2014].

Gollin, D. 2008. 'Nobody's business but my own': self-employment and small enterprise in economic development. *Journal of Monetary Economics*, 55(2):219–233.

Goodwin, C.J. & Goodwin, K.A. 2016. *Research in psychology methods and design*, 8<sup>th</sup> ed. Hoboken, NJ: John Wiley.

Gossow, C., Liprini, R. & Ordman, J. 2015. The conduciveness of starting and conducting business in Africa: an SME landscape study. http://thinkroom.co.za/Reports/Landscape\_Report\_Sum\_Digital.pdf [Accessed 16/02/2016].

Graham, L. 2015. Internal control audit and compliance: documentation and testing under the new COSO framework. Hoboken, NJ: John Wiley.

Grant, L. 2015. Infographic: Poverty in South Africa. *Mail & Guardian*, 5 February. https://mg.co.za/data/2015-02-05-infographic-poverty-in-south-africa [Accessed 28/03/2017].

Gravetter, F.J. & Wallnau, L.B. 2013. *Essentials of statistics for the behavioral sciences*, 8<sup>th</sup> ed. Belmont, CA: Wadsworth Cengage Learning.

Green, P.E.J. 2015. *Enterprise risk management: a common framework for the entire organization*. Amsterdam: Butterworth-Heinemann.

Groepe, F. 2015. The role of small business in the economy. Paper presented at the 25<sup>th</sup> AHI Conference: The Role of Business in Local Government and Local Economic Development, 6–9 October 2015, George, South Africa.

https://www.resbank.co.za/Lists/Speeches/Attachments/452/Role%20of%20small%20business%2020 15%20.pdf [Accessed 18/04/2017].

Gropper, S. & Smith J. 2008. *Advanced nutrition and human metabolism*, 5<sup>th</sup> ed. Belmont, CA: Cengage Learning.

Grose, V.L. 2014. Risk management: a systemic approach. https://www.irmi.com/articles/expert-commentary/risk-has-always-been-there [Accessed 23/02/2016].

Guimond, M., Taza, C. & Loranger, J.M. 2010. Risk management – a hot topic. http://www.lexology.com/library/detail.aspx?g=c1eec052-4540-4933-a7ba-f6ebb6f94e09 [Accessed 02/06/2017].

Haimes, Y.Y. (ed.). 2015. *Risk modeling, assessment, and management*, 4<sup>th</sup> ed. Hoboken, NJ: John Wiley.

Hair, J.F. (Jnr)., Babin, B., Money, A.H. & Samouel, P. 2003. *Essentials of business research methods*. Hoboken, NJ: John Wiley.

Hampton, J.J. 2014. Fundamentals of enterprise risk management: how top companies assess risk, manage exposure, and seize opportunity. New York, NY: AMACOM.

Haran, P. 2014. Driving a robust risk.

https://www.glanbia.com/~/media/Files/G/GlanbiaPlc/documents/Annual%20Report%202014/Glanbia \_AR2014\_risk\_management.pdf [Accessed 19/02/2016].

Harrer, J. 2008. Internal control strategies: a mid to small business guide. Hoboken, NJ: John Wiley.

Harvey, J. 2008. Enterprise risk management.

http://www.cimaglobal.com/Documents/ImportedDocuments/cid\_tg\_enterprise\_risk\_management\_jul0 8.pdf.pdf [Accessed 30 August 2016].

Havenga, A. 2006. Value of enterprise risk management in the South African business environment. Unpublished MBL research report, University of South Africa, Pretoria.

Hayne, C. & Free, C. 2014. Hybridized professional groups and institutional work: COSO and the rise of enterprise risk management. *Accounting, Organizations and Society,* 39(5):309-330, July.

Heale, R. & Twycross, A. 2015. Validity and reliability in quantitative studies. *Evidence-Based Nursing*, 18(3):66-67, July.

Hendricks, B., Hendricks, G., Johnson, R., Manuel, A., Maritz, R., Theron, L. & Bruwer, J.P. 2015. The influence of "sin tax" on the economic sustainability of small, medium and micro liquor store enterprises in the northern suburbs of the Cape Metropolis, South Africa. *Journal of Economics*, 6(2):87-101.

Hernandez, S. (ed.). 2016. Official (ISC) 2 guide to the CISSP CBK. Boca Raton, FL: CRC Press.

Herrington, M. & Kew, J. 2017. Global Entrepreneurship Monitor. Global report: 2016–2017. http://www.gemconsortium.org/report/49812 [Accessed 15/03/2017].

Herrington, M., Kew, J. & Kew, P. 2015. 2014 Global Entrepreneurship Monitor report: South Africa. http://www.gemconsortium.org/report [Accessed 20/11/15].

Hillson, D. 2003. *Effective opportunity management for projects: exploiting positive risk*. Boca Raton, FL: CRC Press.

Hohenwarter, A. 2014. Enterprise Risk Management vs. Traditional risk management – The same... but different. http://www.enterpriseriskmgt.com/boots-on-the-street-erm-vs-trm-the-same-but-different/ [Accessed 01/03/2016].

Høj, J.C, & Lewis, C. 2015. Organisation for Economic Co-operation and Development economic surveys: South Africa 2015.

http://www.treasury.gov.za/publications/other/OECD%20Economic%20Surveys%20South%20Africa %202015.pdf [Accessed 07/08/ 2016].

Holmes, T. 2016. SA's ferocious fast food appetite. *Mail and Guardian*, 8 April. https://mg.co.za/article/2016-04-11-sa-has-an-appetite-for-fast-food [Accessed 16/05/2017].

Hopkin, P. 2014. *Fundamentals of risk management: understanding, evaluating and implementing effective risk management*, 3<sup>rd</sup> ed. London: Kogan Page.

Hopkins, M. 2012. Corporate social responsibility and international development: is business the solution? London: Earthscan.

Howard, J.L. & Jawahar, I.M. 2002. Risk management for small businesses. *The Entrepreneurial Executive*, 7:95-114.

Hoyt, R.E. & Liebenberg, A.P. 2011. The value of enterprise risk management. *The Journal of Risk and Insurance*, 78(4):795-822.

Hubbard, J. & Ashton, M. 2013. Five ways to sex up small business. *Finweek*: 9-15, 4 April. http://www.fetola.co.za/wp-content/uploads/2013/05/Finweek-April-20132.pdf [Accessed 17/04/2017].

Hussain, I., Farooq Z. & Akhtar W. 2012.SMEs development and failure avoidance in developing countries through public private partnership. *African Journal of Business Management*, 6(4):1581-1589, February.

Hussain, S. & Shafi, M. 2014. Operational risk in banks: a review. ELK Asia Pacific Journal of Frinance and Risk Management, 5(2).

http://www.elkjournals.com/MasterAdmin/UploadFolder/4.%20OPERATIONAL%20RISK%20IN%20B ANKS%20A%20REVIEW%20-

<u>%20Final/4.%20OPERATIONAL%20RISK%20IN%20BANKS%20A%20REVIEW%20-%20Final.pdf</u> [Accessed 10/06/2016].

Hyden, T. 2011. Risk management: planning to avoid losses. *Fire Engineering*, 164(10): 99-101.

Industrial Development Corporation. 2015. Economic Overview: Recent developments in the global and South African economies. http://www.idc.co.za/images/download-files/economic-overview\_jun\_2015.pdf [Accessed 17/02/2016].

Inflation.eu. 2017. Historic inflation South Africa – CPI inflation. http://www.inflation.eu/inflation-rates/south-africa/historic-inflation/cpi-inflation-south-africa.aspx [Accessed 30/03/2017].

Institute for Digital Research and Education. 2017. What does Cronbach's alpha mean? https://stats.idre.ucla.edu/spss/faq/what-does-cronbachs-alpha-mean/ [Accessed 05/04/2017].

Institute of Directors in Southern Africa. 2009. Draft code of governance principles for South Africa – 2009. http://admin.sun.ac.za/admin/Finance/new/docs/king%20III.pdf [Accessed 19/08/2016].

Institute of Directors in Southern Africa. 2010. Governance in SMEs: a guide to the application of corporate governance in small and medium enterprises. https://c.ymcdn.com/sites/www.iodsa.co.za/resource/resmgr/Docs/GovernanceinSMEsGuidelowres.p df [Accessed 08/03/2016].

Institute of Risk Management. 2002. A structured approach to Enterprise Risk Management (ERM) and the requirements of ISO 31000. https://www.theirm.org/media/886062/ISO3100\_doc.pdf [Accessed 01/03/2016].

Institute of Risk Management. 2016. About risk management – What is risk management? https://www.theirm.org/about/risk-management/ [Accessed 30August 2016].

Institute of Risk Management South Africa. 2014. Are you a 'risk-a-holic' or 'risk-denialist'? http://www.irmsa.org.za/blogpost/951512/190289/Are-you-a-risk-a-holic-or-risk-denialist [Accessed 12/08/2016]

International Labour Organization. 2015. World employment and social outlook: Trends 2015. http://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/--publ/documents/publication/wcms\_337069.pdf [Accessed 17/04/2017].

International Monetary Fund. 2016. Report for selected countries and subjects. http://www.imf.org/external/pubs/ft/weo/2015/02/weodata/weorept.aspx?pr.x=44&pr.y=12&sy=2013&e y=2020&scsm=1&ssd=1&sort=country&ds=.&br=1&c=199&s=NGDPDPC&grp=0&a= [Accessed 09/02/2016].

International Organization for Standardization. 2009. ISO 3100:2009: Risk management – principles and guidelines. https://www.iso.org/obp/ui/#iso:std:iso:31000:ed-1:v1:en [Accessed 19/02/2016].

International Organization for Standardization. 2014. Plain English ISO 31000 2009: Risk management dictionary. http://www.praxiom.com/iso-31000-terms.htm [Accessed 23/02/2016].

International Risk Governance Council. 2005. An introduction to the IRGC Risk Governance Framework.

http://www.irgc.org/IMG/pdf/An\_introduction\_to\_the\_IRGC\_Risk\_Governance\_Framework.pdf [Accessed 19/02/2016].

International Risk Management Institute. 2016. Pure risk. https://www.irmi.com/online/insurance-glossary/terms/p/pure-risk.aspx [Accessed 24/07/2016].

Janakiraman, H. 2015. Managing risk for your enterprise: KPMG's Harikrishnan Janakiraman explains how. *Entrepreneur Middle East*, 21 July. http://www.entrepreneur.com/article/248611 [Accessed 08/03/2016[.

Johnson & Johnson. 2013. Framework for enterprise risk management. https://www.jnj.com/sites/default/files/pdf/JnJ\_RiskMgmt\_ERMFramework\_guide\_v16a.pdf [Accessed 19 August 2016]. Jones, S. 2017. No escape from trap of poverty, unemployment. *Cape Argus*, 22 February. http://www.pressreader.com/south-africa/cape-argus/20170222/281835758467589 [Accessed 30/03/2017].

Jules Halpern Associates. 2010. Why is confidentiality important? https://www.halpernadvisors.com/why-is-confidentiality-important/ [Accessed 11/07/2016].

Kadocsa, G. & Francsovics, A. 2011. Macro and micro economic factors of small enterprise competitiveness. *Acta Polytechnica Hungarica*, 8(1):23-40.

Kanona, R.M. & Tawalbeh, L.A. 2007. The risk management process. http://www.just.edu.jo/~tawalbeh/aabfs/iss6753/presentations/RMP.ppt [Accessed 22 August 2016].

Kaplan, R.S. & Mikes, A. 2012. Managing risks: a new framework. *Harvard Business Review*, 90(6), June. https://hbr.org/2012/06/managing-risks-a-new-framework [Accessed 01/03/2016].

Karpak, B. & Topcu, I. 2010. Small medium manufacturing enterprises in Turkey: an analytic network process framework for prioritizing factors affecting success. *International Journal of Production Economics*, 125(1):60-70, May.

Keeton, G. 2014. Inequality in South Africa. *The Journal of the Helen Suzman Foundation*, (74):26-31, November.

Kelliher, P.O.J., Acharyya, M., Couper, A., Grant, K., Maguire, E., Nicholas, P., Smerald, C., Stevenson, D., Thirlwell, J. & Cantle, N. 2016. Good practice guide to setting inputs for operational risk models. https://www.actuaries.org.uk/documents/good-practice-guide-setting-inputs-operational-risk-models [Accessed 28/06/2016].

Kelley, D., Singer, S. & Herrington, M. 2016. Global Entrepreneurship Monitor: 2015/16 global report. http://www.gemconsortium.org/report [Accessed 09/02/2016].

Kengne, S. 2015. Inhibiting factors in the strategic financial management decision making process: evidence from South African SMMEs. *Mediterranean Journal of Social Sciences*, 6(2):113-124, March.

Kganyago, L. 2016. Statement of the Monetary Policy Committee – South African Reserve Bank. 19 May.

https://www.resbank.co.za/Lists/News%20and%20Publications/Attachments/7063/MPC%20Statemen t%2028%20January%202016.pdf [Accessed 17/02/2016].

Knoema. 2016. GDP per capita ranking 2015 | data and charts. https://knoema.com/sijweyg/gdp-per-capita-ranking-2015-data-and-charts [AcCessed 17/05/2016].

Knoema. 2017. GDP per capita ranking 2016: data and charts. https://knoema.com/sijweyg/gdp-per-capita-ranking-2016-data-and-charts [Accessed 30/03/2017].

Knox, V. 2012. Compliance and risk management for SMEs. 2 July. http://www.fin24.com/Entrepreneurs/Compliance-and-risk-management-for-SMEs-20120702 [Accessed 20/10/2012].

Kokemulle, N. 2016. Six micro environmental factors that affect businesses. *Houston Chronicle*. http://smallbusiness.chron.com/six-microenvironmental-factors-affect-businesses-78023.html [Accessed 10/01/2017].

Koralun-Bereźnicka, J.K. 2013. *Corporate performance: a ratio-based approach to country and industry analyses*. Heidelberg: Springer.

Kotelnikov, V. 2013. Small and medium enterprises and ICT/SME adoption of ICT. https://en.wikibooks.org/wiki/Small\_and\_Medium\_Enterprises\_and\_ICT/SME\_Adoption\_of\_ICT [Accessed 17/02/2016]. Kouns, J. & Minoli, D. 2011. Information technology risk management in enterprise environments: a review of industry practices and a practical guide to risk management teams. Chichester: John Wiley.

Kraja, Y.B. & Osmani, E. 2015. Importance of external and internal environment in creation of competitive advantage to SMEs. (Case of SMEs, in the northern region of Albania.) *European Scientific Journal*, 11(13):120-130, May.

Krause, M. 2015. Integration von Risikomanagement in Planungs-, Steuerungs- Und Kontrollsysteme: Die Synthese von COSO Enterprise Risk Management und Balanced Scorecard. Hamburg: Igel Verlag.

Krishnaswamy, K.N., Sivakumar, A.I. & Mathirajan, M. 2009. *Management research methodology: integration of principles, methods and techniques*. New Delhi: Pearson Education.

Krosinsky, C. 2013. Sustainability and systemic risk: what's the SEC's role? *The Guardian*, 23 October. https://www.theguardian.com/sustainable-business/sustainability-risk-investors-sec [Accessed 10/08/2016].

Kula, M.C., Panday, P. & Mantia, K. 2010. Real GDP, well-being, and happiness. *International Journal of Arts and Sciences* 3(7):431-443.

Kupka, J. & Thomas, A. 2014. Anti-competitive behaviour in the agri-food and steel value chains in the South African manufacturing sector. *Journal of Economic and Financial Sciences*, 7(2):315-340, July.

Lam, J. 2014. *Enterprise risk management: from incentives to controls*, 2<sup>nd</sup> ed. Hoboken, NJ: John Wiley.

Lam, J. 2017. *Implementing enterprise risk management: from methods to applications*. Hoboken, NJ: John Wiley.

Landman, J.P., Bhorat, H., Van der Berg, S. & Van Aardt, C. 2003. Breaking the grip of poverty and inequality in South Africa 2004–2014: current trends, issues and future policy options. EFSA report on poverty & inequality in South Africa. http://www.sarpn.org/documents/d0000649/P661-Povertyreport3b.pdf [Accessed 18/04/2017].

Laughland, P. & Bansal, T. 2011. The top ten reasons why businesses aren't more sustainable. *Ivey Business Journal*: 12-19, January/February. http://iveybusinessjournal.com/publication/the-top-ten-reasons-why-businesses-arent-more-sustainable/ [Accessed 19 August 2016].

Leedy, P.D. & Ormrod, J.E. 2005. *Practical research: planning and design*, 8<sup>th</sup> ed. Upper Saddle River, NJ: Merrill Prentice Hall.

Lehohla, P. 2014. Poverty trends in South Africa: an examination of absolute poverty between 2006 and 2011. Statistics South Africa. Report No. 03-10-06. Pretoria: Statistics South Africa. http://www.statssa.gov.za/publications/Report-03-10-06/Report-03-10-06March2014.pdf [Accessed 25/01/2016].

Lekhanya, L.M. 2015. Public outlook on small and medium enterprises as a strategic tool for economic growth and job creation in South Africa. *Journal of Governance and Regulation*, 4(4):412-418.

León, P. 2001. Four pillars of financial sustainability. Arlington, VA: Nature Conservancy. http://pdf.usaid.gov/pdf\_docs/Pnadf342.pdf [Accessed 12/08/2016].

Leung, L. 2015. Validity, reliability, and generalizability in qualitative research. *Journal of Family Medicine and Primary Care*, 4(3):324-327, July–September.

Leverty, T. 2015. How successful companies manage risk. https://bus.wisc.edu/knowledge-expertise/faculty-research/forward-thinking-blog/2015/07/17/how-successful-companies-manage-risk [Accessed 23/02/2016].

Lewis, N.da C. 2012. The fundamental rules of risk management. Boca Raton, FL: CRC Press.

Liberta, 2016. Prime interest rate in South Africa. http://liberta.co.za/blog/prime-interest-rate-in-south-africa-current-and-historical/ [Accessed 17/02/2017].

Liebenberg, A.P. & Hoyt, R.E. 2003. The determinants of enterprise risk management: evidence from the appointment of chief risk officers. *Risk Management and Insurance Review*, 6(1):37-52.

Ligthelm, A.A. 2010. Entrepreneurship and small business sustainability. *Southern African Business Review*, 14(3):131-153.

Lin, L., Horng, J.S., Chen, Y.C. & Tsai, C.Y. 2011. Factors affecting hotel human resource demand in Taiwan. *International Journal of Hospitality Management*, 30(2):312-318, June.

Longenecker, J., Petty, J., Palich, L. & Hoy, F. 2013. *Small business management*, 17<sup>th</sup> ed. Stamford, CT: Cengage Learning .

Louisot, J.P. & Ketcham, C. 2014. *ERM – Enterprise risk management: issues and cases.* Chichester: John Wiley.

Lundquist, A.E. 2014. Lessons from the academy: ERM implementation in the university setting. In: Fraser, J.R.S., Simkins, B.J. & Narvaez, K. (eds). *Implementing enterprise risk management: case studies and best practices*. Hoboken, NJ: John Wiley: 143-178.

Lussier, R.N. 2008. *Management fundamentals: concepts, applications, skill development*, 4<sup>th</sup> ed. Mason, OH: South-Western/Cengage Learning.

Mafini, C. & Muposhi, A. 2017. The impact of green supply chain management in small to medium enterprises: cross-sectional evidence. *Journal of Transport and Supply Chain Management*, 11(0): Art 270.

Magan, R. 2016. Internal control and risk management framework. http://www.slideshare.net/RahulMaganMBAFinance2/internal-control-risk-management-framework-pdf-version-63880480 [Accessed 10 September 2016].

Maharaj, Y. 2015. Fast food industry report South Africa: Is SA's love for fast food defying Banting? http://www.insightsurvey.co.za/blog/is-sas-love-for-fast-food-defying-banting [Accessed 15/12/2015].

Makakane, M. 2014. Do SMMEs play a role in the South African economy? http://morongwamakakane.co.za/smmes-play-role-south-african-economy/ [Accessed 10/02/2016].

Makos, J. 2015. What is environmental analysis? http://pestleanalysis.com/what-is-environmentalanalysis/ [Accessed 11/02/2016].

Mandel, C. 2008. Enterprise risk management is here to stay. *Business Insurance*, 42(15):10, April.

Masama, B.T., Ndlovu, E., Mambwe, T., Rabohome, C., Chakabva, O., Fologang, B., Badze, T. & Bruwer, J.P. 2012. Enterprise risk management: a managing 'partner' for business success. *African Journal of Business Management*, 6(48):11782-11786, December.

Masutha, M. & Rogerson, C.M. 2014. Small enterprise development in South Africa: the role of business incubators. *Bulletin of Geography. Socio-economic Series* (26):141-155.

Maumbe, B. 2012. The rise of South Africa's quick service restaurant industry. *Journal of Agribusiness in Developing and Emerging Economies*, 2(2):147-166.

Maye, M. 2014. Small business in South Africa: What the Department of Small Business Development can do to stimulate growth. Pretoria: Southern African Catholic Bishops' Conference Parliamentary Liaison Office. Occasional Paper 35, September.

Mbuyisa, B. & Leonard, A. 2015. ICT adoption in SMEs for the alleviation of poverty. *Proceedings of the International Association for Management of Technology (IAMOT) 2015 Conference, Cape Town, 8–11 June 2015.* Pretoria: University of Pretoria: 858-878.

McGroarty, P. 2013. Poverty still plagues South Africa's black majority. *Wall Street Journal*, 8 December. http://www.wsj.com/articles/SB10001424052702303560204579246300838033732 [Accessed 27/01/16].

McKay, S. 2016. *Risk assessment for mid-sized organisations:* COSO tools for a tailored approach, 2<sup>nd</sup> ed. Hoboken, NJ: John Wiley.

McKimm, J., Forrest, K. & Thistlethwaite, J. 2017. *Medical education at a glance*. Hoboken, NJ: Wiley Blackwell.

McNally, J.S. & Tophoff, V. 2015. Stacked in your favor: you can increase your chances for organizational success by upgrading your risk management and internal control. *Strategic Finance*, 96(10):30-39, April.

McNeil, A.J., Frey, R. & Embrechts, P. 2015. *Quantitative risk management: concepts, techniques and tools.* Princeton, NJ: Princeton University Press.

McTaggart, D., Findlay, C. & Parkin, M. 2013. *Economics*, 7<sup>th</sup> ed. Sydney: Pearson Higher Education Australia.

Merriam-Webster. 2016a. Definition of 'factor'. https://www.merriam-webster.com/dictionary/factor [Accessed 17/08/2016].

Merriam-Webster. 2016b. Definition of 'sustainable'. http://www.merriam-webster.com/dictionary/sustainable [Accessed 08/02/2016].

Merriam-Webster. 2017a. Definition of 'adequate'. https://www.merriam-webster.com/dictionary/adequate [Accessed 12/04/2017].

Merriam-Webster. 2017b. Definition of 'customize'. https://www.merriam-webster.com/dictionary/customized [Accessed 14/04/2017].

Merriam-Webster. 2017c. Definition of 'effective'. https://www.merriam-webster.com/dictionary/effective [Accessed 12/04/2017].

Meyer, M., Roodt, G. & Robbins, M. 2011. Human resources risk management: governing people risks for improved performance. *SA Journal of Human Resource Management, 9*(1):366-378.

Miles, D.A. 2011. *Risk factors and business models: understanding the five forces of entrepreneurial risk and the causes of business failure.* Boca Raton, FL: Universal.

Mishra, C.S. 2015. *Getting funded: proof-of-concept, due diligence, risk and reward*. New York, NY: Palgrave Macmillan.

Moeller, R.R. 2011. COSO enterprise risk management: establishing effective governance, risk, and compliance (GRC) processes, 2<sup>nd</sup> ed. Hoboken, NJ: John Wiley.

Moeller, R.R. 2013. *Executive's guide to COSO internal controls: understanding and implementing the new framework*. Hoboken, NJ: John Wiley.

Moeller, R.R. 2016. *Brink's modern internal auditing: a common body of knowledge*, 8<sup>th</sup> ed. Hoboken, NJ: John Wiley.

Mohammed, A. & Sykes, R. 2012. Sharpening strategic risk management. http://www.pwc.com/gx/en/services/advisory/consulting/risk/resilience/publications/sharpeningstrategic-risk-management.html [Accessed 09/09/2016].

Morgan, S.L. & Waring, C.G. 2004. Guidance on testing data reliability. Austin, TX: Office of the City Auditor.

http://www.auditorroles.org/files/toolkit/role2/Tool2aAustinCityAud\_GuidanceTestingReliability.pdf [Accessed 19/02/2016].

Mouton, J. 2001. *How to succeed in your master's and doctoral studies: a South African guide and resource book.* Pretoria: Van Schaik.

Mthabela, T.E. 2015. Assessing the causal failures of emerging manufacturing SMEs in Johannesburg. Unpublished Master of Management (Entrepreneurship and New Venture Creation) research report, University of the Witwatersrand, Johannesburg, South Africa.

Muehlenbrock, S. Messini, F. & Segui, B. 2012. KPMG Business Dialogue: Operational risk. https://www.kpmg.com/lu/en/services/advisory/riskconsulting/financialregulatoryreporting/documents/operational-risk.pdf [Accessed 19/02/2016].

Murisa, T. & Chikweche, T. 2015. *Beyond the crises: Zimbabwe's prospects for transformation*. Harare: Weaver Press.

Murray, F. 2017. South African food service industry report 2016. http://www.franchisedirect.co.za/information/southafricanfoodserviceindustryreport2016/?r=5798 [Accessed 17/05/2017]

Musa, H. & Chinniah, M. 2016. Malaysian SMEs development: future and challenges on going green. *Procedia – Social and Behavioral Sciences*, 224:254-262.

Musgrave, A. 2015. More people living below breadline – Stats SA. *The Star – IOL*, 4 February. http://www.iol.co.za/business-report/economy/more-people-living-below-breadline---stats-sa-1813099 [Accessed 28/03/2017[.

Mwangi, M.L.K. 2014. The relationship between risk management and financial performance of the top 100 small and medium enterprises in Kenya. Unpublished Master of Science (Finance) research report, University of Nairobi, Kenya.

Naidoo R. & Urban B. 2010. The relevance of operational skills towards business sustainability: a focus on SMME manufacturers in the Vaal triangle region. *Acta Commercii*, 10(1):234-248.

National Planning Commission. 2012. Our future: make it work: National Development Plan 2030. Pretoria: National Planning Commission.

Nedbank. 2017. Labour force survey – Q4 2016. https://www.nedbank.co.za/content/dam/nedbank/siteassets/AboutUs/Economics\_Unit/Research/Eco nomic%20Commentary/QLFS\_Q4\_2016.pdf [Accessed 30/03/2017].

Neiman, M. 2010. Prime rate: what is it and how is it set? http://www.bankrate.com/finance/mortgages/prime-rate-what-is-it-and-how-is-it-set.aspx [Accessed 08/09/2016].

Neneh, N.B. & Van Zyl, J.H. 2012. Achieving optimal business performance through business practices: evidence from SMEs in selected areas in South Africa. *South Africa Business Review*, 16(3):118-144.

Ngary, C., Smit, Y., Bruwer, J.P. & Ukpere, W.I. 2014. Financial performance measures and business objectives attainment in fast food SMMEs in the Cape Metropolis: a preliminary liability and suitability analysis. *Mediterranean Journal of Social Sciences*, 5(20):909-921.
Nichifor, M.A. 2016. Specific risk dimension in the wind energy field: case study – Romania. Bilgrin, M.H. & Danis, H. (eds). *Proceedings of the 15<sup>th</sup> Eurasia Business and Economics Society Conference, 8–10 January, Lisbon, Portugal. Vol 2: Entrepreneurship, Business and Economics.* Cham: Springer: 247-258.

Nicolson, G. 2015. South Africa: where 12 million live in extreme poverty. *Daily Maverick*. http://www.bankrate.com/finance/mortgages/prime-rate-what-is-it-and-how-is-it-set.aspx I [Accessed 10/02/2016].

Nicolson, G. 2017. Small business SA: Waiting to be truly free. *Daily Maverick*, 26 March. https://www.dailymaverick.co.za/article/2015-03-26-small-business-sa-waiting-to-be-truly-free/#.WOtW-UWGPIU [Accessed 10/01/2017].

Nieman, G. 2001. Training entrepreneurs and small business enterprises in South Africa: a situational analysis. *Education* + *Training*, 43(8/9):445-450.

Norman, L. 2016. What are the four basic functions that make up the management process? *Houston Chronicle*. http://smallbusiness.chron.com/four-basic-functions-make-up-management-process-23852.html [Accessed 20/01/16].

Nwankwo, S. & Aiyeku, J. 2015. Supplier selection in business-to-business markets: scope for applying artificial intelligence systems. Spotts, H.E. & Meadow, H.L. (eds). *Proceedings of the 2000 Academy of Marketing Science (AMS) Annual Conference: Developments in Marketing Science, Montreal, Canada, 24–27 May.* Cham: Springer: 15-19.

O'Leary, Z. 2005. Researching real-world problems: a guide to methods of inquiry. London: Sage.

Okhahlamba Local Municipality. 2013. Risk management framework: 2013–2014. http://www.okhahlamba.org.za/docs/risk/2013/ERM%20Framework%20-%202013%202014.pdf [Accessed 05/03/2016].

Okuboyejo, S.B. 2013. Environmental sustainability orientation of small and medium sized businesses in South Africa. Unpublished MBA dissertation, University of Pretoria, South Africa.

Olson, D.L. & Wu, D.D. 2008. Enterprise risk management. Singapore: World Scientific.

Olson, D.L. & Wu, D.D. 2010. A review of enterprise risk management in supply chain. *Kybernetes*, 39(5):694-706.

Organisation for Economic Co-operation and Development. 2009. National accounts at a glance 2009. http://www.oecd-ilibrary.org/sites/9789264067981en/01/03/index.html?itemId=/content/chapter/9789264075108-5-en [Accessed 07 September 2016].

Opdam, P. 2006. Ecosystem networks: a spatial concept for intergrative research and planning of landscapes. In: Tress, B., Tress, G., Fry, G. & Opdam, P. (eds). *From landscape research to landscape planning: aspects of integration, education and application.* Dordrecht: Springer: 51-66

Payne, V. 2009. Practical implementation of ERM and its challenges – Part 9. *Enterprise Risk*, 3(9): 28-29, October.

Payne, V. 2010. ERM framework. Sustainable embedding of ERM. *Enterprise Risk*, 3(11):23, December–January.

Perez, W. 2015. Business entities. https://www.thebalance.com/business-entities-3193420 [Accessed 22/04/2017].

Peters, R.M., Van Gensen, G., Isaacs, E.B.H., Botha, M.J. & Naicker, V. 2014. Education and small business growth: a gender perspective of two divergent provinces in South Africa. *International Business & Economics Research Journal*, 13(5):1127-1140, September/October.

Pettinger, T. 2011. GDP per capita statistics. http://www.economicshelp.org/blog/343/economics/gdp-per-capita-statistics/ [Accessed 09/02/2016].

Pezzuto, I. 2016. *Predictable and avoidable: repairing economic dislocation and preventing the recurrence of crisis.* Abingdon: Routledge.

Phillips, J.J. & Phillips, P.P. 2016. *Handbook of training evaluation and measurement methods*, 4<sup>th</sup> ed. Abingdon: Routledge.

Picciano, A.G. 2016. Models of survey research. In: Dziuban, C.D., Picciano, A.G., Graham, C.R. & Moskal, P.D. (eds). *Conducting research in online and blended learning environments: new pedagogical frontiers*. New York, NY: Routledge: 55-69.

Pilinkienė, V., Deltuvaitė, V., Daunorienė, A. & Gaidelys, V. 2016. *Competitiveness creation and maintenance in the postal services industry: a Lithuanian case study*. Cham: Springer.

Plan Projections. 2015. Economic factors. http://www.planprojections.com/glossary/economic-factors/ [Accessed 11/02/2016].

Plumridge, A. 2010. Costing the earth: the economics of sustainability in the curriculum. In: Jones, P., Selby, D. & Sterling, S.R. (eds). *Sustainability education: perspectives and practice across higher education*. Abingdon: Earthscan: 273-294.

Pojasek, R.B. 2013. Organizations and their contexts: where risk management meets sustainability performance. *Environmental Quality Management*, 22(3):81-93, Spring.

Pollard, D. & Stephen, D.W. 2008. Sustainability, risk and opportunity: a holistic approach. http://www.accountingforsustainability.org/wp-content/uploads/2011/10/CICA-A-Holistic-Approach-Report.pdf [Accessed 28/04/2016].

Posel, D., Casale, D. & Vermaak, C. 2013. The unemployed in South Africa: why are so many not counted? http://www.econ3x3.org/article/unemployed-south-africa-why-are-so-many-not-counted [Accessed 07/09/2016].

Prescott, J., Young, O., O'Neill, L., Yau, N.J.N. & Stevens, R. 2002. Motives for food choice: a comparison of consumers from Japan, Taiwan, Malaysia and New Zealand. *Food Quality and Preference*, 13(7-8):489-495, October.

PricewaterhouseCoopers. 2009. The governance of risk (Chapter 4 of the Code). In *King's Counsel: Understanding and unlocking the benefits of sound corporate governance: Corporate governance: Executive guide to King III.* Johannesburg: PwC: 29-34. http://www.pwc.co.za/en/assets/pdf/executive-guide-to-kingiii.pdf [Accessed 11/07/2016].

Pushparaj, A. 2015. Meaning and definition of economic environment. http://www.publishyourarticles.net/knowledge-hub/business-studies/economic-environment/945/ [Accessed 07/09/2016].

Ramakrishna, S.P. 2015. *Enterprise compliance risk management: an essential toolkit for banks and financial services*. Singapore: John Wiley.

Ramukumba, T. 2014. Overcoming SMEs challenges through critical success factors: a case of SMEs in the Western Cape province, South Africa. *Economic and Business Review*, 16(1):19-38.

Rao, A. 2007. Evaluation of enterprise risk management (ERM) in Dubai – an emerging economy. *Risk Management*, 9(3):167-187.

Rawson, B. 2013. Operational risk. http://searchcompliance.techtarget.com/definition/operational-risk [Accessed 12/03/2015].

Razali, A.R. & Tahir, I.M. 2011. Review of the literature on enterprise risk management. *Business Management Dynamics*, 1(5):8-16, November.

Refsdal, A., Solhaug, B. & Stølen, K. 2015. Cyber-risk management. Cham: Springer.

Reniers, G.L.L. & Van Erp, H.R.N. 2016. *Operational safety economics: a practical approach focused on the chemical and process industries*. Chichester: John Wiley.

Renn, O. 2012. *Risk governance: coping with uncertainty in a complex world*. Hoboken, NJ: Taylor & Francis.

Research & Education Association Editors & Fogiel, M. 2012. *Microeconomics super review: all you need to know*. Piscataway, NJ: Research & Education Association.

Risk Management Association. 2017. Operational risk management. http://www.rmahq.org/operational-risk/ [Accessed 30/03/2017].

*Risk Management Dictionary*. 2016. Definition of risk management. http://www.praxiom.com/iso-31000-terms.htm#2.9\_Establishing\_the\_context [Accessed 19/08/2016].

Rittenberg, L.E. 2006. Internal control: no small matter; help for small businesses has arrived in the form of COSO's latest guidance on internal control over financial reporting. *Internal Auditor*, October. https://www.highbeam.com/doc/1G1-153425918.html [Accessed 19/06/2017].

RobecoSAM. 2016. Corporate sustainability. http://www.sustainability-indices.com/sustainability-assessment/corporate-sustainability.jsp [Accessed 28/04/2016].

Rodríguez-Perez, J.R. 2012. *Quality risk management in the FDA-regulated industry*. Milwaukee, WI: ASQ.

Rossoni, L., Engelbert, R. & Bellegard, N.L. 2016. Normal science and its tools: reviewing the effects of exploratory factor analysis in management. *Revista de Administração (Sãlo Paulo)*, 51(2):198-211, April/June.

Rubino, M. & Vitolla, F. 2014. Corporate governance and the information system: how a framework for IT governance supports ERM. *Corporate Governance*, 14(3):320-338.

Ruel, E.E., Wagner III, W.E. & Gillespie, B.J. 2016. *The practice of survey research: theory and applications*. Thousand Oaks, CA: Sage.

SA Places. 2016. South Africa popular destinations. http://www.places.co.za/html/popular\_dest.html [Accessed 20/09/2016].

Sadgrove, K. 2016. The complete guide to business risk management, 3<sup>rd</sup> ed. Abingdon: Routledge.

Sales, F. 2014. Compliance risk. http://searchcompliance.techtarget.com/definition/compliance-risk [Accessed 17/04/2017].

Sanieni-Pour, A. 2015. Inequality in South Africa: a post-apartheid analysis. http://www.worldpolicy.org/blog/2015/07/07/inequality-south-africa-post-apartheid-analysis [Accessed 28/03/2017].

Saunders, M., Lewis, P. & Thornhill, A. 2007. *Research methods for business students*, 4<sup>th</sup> ed. Harlow: Pearson Education.

Saunders, M., Lewis, P. & Thornhill, A. 2012. *Research methods for business students*, 6<sup>th</sup> ed. Harlow: Pearson Education.

Schmidt, H.J., Mason, R.B., Bruwer, J.P & Aspeling, J. 2016. Exploratory study of finance accessibility for small retailers. Paper presented at the University of Zululand MBALI Conference, Richards Bay, KwaZulu-Natal, South Africa, 6–8 July 2016.

Servon, L.J., Visser, M.A. & Fairlie, R.W. 2010. The continuum of capital for small and micro enterprises. *Journal of Developmental Entrepreneurship*, 15(3):301-323, September.

Shahzad, B., Al-Ohali Y. & Abdullah, A. 2011. Trivial model for mitigation of risks in software development life cycle. *International Journal of the Physical Sciences*, 6(8):2072-2082.

Sharma, R.K. 1997. Sociological methods and techniques. Delhi: Atlantic.

Shaw, G.K. 2010. A risk management model for the tourism industry in South Africa. Unpublished PhD (Tourism) thesis, North-West University, Potchefstroom, South Africa.

Shenkir, W.G. & Walker, P.L. 2011. Enterprise Risk Management: frameworks, elements, and integration. Montvale, NJ: Institute of Management Accounts. http://www.stjohns.edu/sites/default/files/documents/academics/tobin/enterprise\_risk\_management\_fr ameworks.pdf [Accessed 12/09/2016].

Shortreed, J. 2010. Enterprise risk management and ISO 31000. *Journal of Policy Engagement,* 2(3): 8-10, June.

Simona-Iulia, C. 2014. Comparative study between traditional and enterprise risk management – a theoretical approach. *Annals of the University of Oradea, Economic Science* Series, 23(1):276-282. http://steconomiceuoradea.ro/anale/volume/2014/n1/029.pdf [Accessed 01/03/2016].

Sincero, S.M. 2012. Types of survey. https://explorable.com/types-of-survey [Accessed 21/01/16].

Singer, S. Amorós, J.E. & Moska, D. 2015. Global Entrepreneurship Monitor 2014 global report. London: London Business School. http://www.gemconsortium.org/report [Accessed 09/02/2016].

Singh, Y.K. & Bajpai, R.B. 2007. *Research methodology: techniques and trends*. New Delhi: APH Publishing.

Sippitt, M. 2014. SMEs are the lifeblood of economies and the Commonwealth can make a real difference.

http://www.clarkslegal.com/Legal\_Updates/Read/SMEs\_are\_the\_lifeblood\_of\_economies\_and\_the\_C ommonwealth\_can\_make\_a\_real\_difference\_UK\_lawyer\_tells\_international\_conference\_in\_Banglade sh. [Accessed 17/04/2017].

Siwangaza, L. Smit, Y., Bruwer, J.P. & Ukpere, W.I. 2014. The status of internal controls in fast moving small medium and micro consumer goods enterprises within the Cape Peninsula. *Mediterranean Journal of Social Sciences*, 5(10):163-175, June.

Small Business Project. 2013. Easier, harder for small business in South Africa. http://www.sbp.org.za/uploads/media/SME\_Growth\_Index\_-\_2012\_Headline\_Report.pdf [Accessed 01/04/2013].

Smit, A.v.A. & Fatoki, O.O. 2012. Debt financing to new small ventures in South Africa: the impact of collateral, ethics and the legal system. *African Journal of Business Management*, 6(3):1136-1146.

Smit, Y. 2012. A structured approach to risk management for South African SMEs. Unpublished DTech (Internal Auditing) thesis, Cape Peninsula University of Technology, Cape Town, South Africa.

Smit, Y. & Watkins, J.A. 2012. A literature review of small and medium enterprises (SME) risk management practices in South Africa. *African Journal of Business Management* 6(21):6324-6330.

Smith, K. 2011. List of 16 major leading & lagging economic indicators. http://www.moneycrashers.com/leading-lagging-economic-indicators/ [Accessed 17/08/2016].

Society for Risk Analysis. 2013. SRA 2013 President's message from George Gray. http://www.sra.org/news/sra-2013-presidents-message-george-gray#\_ftn1 [Accessed 19/02/2016].

Society of Actuaries. 2016. About ERM. http://www.ceranalyst.org/about-erm.asp [Accessed 24/08 2016].

Sokoto, A.A. & Abdullahi, Y.A. 2013. Strengthening small and medium enterprises (SMEs) as a strategy for poverty reduction in north western Nigeria. *American Journal of Humanities and Social Sciences*, 1(3):189-201.

South Africa. 1995. White Paper on national strategy for the development and promotion of small business in South Africa. Pretoria: DTI.

South Africa. 1996. National Small Business Act, No. 102 of 1996. Pretoria: Government Printer.

South Africa. 2014. Small Business Development launches National SMME Policy Colloquium. http://www.gov.za/sbdi-and-south-african-department-small-business-development-launchcountry%E2%80%99s-first-national-smme [Accessed 10/02/2016].

South Africa. 2016. South Africa GDP per capita. http://southafrica.opendataforafrica.org/oxlkluf/south-africa-gdp-per-capita. [Accessed 10/02/2017].

South African Government News Agency. 2016. Budget 2016: Tough balancing act in a tough global environment. http://www.southafrica.info/news/business/2016-budget-260216.htm#.VztzRNJ97IU [Accessed 17/05/2016].

South African Reserve Bank. 2015. Measuring poverty and inequality in South Africa. https://www.resbank.co.za/Lists/News%20and%20Publications/Attachments/7143/032015Measuring %20poverty%20and%20inequality%20in%20South%20Africa.pdf [Accessed 17/04/2017].

South African Reserve Bank. 2016. Financial stability review. https://www.resbank.co.za/Lists/News%20and%20Publications/Attachments/7278/FSR%20First%20e dition%202016.pdf [Accessed 04/01/2017].

South African Reserve Bank. 2017a. Unemployment. http://www2.resbank.co.za/internet/Glossary.nsf/0/9028ea9be02366d342256b43002f51f5?OpenDocu ment [Accessed 30/03/2017].

South African Reserve Bank. 2017b. Selected historical rates. https://www.resbank.co.za/Research/Rates/Pages/SelectedHistoricalExchangeAndInterestRates.aspx [Accessed 30/03/201]7.

South African Risk and Vulnerability Atlas. 2009. Purpose of the socio-economic theme page. http://rava.qsens.net/themes/socioeconomic\_template/purpose-of-the-theme-page [Accessed 05/10/2016].

Stainbank, L. 2008. The development of financial reporting for SMEs in South Africa: implications of recent and impending changes. *African Journal of Accounting, Economics, Finance and Banking Research*, 3(3):1-17.

Stamler, R.T., Marschdorf, H.J. & Possamai, M. 2014. *Fraud prevention and detection: warning signs and the red flag system*. Boca Raton, FL: CRC Press.

State of Queensland (Queensland Treasury). 2011. A guide to risk management. https://www.treasury.qld.gov.au/publications-resources/risk-management-guide/guide-to-risk-management.pdf [Accessed 19/02/2016]. Statistics South Africa. 2014. Quarterly labour force survey: Quarter 4, 2013. http://www.statssa.gov.za/publications/P0211/P02114thQuarter2013.pdf [Accessed 17/04/2017].

Statistics South Africa. 2015. Quarterly labour force survey: Quarter 4, 2014. http://www.statssa.gov.za/publications/P0211/P02114thQuarter2014.pdf [Accessed 17/04/2017].

Statistics South Africa. 2016. Quarterly labour force survey: Quarter 4, 2015. http://www.statssa.gov.za/publications/P0211/P02114thQuarter2015.pdf [Accessed 17/04/2017].

Statistics South Africa. 2017a. Gross domestic product: Fourth quarter 2016. http://www.statssa.gov.za/publications/P0441/P04414thQuarter2016.pdf [Accessed 30/03/2017].

Statistics South Africa. 2017b. Quarterly labour force survey: Quarter 4, 2016. http://www.statssa.gov.za/publications/P0211/P02114thQuarter2016.pdf [Accessed 17/04/2017].

Steinberg, R.M., Everson, M.E.A., Martens, F.J. & Nottingham, L.E. 2004. Enterprise risk management – Integrated framework. http://www.coso.org/documents/coso\_erm\_executivesummary.pdf [Accessed 05/03/2016].

Stofile, U., Mushatu, A., Nair, N., Daniels, R., Goosen, K., Khalil, Y. & Essa, N. 2011. Independent food retailers in the Republic of South Africa. Can they ensure sustainability in an evolving retail landscape? http://www.wrseta.org.za/downloads/ILDP/Imitha%20Final.pdf [Accessed 20/10/2016].

Sullivan, G.M. 2011. A primer on the validity of assessment instruments. *Journal of Graduate Medical Education*, 3(2):119-120, June.

Sunjka, B.P. & Emwanu, B. 2015. Risk management in manufacturing SMEs in South Africa. *Proceedings of the International Association for Management of Technology (IAMOT) 2015 Conference, Cape Town, 8–11 June 2015.* Johannesburg: University of the Witwatersrand: 1469-1486.

Sunjka, B.P. & Sklar-Chik, M.S. 2012. Supply chain risk and small and medium manufacturing enterprises in South Africa. In Schutte, C.S.L. (ed.). *Proceedings of the 42<sup>nd</sup> International Conference on Computers & Industrial Engineering 2012, Cape Town, 15–18 July 2012.* Red Hook, NY: Curran Associates, Inc: 741-754.

Sustainable Environment. 2016. Sustainable development. http://www.sustainable-environment.org.uk/Society/society.php [Accessed 06/10/2016].

Tabussum, R. & Mahmood, M.M. 2015. Importance of service quality and service encounter: as a foundation for customer satisfaction in fast food industry. *International Journal of Business & Management*, 3(4):61-65, April.

Tang, W., Cui, Y. & Babenko, O. 2014. Internal consistency: do we really know what it is and how to assess it? *Journal of Psychology and Behavioral Science*, 2(2):205-220, June.

Taroun, A., Yang, J.B. & Lowe, D. 2011. Construction risk modelling and assessment: insights from a literature review. *The Built & Human Environment Review*, 4(Special Issue 1):87-97.

Taylor, J.C. 2012. Concepts of strategic risks and managing strategic risks. http://vsu.edu/files/docs/internal-audit/risk-concepts.pdf [Accessed 19/02/2016].

Tella, A. & Tella, A.O. 2013. The promise of open source systems/software in developing requisite egovernment solutions for developing countries: a review of literature. In Bwalya, K.J. (ed.). *Technology development and platform enhancements for successful global e-government design*. Hershey, PA: IGI Global: 191-192.

Tonello, M. 2012. Strategic risk management: a primer for directors. https://corpgov.law.harvard.edu/2012/08/23/strategic-risk-management-a-primer-for-directors/ [Accessed 19/02/2016]. Trading Economics. 2016a. South Africa unemployed persons. http://www.tradingeconomics.com/south-africa/unemployed-persons [Accessed 18/08/2016].

Trading Economics. 2016b. South Africa lending rate. http://www.tradingeconomics.com/south-africa/lending-rate [Accessed 17/02/2016].

Trading Economics. 2016c. South Africa unemployment rate. http://www.tradingeconomics.com/south-africa/unemployment-rate [Accessed 08/02/16].

Trochim, W.M.K. 2006. Levels of measurement. http://www.socialresearchmethods.net/kb/measlevl.php [Accessed 19 September 2016].

Tsiouras, I. 2015. *The risk management according to the standard ISO 31000.* Youcanprint. Self-Publishing.

Tulashie, S.K., Addai, E.K. & Annan, J.S. 2016. Exposure assessment, a preventive process in managing workplace safety and health, challenges in Ghana. *Safety Science*, 84:210-215.

Tustin, D. 2015. The physiognomy of SMMEs in South Africa and consequential national strategy reinforcement. *Retail and Marketing Review*, 11(1):78-91.

UK Data Archive. 2017. Create & manage data: consent & ethics. http://www.data-archive.ac.uk/create-manage/consent-ethics [Accessed 17/04/2017].

University of Stellenbosch Business School. 2014. Khayelitsha small business project. Need and scope for education and training in Khayelitsha's small-business sphere. http://www.usb.ac.za/Common/pdfs/sba\_facts/FinancingBusinessesin\_6.pdf [Accessed 16/05/2017].

USForex. 2016. Yearly average rates. http://www.usforex.com/forex-tools/historical-rate-tools/yearly-average-rates [Accessed 16/05/2016].

Vadiveloo, J.J. & Parsa, H.G. 2012. Enterprise risk management – small businesses. Paper presented at the Society of Actuaries 2012 Annual Meeting & Exhibit, Washington, DC, 14–17 October 2012.

Van Weele, G. & Maree, G. 2013. State of environment outlook report for the Western Cape Province. Introductory matter.

http://soer.deat.gov.za/dm\_documents/Western\_Cape\_2013\_State\_of\_Environment\_Report\_Lhiif.pd f [Accessed 21/09/2016].

Verbano, C. & Venturini, K. 2013. Managing risks in SMEs: a literature review and research agenda. *Journal of Technology Management & Innovation*, 8(3):186-197.

Verma A. 2012. Macro factors affecting business environment. http://www.slideshare.net/brucemasama/savedfiles?s\_title=macro-factors-affecting-businessenvironment&user\_login=aayush30 [Accessed 09/11/14].

Vermeulen, C. 2015. Inflation, growth and employment in South Africa: trends and trade-offs. Economic Research Southern Africa (ERSA) Working Paper 547. Cape Town: ERSA.

Vijfvinkel, S., Bouman, N. & Hessels, J. 2011. Environmental sustainability and financial performance of SMEs. Zoetermeer, Netherlands: EIM.

Von Ketelhodt, A. & Wöcke, A. 2008. The impact of electricity crises on the consumption behaviour of small and medium enterprises. *Journal of Energy in Southern Africa*,19(1):4-12, February.

Von Rössing, R. 2007. *IT control objectives for Basel II: the importance of governance and risk management for compliance*. Rolling Meadows, IL: IT Governance Institute.

Walker, P.L. & Shenkir, W.G. 2008. Implementing enterprise risk management. *Journal of Accountancy*, 205(3):31-33.

Watson, J. 2010. SME performance: separating myth from reality. Cheltenham: Edward Elgar.

Wealth Management. 2014. List of major leading & lagging economic indicators. http://wtwealthmanagement.com/documents/pdf/WTWealth\_2014-07\_MajorLeadingList.pdf [Accessed 16/05/2016].

Weller, N. 2015. COSO Enterprise Risk Management Framework. http://www.accaglobal.com/uk/en/student/exam-support-resources/professional-exams-studyresources/p1/technical-articles/coso-enterprise-risk-management-framework-part-1.html [Accessed 05/03/2016].

Western Cape Government Provincial Treasury. 2015. Provincial economic review and outlook 2015. https://www.westerncape.gov.za/assets/departments/treasury/Documents/2015\_pero\_final\_to\_web\_1 5\_otober\_2015.pdf [Accessed 21/09/2016].

Whittington, O.R. 2015. *Wiley CPA excel exam review 2016 study guide January: business environment and concepts.* Hoboken, NJ: John Wiley.

Wiese, J.S. 2014. Factors determining the sustainability of selected small and medium-sized enterprises. Unpublished MBA dissertation, North-West University, Potchefstroom, South Africa.

Willard, B. 2012. *The new sustainability advantage: seven business case benefits of a triple bottom line.* Gabriola Island, BC: New Society.

Winks, S. 2008. Aiding SMMEs in SA: small business. *Enterprise Risk*, 2(4):20-21, May.

Wolf, R. 2008. The evolution of enterprise risk management. *The Actuary*, 5(3):19-22, June/July.

World Bank. 2013. GINI index. http://data.worldbank.org/indicator/SI.POV.GINI [Accessed 16/03/2015].

World Bank. 2016. Worldwide governance indicators. http://info.worldbank.org/governance/wgi/index.aspx#countryReports [Accessed 17/02/2016].

Worldometers. 2016. South Africa population. http://www.worldometers.info/world-population/south-africa-population/ [Accessed 17/08/2016].

Worldometers. 2017. South Africa population. http://www.worldometers.info/world-population/south-africa-population/ [Accessed 30/03/2017].

Wurzler, J. 2013. Information risks & risk management. http://www.sans.org/reading-room/whitepapers/policyissues/information-risks-risk-management-34210 [Accessed 19/02/2016].

Yusuf, T.O. & Dansu, F.S. 2013. SMES, business risks and sustainability in Nigeria. *European Journal of Business and Social Sciences*, 2(9):76-94, December.

Zhao, X., Hwang, B.G. & Low, S.P. 2015. *Enterprise risk management in international construction operations.* Singapore: Springer.

Zikmund, W.G., Babin, B.J., Carr, J.C. & Griffin, M. 2013. *Business research methods*, 9<sup>th</sup> ed. Mason, OH: Cengage Learning.

Zink, K.J., Steimle, U. & Fischer, K. 2008. *Corporate sustainability as a challenge for comprehensive management*. Dordrecht: Springer.

Zulu, L. 2014a. Importance of SMMEs is big business for country. http://www.iol.co.za/business-report/opinion/importance-of-smmes-is-big-business-for-country-1768670 [Accessed 25/01/16].

Zulu, L. 2014b. SMMEs hold key to economic growth. http://www.gov.za/blog/smmes-hold-key-economic-growth [Accessed 25/01/2015].

Zulu, L. 2015. SMMEs hold key to economic growth. http://www.sanews.gov.za/south-africa/smmes-hold-key-economic-growth [Accessed 07/09/2016].

#### ANNEXURES

#### ANNEXURE A: RESEARCH QUESTIONNAIRE

RESEARCHER DETAILS				
Name:	Bruce Taona			
Surname:	Masama			
Student number:	208179380			
E-mail:	masamab@cput.ac.za			
Contact number:	084 874 7037			

SUPERVISOR DETAILS					
Name:	Juan-Pierré (JP)				
Surname:	Bruwer				
E-mail:	BruwerJP@cput.ac.za				

#### **RESEARCH TITLE**

The utilisation of Enterprise Risk Management in fast-food Small, Medium and Micro Enterprises (SMMEs) operating in the Cape Peninsula

#### CONFIDENTIALITY AND ANONYMITY

Please note that ALL information provided by the respondent will be kept strictly confidential and that the anonymity of the respondent is guaranteed. The information provided will be used strictly for research purposes only. Respondents also may withdraw from this study at any point in time as participation is voluntary.

#### HOW TO COMPLETE THIS SURVEY

This survey comprises mostly closed-ended questions which require the respondent to fill in a numerical digit and/or mark an 'x' in the most appropriate boxes. Clear instructions for each question are given under each section. If respondents do not understand a specific question, please feel free to contact either the researcher and/or supervisor indicated on this front page. By completing and submitting this survey questionnaire, you are providing informed consent to participate in the research.

#### PRIMARY OBJECTIVES OF THE SURVEY

The Faculty of Business and Management Sciences of the Cape Peninsula University of Technology's research niche area reads: *The effective management of SMMEs*.

Research has shown that approximately 80% of all small businesses which start up, fail within the first five years of existence. Essentially this research is based on the perception that small businesses do not make adequate use of risk management systems, particularly Enterprise Risk Management (ERM). The objectives of this survey include:

- > To determine the factors that affect small and micro business sustainability.
- > To determine the risks faced by small and micro enterprises.
- > To gain insight into how small and micro enterprises deal with risks they face.
- > To determine the extent to which small and micro businesses have implemented Enterprise Risk Management (ERM) or its elements.

SECTION A: BUSINESS IDENTIFICATION
1) In what <b>industry</b> do you operate?
2) How long has your business been in existence? (years)
years
3) How many <b>employees</b> do you employ? (number)
employees
4) Are you part of a <b>franchise</b> ?
Yes No 🗆
5) <b>You</b> are the (tick the best answer)
Owner 🗆 Manager 🗆
Owner and manager  Other
5.1) If other, please specify

SECTION B: FACTORS AFFECTING FAST FOOD SMMEs								
6) Statement: The following barriers affect my business's sustainability: (Mark an 'X' in the appropriate box below. 1 = strongly disagree, 2 = disagree, 3 = agree, 4 = strongly agree)								
	1	2	3	4				
6.1) Food prices of suppliers								
6.2) Unemployment rates								
6.3) Inflation rates								
6.4) Size of the business								
6.5) Difficult access to finance (capital)								
6.6) Stiff competition								
6.7) Low demand for products and/or services								
6.8) Location of the business								
6.9) Ineffective marketing								
6.10) 'Substitute' products and/or services at								
cheaper price								
6.11) Unproductive employees								
6.12) Poor service quality								
6.13) Poor product quality								

6.14) Cost of electricity		
6.15) Cost of water		
6.16) Lack of adequate credit facilities		
6.17) Unreliable supply of raw materials		
6.18) Interest rates		
6.19) Legislation, rules and regulations		
6.20) Labour costs		
6.21) Crime rate		
6.22) Bad debts (customers not pay debts)		
6.23) Taxation		
6.24) Rapid change in technology		

SECTION C: RISKS FACED BY FAST-FOOD SMMEs							
7) Statement: My business is faced with the following <i>problems</i> :							
(Mark an 'X' in the appropriate box below. 1 = strongly disagree, 2 =							
disagree, 3 = agree, 4 = strongly agree)							
STRATEGIC RISK	1	2	3	4			
7.1) Strong competition							
7.2) Low demand for products							
7.3) Delays in supply chain process							
7.4) Weak business growth							
7.5) Unstable business conditions							
COMPLIANCE RISK	1	2	3	4			
7.6) Health and safety risks							
7.7) Employee risks (immigration, etc.)							
7.8) Environmental risks (waste, etc.)							
FINANCIAL RISK	1	2	3	4			
7.9) Weak business profitability (profits)							
7.10) Weak business liquidity (cash)							
7.11) Weak business solvency (more liabilities)							
7.12) Weak business efficiency (poor efficiency)							
OPERATIONAL RISK	1	2	3	4			
7.13) Poor service quality							
7.14) Employee theft							
7.15) Large expenses (in relation to income)							
7.16) Increase in cost prices of products							
7.17) Poor product quality							

SECTION D: RISK MANAGEMENT BY FAST FOOD SMMEs							
8) Statement: My business deals with the aforementioned problems							
by means of: (Mark an 'X' in the appropriate box below.	1 = st	trong	gly				
disagree, 2 = disagree, 3 = agree, 4 = strongly agree)							
Risk management methods	1	2	3	4			
8.1) A code of ethics (principles guiding employees)							
8.2) Risk policies							
8.3) Setting the right 'tone at the top'							
8.4) Setting of risk appetite(s)							
8.5) An adequate company strategy							
8.6) Setting clear objectives							
8.7) Internal interviews and discussion (e.g SWOT)							
8.8) External sources (e.g. risk consultants)							
8.9) Tools diagnostics and processes (e.g. checklists)							
8.10) Risk analyses							
8.11) Risk evaluation(s)							
8.12) Risk treatment(s) (reducing actual risks)							
8.13) Implementing controls to reduce the effect of risks							
8.14) Using budgets							
8.15) Using policies and procedures							
8.16) Timely communication of risks							
8.17) Consulting risk management experts							
8.18) Conducting performance appraisals regularly							
8.19) Performing separate evaluations							

SECTION E: ENTERPRISE RISK MANAGEMENT (ERM) Mark an 'X' in the appropriate box below. 1= None 2 = Little, 3 = Very little, 4 = Average, 5 = Some, 6 = A lot								
	1	2	3	4	5	6		
9.1) How well do you understand the ERM process?								
9.2) How many of the ERM initiatives have you								
implemented in your business?								
9.3) How much value does ERM add to your business?								

THANK YOU									
Thank you for your time and effort in completing this survey for the benefit of									
academic research in the	field of internal auditing.								
Business Name:									
Address									
Telephone									

Would you like <b>e-mail feedback</b> of this study?							
Yes		No					
If yes, please supply e-mail addr							

### ANNEXURE B: DEMOGRAPHICS OF SAMPLE

# **Business demographics**

#### In what industry do you operate?

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Fast-food industry	1.00	116	100.00	100.00	100.00
Total		116	100.00	100.00	

### How long has your business been in existence? (years)

Value Label				Valid Percent	Cumulative
	Value	rioquonoy			Percent
	2.00	17	14.66		
	3.00				
	4.00				
	5.00				
	6.00	6	5.17	5.17	57.76
	7.00	7	6.03	6.03	63.79
	8.00		4.31	4.31	68.10
	9.00		1.72	1.72	69.83
	10.00	12	10.34	10.34	80.17
	11.00	4	3.45	3.45	83.62
	12.00		2.59	2.59	86.21
	13.00	2 3	1.72	1.72	87.93
	14.00	3	2.59	2.59	90.52
	15.00			4.31	94.83
	16.00	1	.86	0.86	95.69
	18.00	2	1.72	1.72	97.41
	20.00	1	0.86	0.86	98.28
	30.00	1	0.86	0.86	99.14
	40.00	1	0.86	0.86	100.00
Total		116	100.00	100.00	

Value Label				Valid Percent	Cumulative
					Percent
	1.00			4.31	4.31
	2.00	15	12.93	12.93	17.24
	3.00			16.38	33.62
	4.00	12			
	5.00	9	7.76	7.76	51.72
	6.00	4	3.45	3.45	55.17
	7.00		5.17	5.17	60.34
	8.00	3	2.59	2.59	62.93
	9.00	4	3.45	3.45	66.38
	10.00		6.03	6.03	72.41
	11.00	3	2.59	2.59	75.00
	12.00	4	3.45	3.45	78.45
	13.00	1	0.86	0.86	79.31
	15.00	6	5.17	5.17	84.48
	16.00	1	0.86	0.86	85.34
	18.00	3	2.59	2.59	87.93
	20.00		1.72	1.72	89.66
	25.00	1	0.86	0.86	90.52
	27.00	1	0.86	0.86	91.38
	30.00	2	1.72	1.72	93.10
	31.00	1	0.86	0.86	93.97
	35.00	1	0.86	0.86	94.83
	40.00	2	1.72	1.72	96.55
	50.00	4	3.45	3.45	100.00
Total	-	116	100.00	100.00	

# How many employees do you employ? (number)

# Are you part of a franchise?

, ao you puit of a nanomoof									
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative				
					Percent				
No	2.00	116	100.00	100.00	100.00				
Total		116	100.00	100.00					

# You are the . . . (tick the best answer)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative			
					Percent			
Owner	1.00	31	26.72	26.72	26.72			
Manager	2.00	78	67.24	67.24	93.97			
Owner-manager	3.00	7	6.03	6.03	100.00			
Total	•	116	100.00	100.00				

## ANNEXURE C: FREQUENCIES OF ORDINAL DATA

#### Economic factors affecting the sustainability of sampled SMMEs

The following barrier affects my business's sustainability: Food prices of suppliers								
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative			
		-			Percent			
Strongly disagree	1.00	8	6.90	6.90	6.90			
Disagree	2.00	16	13.79	13.79	20.69			
Agree	3.00	45	38.79	38.79	59.48			
Strongly agree	4.00	47	40.52	40.52	100.00			
Total		116	100.00	100.00				

The following barrier affects my business's sustainability: Food prices of suppliers

#### The following barrier affects my business's sustainability: Unemployment rates

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	7	6.03	6.03	6.03
Disagree	2.00	36	31.03	31.03	37.07
Agree	3.00	40	34.48	34.48	71.55
Strongly agree	4.00	33	28.45	28.45	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Inflation rates

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	6	5.17	5.17	5.17
Disagree	2.00	13	11.21	11.21	16.38
Agree	3.00	57	49.14	49.14	65.52
Strongly agree	4.00	40	34.48	34.48	100.00
Total	•	116	100.00	100.00	

The following barrier affects my business's sustainability: Size of the business

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	17	14.66	14.66	14.66
Disagree	2.00	32	27.59	27.59	42.24
Agree	3.00	39	33.62	33.62	75.86
Strongly agree	4.00	28	24.14	24.14	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Difficult access to finance (capital)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		_			Percent
Strongly disagree	1.00	14	12.07	12.07	12.07
Disagree	2.00	20	17.24	17.24	29.31
Agree	3.00	33	28.45	28.45	57.76
Strongly agree	4.00	49	42.24	42.24	100.00
Total	•	116	100.00	100.00	

The following barrier affects my business's sustainability: Stiff competition

The following burner		, ouoluinas	mity. Othi oompo		
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	8	6.90	6.90	6.90
Disagree	2.00	19	16.38	16.38	23.28
Agree	3.00	37	31.90	31.90	55.17
Strongly agree	4.00	52	44.83	44.83	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Low demand for products and/or services

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	20	17.24	17.24	17.24
Disagree	2.00	44	37.93	37.93	55.17
Agree	3.00	40	34.48	34.48	89.66
Strongly agree	4.00	12	10.34	10.34	100.00
Total	•	116	100.00	100.00	

The following barrier affects my business's sustainability: Location of the business

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	33	28.45	28.45	28.45
Disagree	2.00	40	34.48	34.48	62.93
Agree	3.00	25	21.55	21.55	84.48
Strongly agree	4.00	18	15.52	15.52	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Ineffective marketing

	V / I I				
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	28	24.14	24.14	24.14
Disagree	2.00	40	34.48	34.48	58.62
Agree	3.00	31	26.72	26.72	85.34
Strongly agree	4.00	17	14.66	14.66	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: 'Substitute' products and/or services at cheaper prices

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	10	8.62	8.62	8.62
Disagree	2.00	40	34.48	34.48	43.10
Agree	3.00	52	44.83	44.83	87.93
Strongly agree	4.00	14	12.07	12.07	100.00
Total	•	116	100.00	100.00	

The following barrier affects my business's sustainability: Unproductive employees

V					
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	44	37.93	37.93	37.93
Disagree	2.00	23	19.83	19.83	57.76
Agree	3.00	30	25.86	25.86	83.62
Strongly agree	4.00	19	16.38	16.38	100.00
Total		116	100.00	100.00	

### The following barrier affects my business's sustainability: Poor service quality

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	41	35.34	35.34	35.34
Disagree	2.00	30	25.86	25.86	61.21
Agree	3.00	30	25.86	25.86	87.07
Strongly agree	4.00	15	12.93	12.93	100.00
Total		116	100.00	100.00	

#### The following barrier affects my business's sustainability: Poor product quality

The felletting barrier andete m			y buomboo o buotamability. I bor product quality			
Value	Frequency	Percent	Valid Percent	Cumulative		
				Percent		
1.00	42	36.21	36.21	36.21		
2.00	37	31.90	31.90	68.10		
3.00	20	17.24	17.24	85.34		
4.00	17	14.66	14.66	100.00		
	116	100.00	100.00			
	Value 1.00 2.00 3.00	Value Frequency 1.00 42 2.00 37 3.00 20 4.00 17	Value         Frequency         Percent           1.00         42         36.21           2.00         37         31.90           3.00         20         17.24           4.00         17         14.66	Value         Frequency         Percent         Valid Percent           1.00         42         36.21         36.21           2.00         37         31.90         31.90           3.00         20         17.24         17.24           4.00         17         14.66         14.66		

### The following barrier affects my business's sustainability: Cost of electricity

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	15	12.93	12.93	17.24
Agree	3.00	33	28.45	28.45	45.69
Strongly agree	4.00	63	54.31	54.31	100.00
Total		116	100.00	100.00	

#### The following barrier affects my business's sustainability: Cost of water

ine leneting same aneote m		,			
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	21	18.10	18.10	18.10
Disagree	2.00	22	18.97	18.97	37.07
Agree	3.00	33	28.45	28.45	65.52
Strongly agree	4.00	40	34.48	34.48	100.00
Total		116	100.00	100.00	

### The following barrier affects my business's sustainability: Lack of adequate credit facilities

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		_			Percent
Strongly disagree	1.000	14	12.07	12.07	12.07
Disagree	2.000	45	38.79	38.79	50.86
Agree	3.000	41	35.34	35.34	86.21
Strongly agree	4.000	16	13.79	13.79	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Unreliable supply of raw materials

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	26	22.41	22.41	22.41
Disagree	2.00	42	36.21	36.21	58.62
Agree	3.00	37	31.90	31.90	90.52
Strongly agree	4.00	11	9.48	9.48	100.00
Total		116	100.00	100.00	

### The following barrier affects my business's sustainability: Interest rates

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	13	11.21	11.21	11.21
Disagree	2.00	27	23.28	23.28	34.48
Agree	3.00	46	39.66	39.66	74.14
Strongly agree	4.00	30	25.86	25.86	100.00
Total		116	100.00	100.00	

#### The following barrier affects my business's sustainability: Legislation, rules and regulations

					<u> </u>
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	15	12.93	12.93	12.93
Disagree	2.00	33	28.45	28.45	41.38
Agree	3.00	42	36.21	36.21	77.59
Strongly agree	4.00	26	22.41	22.41	100.00
Total	·	116	100.00	100.00	

### The following barrier affects my business's sustainability: Labour costs

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	14	12.07	12.07	12.07
Disagree	2.00	24	20.69	20.69	32.76
Agree	3.00	47	40.52	40.52	73.28
Strongly agree	4.00	31	26.72	26.72	100.00
Total		116	100.00	100.00	

#### The following barrier affects my business's sustainability: Crime rate

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	18	15.52	15.52	15.52
Disagree	2.00	24	20.69	20.69	36.21
Agree	3.00	37	31.90	31.90	68.10
Strongly agree	4.00	37	31.90	31.90	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Bad debts (Customers do not pay debts)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	51	43.97	43.97	43.97
Disagree	2.00	21	18.10	18.10	62.07
Agree	3.00	28	24.14	24.14	86.21
Strongly agree	4.00	16	13.79	13.79	100.00
Total		116	100.00	100.00	

The following barrier affects my business's sustainability: Taxation

The following barrier and to my baometer o baotamability. Taxation					
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	37	31.90	31.90	31.90
Disagree	2.00	24	20.69	20.69	52.59
Agree	3.00	38	32.76	32.76	85.34
Strongly agree	4.00	17	14.66	14.66	100.00
Total		116	100.00	100.00	

## The following barrier affects my business's sustainability: Rapid change in technology

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	29	25.00	25.00	25.00
Disagree	2.00	43	37.07	37.07	62.07
Agree	3.00	29	25.00	25.00	87.07
Strongly agree	4.00	15	12.93	12.93	100.00
Total		116	100.00	100.00	

#### **Risks influencing the sustainability of sampled SMMEs**

My business	s is faced with	the following	'problem':	Strong competition
		· · · · J		

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	6	5.17	5.17	5.17
Disagree	2.00	12	10.34	10.34	15.52
Agree	3.00	32	27.59	27.59	43.10
Strongly agree	4.00	66	56.90	56.90	100.00
Total		116	100.00	100.00	

#### My business is faced with the following 'problem': Low demand for products

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	21	18.10	18.10	18.10
Disagree	2.00	34	29.31	29.31	47.41
Agree	3.00	48	41.38	41.38	88.79
Strongly agree	4.00	13	11.21	11.21	100.00
Total		116	100.00	100.00	

My business is faced with the following 'problem': Delays in supply-chain process

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	21	18.10	18.10	18.10
Disagree	2.00	54	46.55	46.55	64.66
Agree	3.00	29	25.00	25.00	89.66
Strongly agree	4.00	12	10.34	10.34	100.00
Total	•	116	100.00	100.00	

My business is faced with the following 'problem': Weak business growth

INTY DUDITIOUS IS TUDE	ny baomede le labea war are renowing problem . Weak baomede growar						
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative		
					Percent		
Strongly disagree	1.00	12	10.34	10.34	10.34		
Disagree	2.00	43	37.07	37.07	47.41		
Agree	3.00	45	38.79	38.79	86.21		
Strongly agree	4.00	16	13.79	13.79	100.00		
Total		116	100.00	100.00			

### My business is faced with the following 'problem': Unstable business conditions

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	9	7.76	7.76	7.76
Disagree	2.00	43	37.07	37.07	44.83
Agree	3.00	51	43.97	43.97	88.79
Strongly agree	4.00	13	11.21	11.21	100.00
Total		116	100.00	100.00	

#### My business is faced with the following 'problem': Health and safety risks

	ny saoinese le labea mar ale reneming presionir ribalar and salety here					
Value Label	Value	Frequency	Percent	Valid Percent	Cumulative	
					Percent	
Strongly disagree	1.00	27	23.28	23.28	23.28	
Disagree	2.00	31	26.72	26.72	50.00	
Agree	3.00	45	38.79	38.79	88.79	
Strongly agree	4.00	13	11.21	11.21	100.00	
Total	•	116	100.00	100.00		

### My business is faced with the following 'problem': Employee risks (immigration, etc.)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	27	23.28	23.28	23.28
Disagree	2.00	40	34.48	34.48	57.76
Agree	3.00	33	28.45	28.45	86.21
Strongly agree	4.00	16	13.79	13.79	100.00
Total		116	100.00	100.00	

#### My business is faced with the following 'problem': Environmental risks (waste, etc.)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	21	18.10	18.10	18.10
Disagree	2.00	34	29.31	29.31	47.41
Agree	3.00	39	33.62	33.62	81.03
Strongly agree	4.00	22	18.97	18.97	100.00
Total		116	100.00	100.00	

My business is faced with the following 'problem': Weak business profitability (profits)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	13	11.21	11.21	11.21
Disagree	2.00	39	33.62	33.62	44.83
Agree	3.00	49	42.24	42.24	87.07
Strongly agree	4.00	15	12.93	12.93	100.00
Total		116	100.00	100.00	

My business is faced with the following 'problem': Weak business liquidity (cash)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative		
		-			Percent		
Strongly disagree	1.00	13	11.21	11.21	11.21		
Disagree	2.00	56	48.28	48.28	59.48		
Agree	3.00	35	30.17	30.17	89.66		
Strongly agree	4.00	12	10.34	10.34	100.00		
Total		116	100.00	100.00			

My business is faced with the following 'problem': Weak business solvency (more liabilities)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	22	18.97	18.97	18.97
Disagree	2.00	52	44.83	44.83	63.79
Agree	3.00	34	29.31	29.31	93.10
Strongly agree	4.00	8	6.90	6.90	100.00
Total	·	116	100.00	100.00	

### My business is faced with the following 'problem': Weak business efficiency (poor efficiency)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	13	11.21	11.21	11.21
Disagree	2.00	52	44.83	44.83	56.03
Agree	3.00	41	35.34	35.34	91.38
Strongly agree	4.00	10	8.62	8.62	100.00
Total		116	100.00	100.00	

## My business is faced with the following 'problem': Poor service quality

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	46	39.66	39.66	39.66
Disagree	2.00	37	31.90	31.90	71.55
Agree	3.00	22	18.97	18.97	90.52
Strongly agree	4.00	11	9.48	9.48	100.00
Total		116	100.00	100.00	

# My business is faced with the following 'problem': Employee theft

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	32	27.59	27.59	27.59
Disagree	2.00	14	12.07	12.07	39.66
Agree	3.00	43	37.07	37.07	76.72
Strongly agree	4.00	27	23.28	23.28	100.00
Total	•	116	100.00	100.00	

My business is faced with the following 'problem': Large expenses (in relation to income)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	22	18.97	18.97	18.97
Disagree	2.00	27	23.28	23.28	42.24
Agree	3.00	44	37.93	37.93	80.17
Strongly agree	4.00	23	19.83	19.83	100.00
Total		116	100.00	100.00	

### My business is faced with the following 'problem': Increase in cost prices of products

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	23	19.83	19.83	24.14
Agree	3.00	59	50.86	50.86	75.00
Strongly agree	4.00	29	25.00	25.00	100.00
Total		116	100.00	100.00	

#### My business is faced with the following 'problem': Poor product quality

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		. ,			Percent
Strongly disagree	1.00	38	32.76	32.76	32.76
Disagree	2.00	29	25.00	25.00	57.76
Agree	3.00	38	32.76	32.76	90.52
Strongly agree	4.00	11	9.48	9.48	100.00
Total	•	116	100.00	100.00	

#### **Risk-management initiatives used by sampled SMMEs**

My business deals with the aforementioned problems by means of: A code of ethics (principles guiding employees)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	6	5.17	5.17	5.17
Disagree	2.00	14	12.07	12.07	17.24
Agree	3.00	51	43.97	43.97	61.21
Strongly agree	4.00	45	38.79	38.79	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Risk policies

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	6	5.17	5.17	5.17
Disagree	2.00	31	26.72	26.72	31.90
Agree	3.00	54	46.55	46.55	78.45
Strongly agree	4.00	25	21.55	21.55	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Setting the right 'tone at the top'

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	9	7.76	7.76	7.76
Disagree	2.00	17	14.66	14.66	22.41
Agree	3.00	50	43.10	43.10	65.52
Strongly agree	4.00	40	34.48	34.48	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Setting of risk appetite(s)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	7	6.03	6.03	6.03
Disagree	2.00	26	22.41	22.41	28.45
Agree	3.00	57	49.14	49.14	77.59
Strongly agree	4.00	26	22.41	22.41	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: An adequate company strategy

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	15	12.93	12.93	17.24
Agree	3.00	61	52.59	52.59	69.83
Strongly agree	4.00	35	30.17	30.17	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Setting clear objectives

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		_			Percent
Strongly disagree	1.00	3	2.59	2.59	2.59
Disagree	2.00	10	8.62	8.62	11.21
Agree	3.00	40	34.48	34.48	45.69
Strongly agree	4.00	63	54.31	54.31	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Internal interviews and discussion (e.g SWOT)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	7	6.03	6.03	6.03
Disagree	2.00	20	17.24	17.24	23.28
Agree	3.00	42	36.21	36.21	59.48
Strongly agree	4.00	47	40.52	40.52	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: External sources (e.g. risk consultants)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	36	31.03	31.03	31.03
Disagree	2.00	36	31.03	31.03	62.07
Agree	3.00	31	26.72	26.72	88.79
Strongly agree	4.00	13	11.21	11.21	100.00
Total	-	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Tools diagnostics and processes (e.g. checklists)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative		
		-			Percent		
Strongly disagree	1.00	9	7.76	7.76	7.76		
Disagree	2.00	18	15.52	15.52	23.28		
Agree	3.00	50	43.10	43.10	66.38		
Strongly agree	4.00	39	33.62	33.62	100.00		
Total		116	100.00	100.00			

My business deals with the aforementioned problems by means of: Risk analyses

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	4	3.45	3.45	3.45
Disagree	2.00	29	25.00	25.00	28.45
Agree	3.00	51	43.97	43.97	72.41
Strongly agree	4.00	32	27.59	27.59	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Risk evaluation(s)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	26	22.41	22.41	26.72
Agree	3.00	50	43.10	43.10	69.83
Strongly agree	4.00	35	30.17	30.17	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Risk treatment(s) (reducing actual risks)

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	28	24.14	24.14	28.45
Agree	3.00	57	49.14	49.14	77.59
Strongly agree	4.00	26	22.41	22.41	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Implementing controls to manage risk

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	4	3.45	3.45	3.45
Disagree	2.00	13	11.21	11.21	14.66
Agree	3.00	61	52.59	52.59	67.24
Strongly agree	4.00	38	32.76	32.76	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Using budgets

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	2	1.72	1.72	1.72
Disagree	2.00	8	6.90	6.90	8.62
Agree	3.00	41	35.34	35.34	43.97
Strongly agree	4.00	65	56.03	56.03	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Using policies and procedures

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	2	1.72	1.72	1.72
Disagree	2.00	15	12.93	12.93	14.66
Agree	3.00	52	44.83	44.83	59.48
Strongly agree	4.00	47	40.52	40.52	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Timely communication of risks

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
	, and o	requerey			Percent
Strongly disagree	1.00	5	4.31	4.31	4.31
Disagree	2.00	25	21.55	21.55	25.86
Agree	3.00	55	47.41	47.41	73.28
Strongly agree	4.00	31	26.72	26.72	100.00
Total	•	116	100.00	100.00	

My business deals with the aforementioned problems by means of: Consulting experts

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	34	29.31	29.31	29.31
Disagree	2.00	40	34.48	34.48	63.79
Agree	3.00	26	22.41	22.41	86.21
Strongly agree	4.00	16	13.79	13.79	100.00
Total		116	100.00	100.00	

My business deals with the aforementioned problems by means of: Conducting performance appraisals regularly

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
					Percent
Strongly disagree	1.00	6	5.17	5.17	5.17
Disagree	2.00	20	17.24	17.24	22.41
Agree	3.00	64	55.17	55.17	77.59
Strongly agree	4.00	26	22.41	22.41	100.00
Total	•	116	100.00	100.0	

My business deals with the aforementioned problems by means of: Performing separate evaluations

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative
		-			Percent
Strongly disagree	1.00	7	6.03	6.03	6.03
Disagree	2.00	19	16.38	16.38	22.41
Agree	3.00	60	51.72	51.72	74.14
Strongly agree	4.00	30	25.86	25.86	100.00
Total	•	116	100.00	100.00	

# **Enterprise Risk Management**

How well do you understand the ERM process?

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative Percent
None	1.00	25	21.55	21.55	21.55
Little	2.00	24	20.69	20.69	42.24
Very little	3.00	12	10.34	10.34	52.59
Average	4.00	28	24.14	24.14	76.72
Some	5.00	18	15.52	15.52	92.24
A lot	6.00	9	7.76	7.76	100.00
Total		116	100.00	100.00	

# How many of the ERM components have you?

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative Percent
None	1.00	25	21.55	21.55	21.55
Little	2.00	25	21.55	21.55	43.10
Very little	3.00	21	18.10	18.10	61.21
Average	4.00	27	23.28	23.28	84.48
Some	5.00	13	11.21	11.21	95.69
A lot	6.00	5	4.31	4.31	100.00
Total		116	100.00	100.00	

How much value does ERM add to your business?

Value Label	Value	Frequency	Percent	Valid Percent	Cumulative Percent
None	1.00	23	19.83	19.83	19.83
Little	2.00	23	19.83	19.83	39.66
Very little	3.00	23	19.83	19.83	59.48
Average	4.00	18	15.52	15.52	75.00
Some	5.00	20	17.24	17.24	92.24
A lot	6.00	9	7.76	7.76	100.00
Total		116	100.00	100.00	

# ANNEXURE D: FACTOR ANALYSIS

Value Label		Frequency	Percent	Valid Percent	Cumulative Percent
	1.07	1	0.86	0.86	0.86
	1.33	1	0.86	0.86	1.72
	1.87		0.86	0.86	2.59
	1.93	3	2.59	2.59	5.17
	2.33	3 3	2.59	2.59	7.76
	2.40	4	3.45	3.45	11.21
	2.47	3	2.59	2.59	13.79
	2.53	3	2.59	2.59	16.38
	2.60	8	6.90	6.90	23.28
	2.67	3 3 8 2 4	1.72	1.72	25.00
	2.73	4	3.45	3.45	28.45
	2.80	1	0.86	0.86	29.31
	2.87	2 7	1.72	1.72	31.03
	2.93	7	6.03	6.03	37.07
	3.00	9	7.76	7.76	44.83
	3.07	2	1.72	1.72	46.55
	3.13	5	4.31	4.31	50.86
	3.20	9 2 5 6 9 3 5 8 2 6	5.17	5.17	56.03
	3.27	9	7.76	7.76	63.79
	3.33	3	2.59	2.59	66.38
	3.40	5	4.31	4.31	70.69
	3.47	8	6.90	6.90	
	3.53	2	1.72	1.72	79.31
	3.60		5.17	5.17	84.48
	3.67		5.17	5.17	89.66
	3.73		5.17	5.17	94.83
	3.87	5	4.31	4.31	99.14
	3.93	1	0.86	0.86	100.00
Total		116	100.00	100.00	